



# A multi-modal investigation of structural and functional neural bases of pitch discrimination in musicians and non-musicians

Thesis submitted in accordance with the requirements of the University of Liverpool for the degree of Doctor in Philosophy

By

Jamaan Salem Alghamdi

June 2012



## **IMAGING SERVICES NORTH**

Boston Spa, Wetherby West Yorkshire, LS23 7BQ www.bl.uk

# ORIGINAL COPY TIGHTLY BOUND



In the Name of Allah,

the Most Gracíous, the Most Mercíful

## **Table of Contents**

Declarationvi
Acknowledgements vii
Publicationsviii
Abstractix
List of tablesx
List of figuresxi
Abbreviationsxvi
Chapter 1 : Introduction1
1.1 Introduction2
Chapter 2 : The Physiology of the Auditory Pathway, Pitch Processing and Pitch Memory in Human Brain
2.1 Aims
2.2 Physiology of the auditory pathway7
2.3 Pitch processing, pitch memory and pitch discrimination in human brain9
2.3.1 Pitch processing in musicians10
2.4 Anatomy and functional organisation of brain regions supporting pitch processing and pitch memory
2.4.1 Auditory cortex12
2.4.1.1 The anatomy of the auditory cortex12
2.4.1.2 The functional organization of the auditory cortex
2.4.1.2.1 Tonotopic mapping of the human auditory cortex16
2.4.2 Hippocampus24
2.4.3 Inferior parietal lobule
2.5 Plasticity in musicians
2.5.1 Structural plasticity
2.5.2 Functional plasticity32
2.5.2.1 Plasticity of the auditory cortex and tonotopic reorganisation37
Chapter 3 : The Technology of Functional Neuroimaging and Experimental Paradigms Optimisation
3.1 Aims

3.2 Introduction	41
3.3 Magnetic Resonance imaging (MRI)	41
3.3.1 Magnetic components and larmor frequency	42
3.3.2 Relaxation times	43
3.3.3 Blood oxygenation level dependent (BOLD) fMRI	45
3.3.4 Study design considerations	45
3.4 MEG	49
3.4.1 Electrophysiological Basis of MEG Signals	50
3.4.2 MEG measurement	51
3.4.3 Source localisation	52
3.5 Combining fMRI and MEG	53
3.6 Experimental Design optimisation	55
3.6.1 Acoustic stimulation	55
3.6.2 FMRI experiments	56
3.6.3 MEG experiments	57
3.7 Results	
3.7.1 FMRI results	
3.7.1.1 Defining frequency organization	60
3.7.1.1 Defining frequency organization 3.7.2 MEG results	60 62
<ul><li>3.7.1.1 Defining frequency organization</li><li>3.7.2 MEG results</li><li>3.8 Discussion</li></ul>	60 62 63
<ul> <li>3.7.1.1 Defining frequency organization</li> <li>3.7.2 MEG results</li></ul>	60 62 63 and non- 66
<ul> <li>3.7.1.1 Defining frequency organization</li></ul>	60 62 63 and non- 66 66
<ul> <li>3.7.1.1 Defining frequency organization</li></ul>	60 62 63 and non- 66 66 67
<ul> <li>3.7.1.1 Defining frequency organization</li></ul>	60 62 63 and non- 66 66 67 67
<ul> <li>3.7.1.1 Defining frequency organization</li></ul>	60 62 63 und non- 66 66 67 67 68
<ul> <li>3.7.1.1 Defining frequency organization</li></ul>	60 62 63 and non- 66 66 67 67 67 68 70
<ul> <li>3.7.1.1 Defining frequency organization</li></ul>	60 62 63 and non- 66 66 67 67 67 67 67 
<ul> <li>3.7.1.1 Defining frequency organization</li></ul>	60 62 63 and non- 66 67 67 67 67 67 70 71
<ul> <li>3.7.1.1 Defining frequency organization</li></ul>	60 62 63 and non- 66 67 67 67 67 67 71 71 71
<ul> <li>3.7.1.1 Defining frequency organization</li></ul>	60 62 63 and non- 66 66 67 67 67 67 67 71 71 71 72 73
<ul> <li>3.7.1.1 Defining frequency organization</li></ul>	60 62 63 and non- 66 66 67 67 67 67 67 67 71 71 71 71 72 73
<ul> <li>3.7.1.1 Defining frequency organization</li></ul>	60 62 63 and non- 66 66 67 67 67 67 67 71 71 71 71 73 73 76

Chapter 5 : Neuroanatomical correlates of musicianship and PDP78
5.1 Aims
5.2 Introduction79
5.3 Materials and Methods80
5.3.1 Subjects
5.3.2 Anatomical imaging sessions80
5.3.3 Anatomical investigation methods80
5.3.3.1 Voxel based morphometry80
5.3.3.2 Cortical thickness and sulcal depth82
5.3.3.3 Cortical volume and surface area85
5.3.3.4 Hippocampal shape and subfields volume
5.4 Results
5.4.1 Voxel based morphometry89
5.4.1.1 ANCOVA test with TIV as global variable
5.4.1.2 ANCOVA test with PDPs as covariates90
5.4.2 Cortical thickness, sulcal depth and sulcal and gyral topography.91
5.4.2.1 Musical proficiency effect on sulcal and gyral topography91
5.4.2.2 Effect of instrument type on sulcal and gyral topography96
5.4.3 Cortical volume and surface area of auditory regions101
5.4.3.1 Influence of musical performance on cortical surface area 101
5.4.3.2 Correlation between cortical surface area of auditory regions and PDP
5.4.3.3 Correlation between GM volume of auditory regions and PDP. 103
5.4.4 Hippocampal shape difference and and subfields volume differences104
5.4.4.1 Influence of musical performance on hippocampal surface shape
5.4.4.2 Influence of musical performance on hippocampal subfields volume
5.4.4.3 Correlation between volume of right hippocampal subfields and PDP107
5.4.5 Correlation between musicians' characteristics and both volume of hippocampal subfields and cortical surface area and volume in the auditory regions
5.5 Discussion

5.6 Limitations111
5.7 Conclusion114
Chapter 6 : Tonotopic Mapping of the Human Auditory Cortex and its Relationship with Musical Proficiency: an fMRI Study Using a Stroboscopic Event Related Design
6.1 Aims115
6.2 Introduction116
6.3 Materials and Methods116
6.3.1 Subjects116
6.3.2 Stimuli116
6.3.3 Experiment Procedure117
6.3.3.1 Functional imaging sessions118
6.3.4 Statistical analysis119
6.3.5 FMRI and Anatomical Data Analysis119
6.3.5.1 The exclusion criteria120
6.4 Results123
6.4.1 Influence of musical proficiency on overall activation123
6.4.3 Tonotopic Organization124
6.5 Discussion126
6.6 Limitations127
6.7 Conclusion128
Chapter 7 : Tonotopic Mapping of the Human Auditory Cortex and its Relationship with Musical Proficiency: an MEG Study129
7.1 Aims129
7.2 Introduction130
7.2.1 Auditory evoked fields in musicians and non-musicians
7.3 Materials and Methods132
7.3,1 Subjects
7.3.2 Stimuli
7.3.3 Experimental procedure133
7.3.5 Statistical analysis133
7.3.6 MEG Data Analysis134
7.4 Results135
7.4.1 Influence of musical performance on cortical activation135
7.4.1.1 Dipole magnitude of cortical responses

7.4.2 Latency	139
7.4.3 Tonotopic organisation	139
7.5 Discussion	146
7.6 Limitations	147
7.7 Conclusion	
Chapter 8 : Summary, discussion and conclusion	149
8.1 Introduction	150
8.2 Results and Interpretations	150
8.3 General discussion	
8.3.1 Multimodal imaging approach	
8.3.2 Is it genetic predisposition for music?	157
8.4 Limitations	159
8.5 Conclusion	159
8.6 Future work	161
References	

v

٩.

•• •

## Declaration

The work presented in this thesis is the result of my own work. The material contained in this thesis has not been presented, nor is currently being presented, either wholly or in part, for any other degree or qualification.

Jamaan Salem Alghamdi

Dr. Vanessa Sluming

## Acknowledgements

I would like to express my gratitude to King Abdulaziz University for the scholarship that allowed me to carry this research project.

My deepest gratitude and appreciation go to my supervisors Dr. Vanessa Sluming and Prof. Andrej Stancak, whom I sincerely thank for all the guidance and support in helping me make these significant steps into the world of scientific research and in achieving this aim.

I would like to thank my friends in Liverpool. Also I would like to extend my special thanks to my colleagues in MARIARC, senior radiographer Valerie Adams who helped me collecting functional and structural MRI data and during thesis writing up, senior experimental officer Bill Bimson for his help solving technical problems with MRI and MEG scanners. Also I would like to thank Dr. Laura Parkes who helped me during the first stage of this research project especially the fMRI design and whom I learned from how to use BrainVoyager QX. Thanks also to Dr. Peter Schneider for providing me with pitch discrimination tests and for his help with MEG experiment design, and Dr. Roozbeh Rezaie for his help with MEG experiments during the pilot study. I would like to thank Dr. Guy Lightfoot for his help during the calibration tools of MR Confon GmbH for providing me with the calibration tools of MR Confon sound delivery system that used during fMRI experiments.

My warmest thanks go to my parents, my brothers and my sisters, who were always willing to listen and supported me in so many ways. I hope I have made them proud and continue to do so.

Finally, my deepest thanks go to my wife and my kids for their love, patience and support during this research project.

Jamaan 2012

## **Publications**

#### Papers

- ALDHAFEERI, F. M., MACKENZIE, I., KAY, T., ALGHAMDI, J. & SLUMING, V. 2012. Neuroanatomical correlates of tinnitus revealed by cortical thickness analysis and diffusion tensor imaging. Neuroradiology.
- Aldhfeeri F, Mackenzie I, Kaye A, <u>Alghamdi J</u> and Sluming V (2012) Regional brain responses to pleasant and unpleasant IAPS pictures: different networks. Neuroscience Letters. 512(2):94-8
- STANCAK, A., <u>ALGHAMDI, J. & NURMIKKO, T. J. 2011.</u> Cortical Activation Changes during Repeated Laser Stimulation: A Magnetoencephalographic Study. PLoS One, 6, e19744.
- Abdul-Kareem I, Stancak A, Parkes L, Al-Ameen M, <u>Alghamdi J</u>, Aldhafeeri F, Embleton K, Morris D, Sluming V (2011) Plasticity of the superior and middle cerebellar peduncles in musicians revealed by quantitative analysis of volume and number of fibers based on diffusion tensor tractography. Cerebellum.

#### **Conference Proceedings**

- <u>Alghamdi J</u> and Sluming V (2012) Neuroanatomical correlates of musicianship in the right hippocampus. Proceedings of the European Congress of Radiology. ESR, Vienna pp xx (Certificate of merit)
- Aldhafeeri F, <u>Alghamdi J</u>, Sluming V (2011) Functional MRI of regional brain responses to pleasant and unpleasant IAPS pictures: different networks. Proceedings of the European Congress of Radiology. ESR, Vienna pp xx
- <u>Alghamdi J</u>, Stancak A, Parkes L, Lightfoot G, Adams V and Sluming V (2010) Multimodal imaging of frequency coding in the human auditory cortex. Proceedings of 16th Annual Meeting of the Organization for Human Brain Mapping. Elsevier, Barcelona pp xx
- <u>Alghamdi J</u>, Stancak A, Parkes L, Lightfoot G, Adams V and Sluming V (2010) Multimodal imaging of frequency coding in the human auditory cortex. Proceedings of 16th Annual Meeting of the Organization for Human Brain Mapping. Elsevier, Barcelona pp xx
- Stancak A, <u>Alghamdi J</u>, Turo Nurmikko (2010) A magnetoencephalographic study of temporal summation of pain. Annual Meeting of the Organization for Human Brain Mapping. Elsevier, Barcelona pp xx
- <u>Alghamdi J</u>, Stancak A, Parkes L, Sluming V (2009) Tonotopic Mapping using fMRI and MEG: A Multimodal Study of the Auditory Cortex in Pitch Discrimination. Auditory Cortex: Current Concepts in Human and Animal Research. ICAC, Magdeburg pp 67
- <u>Alghamdi J</u>, Stancak A, Parkes L, Sluming V. (2009) Tonotopic Mapping Using Both fMRI and MEG Scans. Proceedings of the National Meeting of the British Neuroscience Association. BNA, Liverpool pp 1

#### Scripts and tools

- Multiple DICOM to NIfTI converter using dcm2nii and AppleScript http://www.cabiatl.com/mricro/mricron/install.html http://www.easyneuroimaging.com
- Using bash script and Applescript to convert freesurfer stats2table (asegstats2table & aparcstats2table) http://surfer.nmr.mgh.harvard.edu/fswiki/freesurferstats2table http://www.easyneuroimaging.com
- Easyfreesurfer and multi mri\_convert http://www.freesurfer.net/fswiki/Easyfreesurfer http://www.easyneuroimaging.com
- Some other scripts were written for Brainvoyager QX and FSL http://www.easyneuroimaging.com

### Abstract

Musicians represent an ideal model for understanding experience-driven neuroplasticity in the human brain, especially in auditory and motor domains. Musicians exert intensive and durable practice of various multimodal skills (e.g., motor, auditory, visual and memory). It has been reported that certain regions of the adult musicians' brains are structurally larger than nonmusicians. Also musicians demonstrate more sensitive pitch discrimination abilities compared to non-musicians because pitch labelling plays an important role in music. Music is made of a highly structured and complex succession of tones that arranged in a specific rhythm and played at specific pitch.

In this thesis I aimed to; (1) Explore the influence of musical proficiency on pitch discrimination ability with investigating the laterality pitch discrimination ability and exploring some factors that could affect pitch discrimination ability such as aging, type of musical instrument and duration of musical proficiency. (2) Investigate anatomical plasticity of selected brain structures in musicians with exploring musical proficiency and the instrument type effect on sulcal and gyral topography. (3) Study the correlations between pitch discrimination performance and some cortical features of selected brain structures. (4) Investigate tonotopic mapping in the human auditory cortex using functional magnetic resonance imaging and (fMRI) and magnetoencephalography (MEG). (5) Examine the influence of musical proficiency expertise on frequency organization and cortical activation of the human auditory cortex.

Different structural methods were implemented to study differences between musicians and non-musicians in some structural features. FMRI and MEG were used to study tonotopic mapping in the human auditory cortex.

During pitch discrimination tasks, musicians demonstrated more sensitivity than non-musicians. Musicians also showed significantly larger volume in various brain regions and shape differences in sulcal and gyral anatomy and the right hippocampus. Additionally, there were significant correlations between pitch discrimination performance and different structural measurements. Musicians had stronger BOLD fMRI in the medial and lateral part of the left HG. ANOVA tests of the amplitude of neuromagnetic N100 component showed a group effect of borderline significance.

Results clearly show behavioural, structural and functional differences between these two groups. These results indicate that the morphology and neurophysiology of various brain regions and the pitch labelling have an essential role in musical proficiency.

ix

## List of tables

Table 3.2 Participants' details. M= Musicians, C= non-musicians. Right ticks are for subjects who participated in the studies. Asterisk flags are for subjects who have been included in fMRI and MEG analysis......65

## List of figures

Figure 2.3 Coronal, axial and surface views of central auditory system; HG (blue), primary auditory cortex (dark blue) and secondary auditory cortex (red), mHG and lHG are medial and lateral HG respectively (subject C13. 3D mesh reconstructed for illustration purpose using BrainVoyager QX 2.3). ...14

Figure 3.6 Best-frequency maps in the auditory cortex (subject C17, P<0.01 cortex-based Bonferroni adjustment (Formisano et al., 2002)), layout map is superimposed on a mesh reconstruction of the subject's cortex......60

Figure 3.7 The cortical best-frequency maps of six tones are displayed on mesh reconstruction and inflated representation of the subject's auditory cortices, HG = Heschl's gyrus, (subject C17, P<0.01 FDR adjusted)......61

Figure 4.2 95% confidence interval for PDP in the left and RE in musicians.

Figure 5.1 Statistical parametric map (extent threshold  $P_{corr}$ =.05) showing clusters with statistically significant of increased GM density in musicians. 90

Figure 5.2 Statistical parametric map (extent threshold  $P_{corr}$ =.05) showing clusters with statistically significant of increased GM density in musicians. 91

Figure 5.8 Average cortical contours from string players (blue) and non-string players (yellow) are superimposed on average target volume for all subjects.

Figure 5.10 T-statistic maps of group difference in cortical thickness are superimposed on average target surface for all subjects......100

Figure 5.14 Statistic maps of surface and shape difference in the right hippocampus. The colour bar indicates the statistic values; an increase from red to blue is going from lower to higher statistical significance......104

Figure 5.15 Estimated means of volume of the right and left hippocampal subfields (presubiculum, subiculum, fimbria, CA1, CA2/3 and CA4/DG)..106

Figure 5.16 Correlation between PDP of all conditions and volume of fimbria (Smaller score of pitch discrimination means better performance)......107

Figure 6.2 Schematic diagram of the stroboscopic event related design. The presentation of the tone occurred at variable offsets before each scan.......118

Figure 7.3 The baseline to peak dipole amplitude of the N100m response of the grand average of each group. The error bars indicate the standard error.

Figure 7.5 The latencies of N100m peaks of musicians and non-musicians are strongly frequency dependent. The error bars indicate the standard error...139

Figure 7.6 Individual tonotopic mapping of the N100m components in nonmusicians (letter C) and musicians (letter M). Dipole loclisations are superimposed on coronal and transverse slices in standard Talairach space.

Figure 7.7 Tonotopic mapping of the N100m components in musicians. Dipole localisations are superimposed on coronal and transverse slices in standard Talairach space of an average volume of all participants......142

## Abbreviations

- AELLF Auditory Evoked Long Latency Fields
- AEMLF Auditory Evoked Short to Middle Latency Fields
- AF Arcuate Fasciculus
- amHG Anteromedial Portion of HG
- ANOVA Analyses of Variance
- AP Absolute Pitch
- B<sub>0</sub> External Magnetic field applied in MRI
- BE Both Ears
- BF Best Frequency
- BM Basilar Membrane
- BOLD Blood Oxygenation Level Dependent
- CC Corpus Callosum
- CF Characteristic Frequency
- CER Conventional Event Related
- CP Corticospinal Projections
- CSF Cerebrospinal Fluid
- DL Difference Limen
- DLF Difference Limen for Frequency
- DTI Diffusion Tensor Imaging
- EEG Electroencephalography
- EPI Echo Planar Image
- EPSP Excitatory Postsynaptic Potential
- FA Fractional Anisotropy
- FDR False Discovery Rate
- FM Frequency-Modulated
- FMDL Frequency Modulation Detection Limen
- FMRI Functional Magnetic Resonance Imaging
- FWHM Full Width at Half Maximum
- GD Gyrus Dentatus
- GLM General Linear Model
- GM Grey matter
- HDE Hybrid Depth Electrode
- HG Heschl's Gyrus
- HP Hippocampus Proper
- HRF Hemodynamic Response Function
- JND Just Noticeable Difference
- IPL Inferior Parietal Lobule
- IPSP Inhibitory Postsynaptic Potential
- ISI Interstimulus Interval
- HRF Hemodynamic Response Function

- LH Left Hemisphere
- 1HG Lateral Aspect of HG
- MEG Magnetoencephalography
- MEP Motor Evoked Potentials
- MMN Mismatch Negativity
- MNI Montreal Neurological Institute
- MRI Magnetic Resonance imaging
- NMR Nuclear Magnetic Resonance
- NMV Net Magnetization Vector
- PAC Primary Auditory Cortex
- PDP Pitch Discrimination Performance
- PET Photon Emission Tomography
- PSP Postsynaptic Potential
- PT Planum Temporal
- rCBF Regional Cerebral Blood Flow
- RE Right Ear
- RH Right Hemisphere
- RHipp Right Hippocampus
- SD Standard Deviation
- SER Silent Event Related
- SICI Short-Latency Intracortical Inhibition
- SIMEX Simultaneous Multi-Slice Excitation
- SL Strong-Learners
- SMG Supramarginal Gyrus
- SNR Signal To Noise Ratio
- SQUIDs Superconducting Quantum Interference Devices
- SPET Single Photon Emission Tomography
- SSF Steady-State Fields
- STG Superior Temporal Gyrus
- STS Superior Temporal Sulcus
- T1 Longitudinal Relaxation Time
- T2 Dephasing Relaxation Time
- T2<sup>\*</sup> Transverse Magnetization Time
- TE Echo Time
- TIV Total Intracranial Volume
- TR Repetition Time
- VBM Voxel-Based Morphometric
- WL Weak-Learners
- WM White Matter

## **Chapter 1 : Introduction**

#### **1.1 Introduction**

Expert musical performance is one of humankind's most complex and extensive multimodal activities, as professional musicians practice their rehearsal exercises with high attention and long lasting commitment. While playing an instrument the musician integrates various sensory modalities and cognitive processes comprehensively and extremely efficiently. These processes activate multiple brain regions (Langheim et al., 2002, Lotze et al., 2003, Meister et al., 2004, Bangert et al., 2006, Sluming et al., 2007a, Groussard et al., 2010a, Herdener et al., 2010). Therefore professional musicians are an ideal population to investigate brain functions and connectivity. Different studies showed induced cortical plasticity in both pitch memory and musical training in studies that investigated brain structure/function before and after a prescribed training programs (Fujioka et al., 2006, Gaab et al., 2006, Lappe et al., 2008, Hyde et al., 2009, Schlaug et al., 2009a, Herdener et al., 2010). Differences between musicians and non musicians regional brain structure and function have also been reported to be attributed to musical skill development (Schlaug et al., 1995a, Pantev et al., 1998, Schlaug, 2001, Schneider et al., 2002, Gaab and Schlaug, 2003, Gaser and Schlaug, 2003, Schulz et al., 2003, Bangert et al., 2006, Musacchia et al., 2007, Bermudez et al., <sup>2009</sup>, Abdul-Kareem et al., 2011a, Halwani et al., 2011, Loui et al., 2011).

Pitch discrimination is one of the important skills for musicianship. Pitch plays a key role in audition, especially in the perception of speech and music (Plack et al., 2005). Theoretically brain processes pitch depending on either the spatial

(Helmholtz and Ellis, 1954) or temporal (Rutherford, 1886) patterns of excitation in the auditory periphery. However, using either model alone would not provide a complete explanation of pitch perception (Bendor, 2011). Some recent studies reported that pitch perception is based on spatiotemporal excitation patterns (Oxenham et al., 2004, Cedolin and Delgutte, 2010). During pitch discrimination tasks, musicians showed superior performance comparing with non-musicians (Kishon-Rabin et al., 2001, Tervaniemi et al., 2005, Micheyl et al., 2006).

The main aim of this thesis was to investigate the tonotopic mapping and the neuroanatomical and neurofunctional correlates of musicianship and neuroanatomical correlates of pitch discrimination performance (PDP) in the human auditory cortex.

Specific objectives include to:

- Investigate the influence of musical proficiency on PDP.
- Explore the neuroanatomical correlates of both musicianship and PDP in some brain regions that associated with pitch memory and at the same time I explored the effects of type of instrument on sulcal and gyral topography.
- Implement several novel features. Firstly, it is the first cross-sectional study that incorporated behavioural, anatomical and functional investigations on the same participants to study the differences between musicians and nonmusicians. Second novel feature was using a multi-method approach (VBM, sulcal depth and cortical thickness measurements, sulcal and gyral topography and hippocampus shape) to investigate the neuroanatomical correlates of musicianship and PDP. Finally, it is the first study that

integrated multimodality (fMRI and MEG) in investigating tonotopic mapping in humans.

The literature review, experiments and the summary are presented in the following chapters that are organised as follows:

Chapter 2 gives a brief overview of physiology of the auditory pathway, pitch processing and pitch memory in the human brain. Also it presents anatomy and functional organisation of brain regions supporting pitch discrimination in the human brain e.g. auditory cortex, hippocampus and Inferior parietal lobule. It then presents a relevant literature about structural and functional plasticity in musicians.

Chapter 3 presents a basic fundamental overview of the physical principles of MRI, fMRI and MEG. It then presents a relevant literature about paradigm designs that have been implemented in some recent studies to overcome the confounding effect of scanner noise. In the last part, results of different optimised designs are discussed.

Chapter 4 gives a literature about pitch perception, pitch discrimination and the difference in pitch discrimination between musicians and non-musicians. It also explores the influence of musical performance expertise on pitch discrimination ability with taking in account laterality effect of pitch discrimination ability. Then it discusses the effect of different factors such as aging and period of musical experience on PDP.

Chapter 5 uses a multi-method approach to investigate the neuroanatomical correlates of musicianship and PDP. Different anatomical techniques are applied to explore plasticity of different brain regions by measuring various cortical features including; grey matter density, cortical thickness and sulcal depth cortical volume and surface area, and hippocampal volume and shape differences. In addition, the effect of instrument type on sulcal and gyral topography is studied.

Chapter 6 investigates tonotopic mapping in the human auditory cortex using fMRI. It also, explores the influence of musical performance expertise on frequency organization and on cortical activation of the human auditory cortex.

Chapter 7 studies tonotopic mapping in the human auditory cortex using MEG. It also, investigates the effect of musical proficiency on frequency organization and on cortical activation of the human auditory cortex.

5

Chapter 8 summarises and discusses the main findings of this thesis.

# Chapter 2 : The Physiology of the Auditory Pathway, Pitch Processing and Pitch Memory in Human Brain

#### 2.1 Aims

Specific aims of this chapter were to describe the:

- i) Physiology of the auditory pathway, pitch processing and pitch memory in the human brain,
- ii) Anatomy and functional organisation of brain regions supporting pitch processing and pitch memory in the human brain including;
  - a. Auditory cortex.
  - b. Hippocampus.
  - c. Inferior parietal lobule.
- iii) Structural and functional plasticity in musicians.

#### **2.2** Physiology of the auditory pathway

Auditory signals travel from the tympanic membrane of each ear to the cochlear nuclei in the respective hemispheres through auditory nerve fibres. These nuclei, known as the ventral and the dorsal nuclei, are located on each side of the medulla. Each auditory nerve fibre splits to project in both nuclei. These nuclei show specific frequency response characteristics and are organized into subdivisions. High frequency sound waves are processed in the dorsal, caudal and medial parts of each cochlear nucleus, while the ventral, rostral, and lateral subdivisions act in response to low frequency sound waves. From the cochlear nuclei, the majority of auditory fibres project in a group of nuclei also located in the basal brainstem known as the superior olivary complex. Each superior olive receives projections from both the ipsilateral and contralateral cochlear nucleus. The medial and lateral superior olive nuclei play an important role in soundsource localization. Time difference is processed in the medial superior olive while lateral superior olive nucleus responds to phase differences (amplitude difference of sound waves). This information is subsequently transformed in an auditory spatial map located in the inferior colliculus (Bazwinsky et al., 2003, Goldstein et al., 2005).

Fibres connect between the superior olivary complex and the inferior colliculus through the lateral lemniscus tract then two distinct pathways arise from the inferior colliculus to project at the ventral and dorsal regions of the medial geniculate nuclei of the thalamus and alternatively, projections from the dorsal

medial geniculate nucleus reach the secondary auditory cortex, which is known as the final output from the auditory pathways (Figure 2.1) (Bazwinsky et al., 2003, Goldstein et al., 2005).



Figure 2.1 The central auditory pathways extend from the cochlear nucleus to the auditory cortex (Adapted from Kandel 2000)

#### 2.3 Pitch processing, pitch memory and pitch discrimination in human brain

Pitch plays an important role as a fundamental auditory attribute in auditory communication, as a part of the basis of melody in music (Moore, 2003, Jones et al., 2010) and prosody in speech (Thompson et al., 2003) (For details see section 4.3).

Some neuroimaging studies considered the lateral part of Heschl's gyrus (HG) as the pitch processing centre (Griffiths et al., 1998a, Griffiths et al., 2001, Patterson et al., 2002, Griffiths, 2003, Krumbholz et al., 2003, Warren and Griffiths, 2003). These studies revealed that there is a hierarchical pitch processing in humans beginning in subcortical structures e.g. brainstem (Griffiths et al., 2001) and extending up through the HG out onto the planum polare and the planume temporale (PT) (Griffiths et al., 1998a, Patterson et al., 2002, Warren and Griffiths, 2003). In contrast, recent studies (Hall and Plack, 2009, Barker et al., 2011) reported that there is no evidence of a pitch centre in lateral HG. Hall and Plack (2009) suggested that parts of the PT are more relevant for pitch processing than lateral HG, and in some listeners, pitch responses occurred in the temporoparieto-occipital junction or prefrontal cortex.

Pitch memory plays an important role during PDP. Reference sounds must be stored in memory to discriminate target pitches. It has been reported that hippocampus plays an important role in pitch memory and auditory temporal information in animal (Sakurai, 2002) and human (Watanabe et al., 2008).

O'Mara et al. (2009) showed that the dorsal part of subiculum, which considered as part of the hippocampus, is relatively more concerned with space and memory. Different studies reported that inferior parietal lobule (including supramarginal gyrus (SMG) and angular gyrus) plays an important role in the auditory and verbal short memory (Paulesu et al., 1993, Salmon et al., 1996, Binder et al., 1997, Celsis et al., 1999, Gaab et al., 2003, Gaab et al., 2006, Romero et al., 2006, Vines et al., 2006, Rinne et al., 2009). Additionally, it has been reported that pitch discrimination processes take place in the HG (Tramo et al., 2002, Brown et al., 2004, Gaab et al., 2006). Mathys et al. (2010) used transcranial direct current stimulation to address the problem of causality in pitch discrimination functions and found that both left and right HG are involved in pitch discrimination processs with stronger contributor from right HG.

#### 2.3.1 Pitch processing in musicians

Pitch discrimination is regarded as a fundamental ability for a successful musician. Musicians show better auditory capability compared to non-musicians, which has sparked multi discipline research interests in hearing science, psychology and neuroimaging. Various explanations have been suggested to describe how musicians present better pitch discrimination performance than non-musicians. Musicians may have an improved overall auditory capability. This may be related to their ability to ignore the interference of dubious timbral attribute of a tone, which could bias pitch perception. Additionally, playing music professionally requires performing fine pitch discrimination tasks in order to tune

musical instruments. The human brain processes pitch in a hierarchal manner starting in the primary auditory cortex to discriminate frequencies moving to higher level processes. Higher level processes include segregation of pitch contour and group of tones into melodies and harmony patterns which originate in the secondary auditory cortex, SMG, inferior frontal gyrus, and superior parietal lobe.

During pitch discrimination task, Gaab and Schlaug (2003) showed that musicians seem to rely more on a neural network that contain SMG and superior parietal cortex, while non-musicians use HG and hippocampal gyrus. In contrast, Schmithorst and Holland (2003) investigated melodic and harmonic processing. Both musicians and non-musicians showed activation similarity for melodic processing in the most anterior part of the STG. Additionally, musicians presented activation in the inferior parietal lobules for both tasks. Brattico et al (2001) used electroencephalography (EEG) to investigate pitch perception in musicians and non-musicians and reported that musicians had a faster neural response for pitch changes than the non-musicians. Although, non-musicians have an innate knowledge of musical regularities, musical practice creates a larger number of explicit memory representations. Therefore, it can be reported that auditory pitch perception for higher level processes are influenced by musical context and syntax (Brattico et al., 2001, Koelsch et al., 2002b).

## 2.4 Anatomy and functional organisation of brain regions supporting pitch processing and pitch memory

#### 2.4.1 Auditory cortex

#### 2.4.1.1 The anatomy of the auditory cortex

Auditory cortex of non-human primates can be divided into a core of primary areas, a surrounding belt and a parabelt region just lateral to the belt on the superior temporal gyrus (Figure 2.2). The core is subdivided into three fields (A1, R and RT), while the belt is subdivided into seven fields that surround the core along the upper surface of the superior temporal gyrus (Figure 2.2) (Hackett et al., 1998, Kaas and Hackett, 2000, Rauschecker and Scott, 2009).



Figure 2.2 Dorsolateral view of the macaque cerebral cortex after removal of the overlying parietal cortex, exposing the ventral bank of the lateral sulcus and insula. The approximate locations of the core region (solid red line), caudal and lateral portions of the belt region (dashed yellow line), and the parabelt region (dashed orange line) are shown. The schematic diagram shows the subdivisions of core, belt and parabelt regions (Adapted from Kaas and Hackett 2000).

The core is characterized by koniocortical architecture and dense myelination. Pyramidal cells of the belt area are larger than those of the core, but cell density, columnar spacing and myelination are lesser in the belt than in the core (Hackett et al., 1998, Kaas and Hackett, 2000). In contrast, human auditory areas can be divided -based on gross morphological criteria- into the primary auditory cortex (PAC) and non-primary auditory cortex (Hall et al., 2003) (see Figure 2.3).

#### Human primary auditory cortex

PAC occupies posteromedial portion of the HG that located on the superior temporal gyrus (STG) in the temporal lobe Figure 2.3 It is surrounded laterally and posteriorly by non-PAC areas, while it is separated medially from the insula by a narrow strip of prokoniocortex. The transverse gyrus is often partially duplicated. If there are multiple transverse gyri, PAC is located in the anteriormost gyrus (Rademacher et al., 2001). It is essential to use a combination of cytoand myelo-architectonic and histochemical markers to provide best reliable determination of anatomical borders (Hall et al., 2003, Abdul-Kareem and Sluming, 2008). The cytoarchitecture of the PAC is described as koniocortex or highly granular cortex and classified by Brodmann as area 41 (termed BA 41). It is also known as region Te1 and surrounded by less granular cortices, areas Te2, Te3, and TI1. PAC has distinctly dense granular cells, relatively small pyramidal cells in layer III, dense myelination, a prominent width of layer IV, and a relative cell-sparse layer V (Rademacher et al., 2001, Abdul-Kareem and Sluming, 2008). The position, the extent and the absolute size of this area vary across individuals and between the left and right hemispheres in one individual (Rademacher et al.,

2001). Comparing to the right hemisphere (RH) the PAC is located more posteriorly in the left hemisphere (LH). Auditory cortical areas in one hemisphere are connected to the reciprocal areas in the other hemisphere through corpus callosum. It has been reported that there are sex differences in the activation and size of the PAC (Morosan et al., 2001, Ruytjens et al., 2007). Furthermore, these authors reported that the transverse temporal gyrus has a larger volume in the left than in the RH.



Figure 2.3 Coronal, axial and surface views of central auditory system; HG (blue), primary auditory cortex (dark blue) and secondary auditory cortex (red), mHG and lHG are medial and lateral HG respectively (subject C13. 3D mesh reconstructed for illustration purpose using BrainVoyager QX 2.3).

PAC contains neurons that respond to a limited range of frequency and register the characteristics of sound identification or perceptual processing, such as loudness, pitch and timbre. The response of neurons in the PAC to specific sound wave frequencies is narrow and binaural (Reser et al., 2000). Some neurons are excited by stimulation from both ears while others respond to one ear and are inhibited by stimulation from the other. PAC is organized in vertical columns. The cells in each column are sensitive to a similar main sound frequency and neighbouring columns are mapped tonotopically (Reser et al., 2000).

#### Human non-primary auditory cortex

The non-primary auditory cortex, which situated laterally to PAC, can be divided into secondary auditory cortex and tertiary auditory cortex. Secondary auditory cortex is classified histologically as Te2 or Brodmann's area 42 (PT). It is less granular than PAC and contains large pyramidalcells in layer IIIc. The function of secondary auditory regions in spectrotemporal processing is likely to be broader than for speech alone. In addition, the transition from primary to secondary auditory cortex is responsible of transformation of the physical characteristics of sound into information about objects in the auditory scene. Furthermore, secondary auditory cortex plays a big role in processing of harmonic, melodic and rhythmic patterns. Finally, the secondary cortex seems to respond preferably to stimuli with sufficiently complex spectral dynamics, while the primary cortex has been found to respond to a broad range of auditory stimuli. The tertiary auditory cortex is located on the superior temporal gyrus and the superior temporal sulcus (STS) within the temporal lobe. This cortex supposedly integrates everything into the overall experience of music (Reser et al., 2000).

#### 2.4.1.2 The functional organization of the auditory cortex

Many studies have been used to assess the functional organization of the auditory regions in mammals including humans. Most of these studies examine tonotopic (Merzenich and Brugge, 1973, Howard et al., 1996, Recanzone et al., 1999), amplitopic (Pantev et al., 1989, Bilecen et al., 2002) and spatiotopic (Griffiths et al., 1998b, Zatorre et al., 2002b) maps of the auditory regions. These maps can be defined as the spatial distribution of neurons that are maximally responsive to certain stimulus frequencies, intensities and position respectively. The following section will provide a detailed overview about the tonotopic mapping of the auditory cortex.

#### 2.4.1.2.1 Tonotopic mapping of the human auditory cortex

In the auditory cortex, higher frequencies are located in more anterior medial regions while the posterior lateral regions responding to lower frequencies (Howard et al., 1996). Tonotopic mapping has been well examined using microelectrode recordings in various animal species (Merzenich and Brugge, 1973, Merzenich et al., 1975, McMullen and Glaser, 1982, Galazyuk, 1990, Schreiner, 1991, Kosaki et al., 1997, Recanzone et al., 1999, Bendor and Wang, 2005, Kalatsky et al., 2005). For example; seven tonotopic mappings and four or more have been observed in the cat and monkey cortices respectively (Merzenich and Brugge, 1973, Merzenich et al., 1975, Schreiner, 1991, Kosaki et al., 1997, Recanzone et al., 1999).
In the auditory cortex of nonhuman primates, each of the three divisions of the core area show tonotopic gradients that are mirror symmetric to each other (Merzenich and Brugge, 1973, Kosaki et al., 1997, Kaas and Hackett, 2000, Bendor and Wang, 2008). Additionally, tonotopic gradients have also been exhibited in the belt area (Kosaki et al., 1997, Kaas and Hackett, 2000, Petkov et al., 2006, Kusmierek and Rauschecker, 2009, Petkov et al., 2009).

However, the number and location of tonotopic organizations within human auditory cortex are still unclear either anatomically or functionally. Compared to animals, humans were shown to process additional complex sound patterns, such as language or music. Various neuroimaging modalities have been used to detect a frequency-depth tonotopic organization. Most of those studies can be divided into two groups, those using two stimulus frequencies and those using more than two frequencies. Studies using two stimulus frequencies have demonstrated that the cortical location of peak response is frequency dependent. On the other hand, studies that used more than two acoustic frequencies generally observed a covariance of position of response with stimulus frequency (Talavage et al., 2004)...

## Hybrid depth electrode (HDE)

HDE has provided a direct demonstration of tonotopic mapping along the HG. Howard et al (1996) implanted this technique in epilepsy surgery patients who required depth electrode placement as part of their standard clinical treatment

plan. The authors found a lateral progression in frequency sensitivity from high frequency (3360 Hz) to lower frequency (1480 Hz).

#### Positron and single photon emission computed tomography (PET) and (SPECT)

In 1985, Lauter et al. used stimuli with sine tones of 500 Hz and 4000 Hz. The authors measured the regional cerebral blood flow (rCBF) and metabolic rates of oxygen and glucose consumption upon both stimuli quantitatively. They could not localize the activated regions in the cortex precisely however; the rCBF changes induced by a 4000 Hz tone were located deeper and posterior to those induced by 500 Hz tone (Lauter et al., 1985). Additionally, Ottaviani et al (1997) showed similar results using SPECT.

#### EEG and MEG

Both systems have been used to investigate tonotopic mapping of human auditory cortex since early eighties (Romani et al., 1982a, Romani et al., 1982b). Different designs have been implemented to investigate tonotopic mapping of the auditory cortex in humans.

# Auditory evoked steady-state fields (SSF)

Auditory steady-state stimuli are amplitude- or frequency-modulated tones with a modulation rate of 35–40 Hz yielding the maximum response energy. The first demonstration of tonotopy in humans using steady-state field (SSF) showed that the source of the response in HG moves more laterally as frequency decreases. (Romani et al., 1982a) Furthermore, Pantev et al. (1996) used sequences of

Gaussian tone pulses with a repetition rate of 39 Hz and carrier frequencies of 250, 500, 1000, 2000, and 4000 Hz, respectively to obtain SSF of the human auditory cortex. The authors found that the depth of the estimated source shifted medially with increasing carrier frequency. Their findings were confirmed by recent SSF studies (Ross et al., 2000, Weisz et al., 2004).

# Auditory evoked short to middle latency fields (AEMLF)

Middle latency evoked fields occur about 25–50 ms post-stimulus and originate in PAC with P30m and P50m representing the most permanent components. While short latency components start in the range of 15 to 30 ms after stimulus and originates in upper brainstem and/or auditory cortex, the N19m-P30m components are distributed in the medial half to the first transverse temporal gyrus of Heschl (Liegeois-Chauvel et al., 1991, Liegeois-Chauvel et al., 1994, Schneider, 2001, Schneider et al., 2002, Lütkenhöner et al., 2003a). Pantev et al. (1995) presented data in the range from middle to long latencies. However, only the P30m/P30 (30 ms) and N100m/N100 (89 ms) were consistently present in all stimulus conditions and all subjects. This was the first study to use multi-channel data of both MEG and EEG recordings to study tonotopic mapping of the human auditory cortex. The source of the P30m/P30 components was in the medial part of HG and shifted laterally by increasing frequencies.

## Auditory evoked long latency fields (AELLF)

The long latency cortical auditory fields start 50 to 200 ms post-stimulus; originating in the auditory cortex with the N100m representing the most

permanent components. Pantev et al. (1995) reported that the tonotopic gradient of N100m/N1mm components was found posterior to the HG in the PT and have opposite orientation compared to P30m/P30 components. This suggests that there is a mirror tonotopic organisation in human auditory cortex with sources of P30m/P30 and N100m/N100 located in the primary and secondary auditory cortices, respectively. Yamamoto et al. (1988) showed that a dipole source analysis of the major component of the long auditory evoked field (N100m) was informative about the tonotopic organization of the human auditory cortex. In addition, Lütkenhöner and Steinstrater (1998) explored the tonotopic mapping of the N100m components in single subject extensively. The authors registered the estimated dipole sources to three-dimensional reconstruction of the cortical surface derived from magnetic resonance images. Tones of 500 ms duration with frequencies of 250, 500, 1000, and 2000 Hz were presented, and the total number of stimuli presented per condition was about 3600. The tonotopic organization of N100m sources were consistent with all of the previous studies (Lütkenhöner and Steinstrater, 1998). The authors also demonstrated specific temporal patterns in each source. This indicates involvement of multiple areas in the generation of N100m. They proved that the source locations of P200m (150-200 ms) components were also shown to be dependent on the frequency of auditory stimulus (Lütkenhöner and Steinstrater, 1998). In contrast, Lütkenhöner et al. (2003b) showed that tonotopic organization of the N100m source of a large interindividual variability. Schneider (2001) observed a mirror tonotopic

organisation in the human primary and secondary auditory cortices with opposite gradients of sources reported by Pantev et al. (1995).

<u>fMRI</u>

Studying tonotopic organization in the human auditory cortex is a challenge as a frequency change by an octave is associated with a change in location by about 2-3 mm which is at the current limit of spatial resolution of fMRI protocols (Romani et al., 1982a, Pantev et al., 1988, Yamamoto et al., 1988, Pantev et al., 1996, Lütkenhöner et al., 2003b). FMRI has relatively high spatial resolution comparing with other non-invasive modalities e.g. EEG, MEG and PET. Advances in hardware, paradigm designs and data-analysis approaches of fMRI have made it the most widely used modality to examine tonotopic mapping in human in the past decade.

Several stimulus paradigm designs and delivery methods have been used to investigate tonotopic mapping, which can be broadly divided into block and event related (ER) designs. The block design involves ON/OFF conditions where stimuli are presented during ON conditions with subsequent rest periods, each condition is typically of 20–40 seconds duration (Rosen et al., 1998, Amaro et al., 2002). In contrast, ER design involves the presentation of discrete events at a certain interstimulus interval. In this design, discrete events would produce different BOLD responses in the same brain region in the case of sampling the full hemodynamic response function (HRF) curve. The ER design provides better temporal resolution than a block design (Amaro and Barker, 2006). In addition, it

is more flexible than the block design, however it has less statistical power (Rosen et al., 1998, Amaro et al., 2002) and comparatively low volume coverage (Yang et al., 2000). Using a block design, Wessinger et al. (1997) obtained a tonotopic map where low-frequency tones were located more anteriorly and laterally than those for high-frequency tones. Strainer et al. (1997) obtained multiple tonotopic mapping using four pure tones. Additionally, Talavage et al. (2000) demonstrated eight regions of response; each exhibiting a greater sensitivity to either lower or higher frequency stimulation. Seifritz et al. (2006) produced tonotopic mirrorsymmetric maps in the auditory cortex, however, the gradient axis of the maps varied across subjects. Additionally, Striem-Amit et al. (2011) represented multiple mirror-symmetric tonotopic maps covering most of the core and highorder human auditory cortex, stretching all the way to the superior temporal sulcus. Yang et al. (2000) used conventional event related (CER) and silent event related (SER) to study tonotopic mapping in the human auditory cortex although the CER design could not detect any frequency gradient, the SER showed distinct spatial shifts in the lateral part of HG with changing frequency. Le et al. (2001) used three designs in the one study; a block, a CER and an SER designs. This study provided evidence for tonotopic organization in the human auditory cortex using both CER and SER, but not using block design. The low-frequency tones were located more lateral and anterior than high frequency tones. In addition, other studies used "stroboscopic" (Belin et al., 1999) event related design (Belin et al., 1999, Formisano et al., 2003, Upadhyay et al., 2007) to produce more detailed frequency organization regions in subdivisions of the auditory cortex. In

this design, auditory stimuli are presented in a pseudorandomised manner at different delays before each volume acquisition (0, 2, 3, 4, 5, 6, 7 and 9 s) (Belin et al., 1999). For instance, Formisano et al. (2003) found two mirror-symmetric representations of the sound frequencies. These maps move from the posterior-medial regions of the Heschl's sulcus (HS) to anterior-lateral locations along the first transverse sulcus. Formisano et al. (2003) and Upadhyay et al. (2007) used the same range of frequencies 0.3, 0.5, 0.8, 1, 2, and 3 kHz with a sound intensity set to approximately 70-dB sound pressure level for each subject. While Formisano et al. (2003) used binaural 3 s sine tones pulsed with an 8 Hz square wave, Upadhyay et al. (2007) used 1.5 s pure tones. In both studies, a single shot echo planar imaging (EPI) sequence with TR of 20 s (Formisano et al., 2003) and 25 s (Upadhyay et al., 2007) was collected with an auditory stimulus being presented at 8 variable offsets.

In contrast, Da Costa et al. (2011), Humphries et al. (2010), Langers and van Dijk (2011) and Woods et al. (2009) presented mirror-symmetric tonotopic fields along an anterior-posterior axis of the HG, while the lateral part of HG occupied by lower frequencies. Higher frequencies regions were located posterior and anterior to HG. Similar mirror-symmetric tonotopic patterns were demonstrated in both hemispheres. Results of these studies were inconsistent with older studies such as Formisano et al. (2003) and Seifritz et al. (2006) which showed that the gradients of those two fields are parallel to HG. The authors also found that the central region of the auditory cortex shows larger differences between preferred and nonpreferred frequencies while lateral parts present smaller differences (Woods et al., 2009, Humphries et al., 2010). Humphries et al. (2010) used a sparse temporal sampling method (Hall et al., 1999) with TR of 9.6 s to present 8 s trains of complex tone bursts with six centre frequencies: 200, 400, 800, 1600, 3200, and 6400 Hz. In contrast, Woods et al. (2009) used both continuous (TR = 2.9s) and sparse sampling (TR = 10.8 s) to present tone patterns of 750 ms that consisted of three different 250 ms tones of different frequency (225, 900, and 3600 Hz). In addition, the sparse sampling design was used in nonhuman primates (Petkov et al., 2006, Petkov et al., 2009). The authors presented 11 fields containing neurons tuned for the frequency of sounds where three of them are located in the primary auditory cortex.

Most of the previous studies (Formisano et al., 2003, Talavage et al., 2004, Woods et al., 2009, Humphries et al., 2010) suggested that human auditory cortex conforms to the general primate model. In this model, two mirror-symmetric tonotopic maps are located in the primary auditory core and belt areas. This model will be examined on participants later in this thesis.

# 2.4.2 Hippocampus

In humans, the hippocampus is a major component of the brains that belongs to the limbic system and located in the medial temporal lobe. It is responsible for some crucial functions e.g. long-term memory and spatial navigation (Watanabe et al., 2008). It has a curved tube (sea horse) shape with larger anterior-inferior

portion near the base of the temporal lobe than the posterior-superior part at the top (Tamraz and Comair, 2000, Andersen, 2007).



**Figure 2.4** Sagittal and coronal views of 3–D shape reconstruction of the right hippocampal white matter (WM)/GM subfields (subject C17) segmented subfields was imported from freesurfer 5.1 and reconstructed for illustration purpose using BrainVoyager QX 2.3.

The hippocampus consists of the hippocampus proper (HP) (which has four fields named CA1–CA4) and the gyrus dentatus (GD) (Duvernoy, 2005) and the subicular complex (prosubiculum, subiculum, presubiculum, and parasubiculum) as shown in Figure 2.4 (Lopes da Silva and Arnolds, 1978). The HP is connected the rest of the temporal lobe at the subfield CA1 via the subiculum. CA1 has scattered triangular pyramidal somata and connected to CA2 that, composed of large, ovoid and densely packed somata. CA3, which connect between CA2 and

the area dentate (involving CA4 and the GD), has similar pyramidal somata with less density comparing with CA2. It is connected to the area dentate via fine, nonmyelinated fibres named mossy fibres. Somata in the area dentate are ovoid, large, less dense and scattered (Duvernoy, 2005, Andersen, 2007).

The subiculum is connected to CA1 through prosubiculum at one side and to the presubiculum at the other side. The presubiculum, which has denes, small and superficial pyramidal neurons, is connected to the parasubiculum, which passes around the margin of the parahippocampal gyrus to the Brodmann's area 28 (Duvernoy, 2005, Andersen, 2007).

The fimbria is a major input-output pathway of the hippocampus that contains accumulation of myelinated axons (Duvernoy, 2005, Andersen, 2007).

# 2.4.3 Inferior parietal lobule

The inferior parietal lobule (IPL) lies below the horizontal segment of the intraparietal sulcus, superior to the lateral fissure and behind the lower part of the postcentral sulcus. It consists of three regions: the supramarginal gyrus (Brodmann's area 40), the angular gyrus (Brodmann's area 39) and the posterior parietal cortex, separated by two intermediate sulci. The SMG arches over the upturned end of the lateral fissure; anterior to the postcentral gyrus, and posterior to the STG, the angular gyrus, arches over the posterior end of the superior temporal sulcus (Tamraz and Comair, 2000) (Figure 2.5). The total cortical thickness of the SMG varies between 3.0–3.5 mm. Its layer I is rich in cells

#### Chapter 2 The Physiology of the Auditory Pathway, Pitch Processing and Pitch Memory in Human Brain

particularly in the superficial sublayer. It has extremely thick II layer with small true granule cells, whereas layer III contains numerous small cells, mostly medium-sized pyramidal cells with large pyramidal cells sporadically disseminated in the deeper zones of this layer. Layer IV is very thick with dens cells in the middle of the layer, whereas layer V is relatively thin with a compact arrangement of its pyramidal cells with spindle cells of layer VI. The border between layer VI and the WM is very sharp and clear. In contrast, the angular gyrus is thinner than the SMG with a variation of cortical thickness from 2.9 to 3.3 mm. The cellular structure is very similar to the STG, with a tendency for the deeper sublayers of the overall thinner layer III to become magnocellular (Economo and Triarhou, 2009).



**Figure 2.5** Axial and surface views of the inferior parietal lobule including both two main regions the SMG (blue) and the angular gyrus (red) (subject C13). 3D mesh reconstructed for illustration purpose using BrainVoyager QX 2.3.

The IPL involved in the comprehension of language, numerical processing, the perception of emotions in facial stimuli and interpretation of sensory information . It is also associated with auditory short memory (Celsis et al., 1999, Gaab et al., 2003, Rinne et al., 2009). In addition, the anterior portion of the SMG is considered to be the center for muscular sense and stereognosis (Economo and Triarhou, 2009). Stoeckel et al. (2009) demonstrated that SMG contributes in visual word recognition.

# 2.5 Plasticity in musicians

Musicians run multi function processes using multi-modal integration of auditory, motor, and cognitive tasks extensively on a daily basis during musical performance. These comprehensive demanding tasks and skills activate multiple brain regions e.g. the motor cortex, cerebellum (Langheim et al., 2002, Lotze et al., 2003, Meister et al., 2004), inferior frontal cortex, the STG and SMG (Bangert et al., 2006), Broca's area (Sluming et al., 2007a) and hippocampus (Groussard et al., 2010a, Herdener et al., 2010). Research on music perception and on musicians with high-level aptitude is an excellent tool to investigate brain plasticity (Altenmüller, 2001, Pantev et al., 2003, Zatorre and McGill, 2005, Groussard et al., 2010a, Herdener et al., 2010, Wan and Schlaug, 2010).

# 2.5.1 Structural plasticity

Differences in brain structure of musicians, in comparison to non-musicians, may indicate adaptations in response to intense and early commencement and continual

long-term practice including the repetitive rehearsal of musical skills. Several studies have reported structural differences in brain between musicians and nonmusicians. Schlaug et al. (1995a) reported that the anterior portion of corpus callosum (CC) was significantly larger in musicians compared to non-musicians with a strong correlation with the age of commencement of musical practice. Significant gender × musicianship interaction was found for anterior and posterior CC size; male musicians had a larger anterior CC than non-musicians (Lee et al., 2003). Schlaug et al. (2009a) tested their hypothesis on the effect of instrumental music training in the size of particular subareas of the CC using diffusion tensor imaging (DTI) on a sample of 31 children aged 5-7, that was divided into three groups: high-practicing, low-practicing, and controls. Increased CC area was found after an average of 29 months of observation in the high-practicing group in the anterior midbody of the CC. In a longitudinal study by Hyde et al. (2009), musically trained children (15 months of musical training in early childhood) showed significant structural brain differences in CC, right precentral gyrus (primary motor area) and right HG compared with matched control group. Additionally, Bangert and Schlaug (2006) tested instrumental effect on string and keyboard players and found that string players display a prominent omega shape (Yousry et al., 1997) of the central sulcus on the RH only, the keyboard player shows a left more than right prominence of the omega shape.

Hutchinson et al. (2003) measured the volume of the cerebellum in musicians (keyboard players) and non-musicians and showed that male musicians have larger cerebellar volumes than matched non-musicians where the relative

cerebellar volume in the male musician was correlated with lifelong musical experience. Using diffusion tensor imaging, Bengtsson et al. (2005) investigated effects of extensive piano practice on WM across the life span. The authors found positive correlations between practicing and fibre tract organization in different regions for each age period (childhood, adolescence and adulthood). Also the authors found a highly organised fibre tract in the right posterior internal capsule in pianists compared to non-musicians. Another study by Abdul-Kareem et al. (2011a) presented that musicians have larger WM volume of the whole right cerebellum, and specifically larger volume in right middle cerebellar peduncles and increased right superior cerebellar peduncles volume and number of streamlines.

Han et al. (2009) investigated both GM distribution and WM integrity in pianists and non-musicians. The authors demonstrated that musicians have higher GM density in the left primary sensory-motor cortex and right cerebellum and higher WM integrity in the right posterior limb of the internal capsule. It has been shown that musicians presented a larger right motor cortex than non-musicians with a negative correlation between the size of motor cortex of both hemispheres and age at which the person began musical training (Amunts et al., 1997).

Gaser & Schlaug (2003) used voxel based morphometry (VBM) analysis to investigate whole brain structural differences and found increased GM density in motor, auditory and visuospatial cerebral areas in musicians. Another study by Sluming et al. (2002) found that VBM identified an increased GM in Broca's area in the left inferior frontal gyrus in musicians with a positive correlation between

years of playing and this increase in GM distribution in under-50-year-old musicians. Abdul-Kareem et al. (2011b) showed similar results to a previous study using manual morphometry (stereology) on the same data.

In a morphologological study of HG, Schneider et al. (2002) reported that the GM volume of the anteromedial portion of HG (amHG) was 130% larger in professional musicians as compared to non-musicians. Additionally, the GM volume of amHG was highly correlated with musical aptitude in professional musicians and amateur musicians. Musicians with perfect pitch revealed increased leftward asymmetry in PT than non-musicians or musicians without perfect pitch (Schlaug et al., 1995b). Bermudez et al. (2009) found greater cortical thickness in musicians with peaks in superior temporal and dorsolateral frontal regions with right-lateralized focus of greater GM concentration in musicians centred on the posterolateral aspect of HG. Aydin et al. (2005) reported significant between musicians and non-musicians in the neurometabolite concentrations in the left PT using quantitative proton MR spectroscopic.

VBM was performed to investigate the structural differences of the hippocampus between musicians and non-musicians. It has been reported that GM density of the hippocampus was higher in musicians than in non-musicians, which indicates that musical expertise critically modifies long-term memory processes (Groussard et al., 2010a). Sluming et al. (2007b) found a significant gender by musicianship interaction in the hippocampal volumes. Male musicians had larger GM concentration in the posterior portion of the right hippocampus (RHipp) and larger

hippocampal volumes bilaterally compared to male non-musicians. Authors showed that hippocampal volume correlated positively with number of years of orchestral playing in male musicians under 50 years of age. In addition, male musicians performed significantly better on visual memory tasks than nonmusicians, and this was significantly positively correlated with the right hippocampal volume.

Using VBM, it has been reported that male musicians with absolute pitch (AP) were more leftward lateralized in the anterior region of the PT than male non-AP musicians (Luders et al., 2004). AP musicians showed heightened connectivity in WM between auditory perception and association cortices predominantly in bilateral superior temporal regions (Loui et al., 2011).

Halwani et al. (2011) used DTI to measure tract volume and fractional anisotropy of the arcuate fasciculus, which connects temporal and frontal brain regions, in musicians (singers and instrumentalists) and non-musicians. Musicians had larger tract volume and higher fractional anisotropy values of the right and left arcuate fasciculus than non-musicians.

# 2.5.2 Functional plasticity

Differences in functional cerebral characteristics between musicians and nonmusicians have been investigated using different modalities. Elbert et al. (1995) used MEG to study somatosensory evoked magnetic fields in string players. The authors found that the cortical representation of the digits of the left hand was

larger in musicians than in controls with no differences in the right hand. Also the authors reported that the cortical reorganization of the representation of the fingers increased with earlier musical training. Transcranial magnetic stimulation was used to measure motor evoked potentials (MEP), corticospinal projections (CP) curve and short-latency intracortical inhibition (SICI) curve. The authors applied paired associative stimulation with interstimulus interval of 25 or 10 ms to test change of synaptic effectiveness. Musicians showed steeper curve of MEP and SICI with greater amplitudes and slope at both interstimulus intervals (Rosenkranz et al., 2007).

MEG has been used to investigate processing of multisensory stimuli in professional trumpet players. When stimulating the lip, the authors found that trumpet players showed a positive peak at 33 ms, which was not found in the control group (Schulz et al., 2003).

Using MEG, Pantev et al. (1998) investigated the evoked magnetic fields of a pseudorandom sequence of piano and pure tones matched in frequency and loudness in musicians and non-musicians. The only different was in the strength of cortical activation for piano tones, which was 25% greater in musicians. Musicians also showed a strong correlation with the age of commencement of musical practice. Schneider et al. (2002) used MEG to compare the processing of sinusoidal tones in the auditory cortex of non-musicians, professional musicians and amateur musicians. The authors found neurophysiological differences between groups. Comparing professional musicians and non-musicians, the

activity evoked in primary auditory cortex (19–30 ms after stimulus onset) was 102% larger in professional musicians. Pantev et al. (2001) showed that musical education and training is reflected in the organization of auditory and somatosensory representational cortex in musicians. The intensive experience of musicians with musical instruments formed representations for musical instruments in and adjacent to auditory association cortex (Hoenig et al., 2011).

Rüsseler et al. (2001) used mismatch negativity (MMN) to investigate the evoked potentials component in auditory cortex of musicians and non-musicians. The authors suggested that temporal integration might be more precise in musicians since musicians showed MMN for regularly spaced tones mistimed by 20 milliseconds, where non-musicians needed longer than 50 milliseconds. Lappe et al. (2008) reported that sensorimotor-auditory training causes plastic reorganizational changes in the auditory cortex over and above changes introduced by auditory training alone. In this study, the authors investigated the difference in MMN response between 'sensorimotor-auditory' group who, learned to play a musical sequence on the piano and 'auditory listened to' group and made judgments about the music that had been played by participants of the sensorimotor-auditory group. The authors showed that first group had significantly enlarged response.

Using fMRI Bangert et al. (2006) investigated cortical auditory motor coupling during the acoustic and the mute motion-related tasks in professional pianists. The professional pianists showed increased activity compared to the non-musicians in a distributed cortical network during both tasks. Applying conjunction analysis showed that either task type coactivated a distinct musicianship-specific network comprised of dorsolateral and inferior frontal cortex (including Broca's area) the STG, the SMG, and supplementary motor and premotor areas. Sluming et al. (2007a) reported that professional musicians showed enhanced performance on a visuospatial task compared to non-musicians, associated with increased activation in Broca's area.

In another study, fMRI has been used to examine developmental aspects and effects of musical training on three different groups; 10-year-old children with varying degrees of musical training, adults non-musicians, and adult musicians. Both adults groups showed larger activations than children in prefrontal and temporal areas, and in the SMG. Musical training was correlated with stronger activations in the frontal operculum and the anterior portion of the STG in both adults and children (Koelsch et al., 2005).

It has been shown that changes in functional organization also extend to subcortical sensory structures. Musacchia et al. (2007) demonstrated that musicians had earlier and larger auditory and audiovisual brainstem responses to speech and music stimuli with a positive correlation between years of musical practice and strength of brainstem response to speech stimulus. Groussard et al. (2010a) and Herdener et al. (2010) identified plastic capabilities of the hippocampus in musicians. Herdener et al. (2010) observed enhanced neural responses to temporal novelty in the anterior left hippocampus in professional musicians. The authors also reported enhanced neural responses to temporal novelty in music students with one-year intensive aural skills training. They also found that musical abilities boost hippocampal sensitivity to temporal novelty. Groussard et al. (2010a) found that musical expertise induced supplementary activations in the hippocampus, medial frontal gyrus, and superior temporal areas on both sides. The authors suggested that there is a broad combination of the neural networks involved in episodic and semantic memory.

Brain activation patterns of pitch memory task were investigated in musicians using fMRI. Musicians and non-musicians showed bilateral activation of superior parietal lobe, the STG, SMG, posterior middle gyrus and inferior frontal gyrus. Musicians showed more activation in the right temporal and SMG, while more activation was shown in right primary and left secondary auditory cortices in musicians. The authors suggested that musicians prefer to use brain regions specialized in short-term memory while non-musicians rely more on brain regions that important for pitch discrimination (Gaab and Schlaug, 2003). Gaab et al. (2006) examined functional activation patterns of an auditory working memory task prior to and after training. The training group was divided into Strong-Learners (SL) and Weak-Learners (WL), with no significant functional or structural brain differences prior to training. Pre-post training comparisons showed a significant signal increase in the left HG, the left posterior superior temporal and SMG for the SL group, while the WL group showed significant signal increases in the left HG, anterior insular cortex and in a lingualorbitofrontal-parahippocampal network. Applying a random effects analysis

comparisons between both groups showed increased activation only in the left SMG but not in HG. It has been suggested that the activation of the SMG reflects its importance in the short-term storage of auditory material (Gaab et al., 2006).

#### 2.5.2.1 Plasticity of the auditory cortex and tonotopic reorganisation

Early experience of sounds with spectral patterning develops the tonotopic mapping of auditory cortex. Many studies demonstrated evidence for this hypothesis. For example, Harrison, Stanton et al. (1993) demonstrated that, distorted tonotopic organization is caused by induced high-frequency hearing loss in neonatal kittens with expanded cortical space devoted to frequencies near the frequency of the hearing loss. Early intensive musical experience has a great effect on brain development (Shahin et al., 2004). However, it is possible to obtain musical expertise later in life. These could change the number of neurons involved in processing, the timing of synchronization and the number and strength of excitatory and inhibitory synaptic connections. However, musical training does not provide a specific change in the responses to noise stimuli (Fujioka et al., 2006). In animals, different studies showed that frequency reorganisation can be induced after exposing to noise (Nakahara et al., 2004, Eggermont, 2005, Pienkowski and Eggermont, 2011) or by training (Recanzone et al., 1993), by mechanical damage to the cochlea (Robertson and Irvine, 1989), by exposing to environmental contaminate (Kenet et al., 2007) and by electrical stimulation of the auditory cortex (Valentine and Eggermont, 2003, Eggermont, 2005). In humans, some population sowed expanded or altered tonotopic organisations such as; blind people (Elbert et al., 2002), schizophrenia patients (Rojas et al., 2002), people

who have develop high-frequency hearing loss from acoustic trauma or sudden hearing loss (Dietrich et al., 2001) and following stabilized changes of the acoustic input entering the central nervous system (Tecchio et al., 2000). In addition, Suga and Ma (2003) found an enlarged representation of the frequency of stimulus. The authors applied a microstimulation in primary auditory cortex of gerbils and echolocating bats because best frequency (BF) of neurons surrounding the stimulation site shifted towards that of the neurons at the stimulation site. These lead to a suggestion of auditory discrimination training. For example, reorganization of the auditory cortex can be approached by training tinnitus patients to discriminate some features of acoustic stimuli that close to the tinnitus frequency (Muhlnickel et al., 1998, Flor et al., 2004). Moreover, both high frequency and low frequency environmental sounds conferred benefit on reorganisation compared with normal acoustic environment. However, high frequency sound is more effective in reducing hearing loss and map reorganization (Eggermont and Roberts, 2004). In contrast, Flor et al. (2004) demonstrated some limitations in their study. Firstly, short training period only provides an initial evaluation of the effects of the training. Secondly, The authors did not collect physiological data, which could give an indication of the specific effects of the training on auditory cortex. Third, the role of nonspecific effects in the alteration of tinnitus severity cannot be determined because they did not employ a control group. Finally, there was an overestimating of the true effects of the training, which could be because corrections were not applied on computation of multiple tests (Flor et al., 2004). Two recent studies approved that auditory

discrimination training could partially reverse the wrong changes in tonotopic representation and improve tinnitus (Herraiz et al., 2007, Herraiz et al., 2009). In addition, it have been reported that, rapid changes can occur in the tuning of neurons in the auditory cortex of adult human following manipulation of the acoustic environment (Pantev et al., 1999) and tonotopic maps change happened during paying selective attention to differences in pitch of the tones (Ozaki et al., 2004). Also, intensive short-term training of pitch discrimination results in a change in the hemodynamic response of the human auditory cortex (Jäncke et al., 2001).

# **Chapter 3 : The Technology of Functional Neuroimaging and Experimental Paradigms Optimisation**

## 3.1 Aims

Specific aims of this chapter were to:

1- Consider the technological aspects in investigating tonotopic mapping in the human auditory cortex using fMRI and MEG.

2- Optimise the experimental designs.

# **3.2 Introduction**

During the last two decades, advances in neuroimaging technologies have provided enormous information about brain structure and function. Exploiting the superior temporal resolution of EEG/MEG and reasonably good spatial resolution of fMRI could improve our understanding of the neural mechanisms that involved in various sensory and cognitive processes. FMRI measures local changes in brain hemodynamics induced by cognitive or perceptual tasks with high spatial resolution. Conversely, electrophysiological methods M/EEG measure instantaneously the current flows induced by synaptic activity.

# 3.3 Magnetic Resonance imaging (MRI)

MRI is one of the most important medical advances during last few decades. It is the most preferable system in neuroimaging because of: minimal invasiveness, no radioactivity, widespread availability, and virtually unlimited study repetitions. It also provides anatomical and functional information for brain regions. Historically, Paul Lauterbur used back projection technique to demonstrate MRI in 1973. In 1974, Sir Peter Mansfield's group invented selective excitation technique. One year later Richard Ernst's group in Zurich invented twodimensional Fourier transform imaging. In 1980, Edelstein and Hutchison developed the first practical 2D FT imaging method (McRobbie, 2003).

## 3.3.1 Magnetic components and larmor frequency

MRI uses property of nuclear magnetic resonance (NMR) to image nuclei of atoms inside the body. Because water and fat are the biggest source of protons in the body, NMR depends on the spinning of hydrogen atom (a single proton). Spinning protons produce small magnetic fields in random distribution. Applying an external magnetic field, make protons align their spin states either with or against the external magnetic field ( $\mathbf{B}_0$ ) as shown in Figure 3.1. Protons obey the Botzmann statistics and therefore majority of protons aligns with the magnetic field, the total magnetic field of the excess protons is called net magnetization vector (NMV) ( $\mathbf{M}_0$ ). The spinning protons precess about the axis of the external field with an angular frequency ( $\boldsymbol{\omega} \boldsymbol{\theta}$ ) proportional of the strength of the magnetic field (McRobbie, 2003) as shown in larmor equation (3.1).

$$\boldsymbol{\omega}\boldsymbol{\vartheta} = \boldsymbol{\gamma} \, \mathbf{B}_0 \qquad 3.1$$

 $\omega \sigma$  = Angular frequency  $\gamma$  = Gyromagnetic ratio  $\mathbf{B}\sigma$  = External magnetic field in tesla



**Figure 3.1** Precession of the protons and their alignments in the magnetic field. a) Precession of protons (hydrogen atom), b) Spinning proton, c) Spinning proton with no magnetic field present, d) Spinning proton if external magnetic field applied.

Applying electromagnetic radio frequency (RF) pulse cause protons to jump between high and low energy states which result in a change in the NMV of protons inducing a voltage in a coil of wire perpendicular to that field as identified by Faraday's law (McRobbie, 2003).

# 3.3.2 Relaxation times

The absorbed RF energy is retransmitted at the angular frequency by turning off RF in order to produce the NMR signal. The excited protons begin to return to the relaxation state where energy absorbed by the excited protons is released back into the surrounding lattice in a process called T1 relaxation, which mostly used in anatomical images. In this configuration, Z component of magnetization  $M_Z$  equals  $M_0$  with no transverse ( $M_X$  or  $M_Y$ ) magnetization. The time that it takes for the system to return to relaxation state is described mathematically by an exponential curve (equation 3.2 and Figure 3.2). This recovery rate is characterized by the time constant T1, which is unique to every tissue depending on water amount in the tissue (Westbrook, 2002, McRobbie, 2003, Bernstein et al., 2004).

$$M_Z = M_0 (1 - e^{-t/T_1})$$
 3.2



Figure 3.2 T1 recovery curve and T2 decay curve, T2 is much quicker than T1.

The second relaxation type depends on de-phasing of the excited protons which is called T2 and T2<sup>\*</sup> relaxation. The time that it takes for the system to return to equilibrium of the transverse magnetization is described mathematically by an exponential curve (Westbrook, 2002, McRobbie, 2003, Bernstein et al., 2004) (see equation 3.3 and Figure 3.2).

$$M_{XY} = M_{XY_0} e^{-t/T^2}$$
 3.3

Usually, protons spin at different rates and hence they get out of phase leading to NMR signal decay depending on interacting magnetic field. Magnetic fields of two nearby protons interact with each other (spin-spin interaction). The temporary and random interaction between two protons causes a cumulative loss in phase resulting in an overall loss of signal which known as T2 relaxation (Westbrook, 2002, McRobbie, 2003, Bernstein et al., 2004).

#### 3.3.3 Blood oxygenation level dependent (BOLD) fMRI

Generally, BOLD fMRI measurements reflect the hemodynamic changes related to the input and intracortical processing of a given area rather than its spiking output (Logothetis et al., 2001, Goense and Logothetis, 2008). This technique, which makes use of blood as an intrinsic contrast, has functional sensitivity in MRI. Gradient echo, which was introduced in 1992 and based on  $T2^*$  is the most widely used BOLD sequence in fMRI because of its sensitivity to the magnetic field inhomogeneity. Excitation of neurons leads to a local increase in energy and oxygen consumption in stimulated areas. This makes blood flow increases so much, that the venous and capillary concentration of deoxyhemoglobin actually decreases with an increase in oxyhaemoglobin, This process causes an increased field homogeneity and thus a higher signal intensity in a series of T2<sup>\*</sup> weighted images. Change of the oxygenated blood flow to/from a region in the brain cause an increase in MR signal by increasing intravoxel T2<sup>\*</sup>. This leads to the image intensity difference between the rest and the task for active regions of the brain (McRobbie, 2003, Stippich et al., 2007).

#### 3.3.4 Study design considerations

FMRI of the auditory system has been regarded as challenging because of the loud gradient switching noise. Scanner noise not only masks the presentation of the stimulus and affects the perception of the auditory stimulus, but also the repetitive nature of scanner noise determined by repetition time (TR) could cause activation of the auditory cortex (Bandettini et al., 1998, Ulmer et al., 1998, Shah et al.,

1999, Hall et al., 2000, Shah et al., 2000). Scanner acoustic noise is mainly produced by the mechanical oscillation of the gradient coils placed in a magnetic field. Switching the electrical current on and off through the gradient coils induces Lorentz forces and makes the coils move.

Different factors affect this displacement such as; voltage applied, strength of the static magnetic field, the amplitude frequency and waveform of the switching (Hedeen and Edelstein, 1997, Mansfield et al., 1998, Shellock et al., 1998, Moelker and Pattynama, 2003). Various techniques have been implemented to eliminate the actual sources of scanner acoustic noise. Passive noise reduction methods such as; earplug, earmuff, and helmet; are the most widely used methods (Amaro et al., 2002, Moelker and Pattynama, 2003). Combination of these methods could effectively reduce acoustic noise by up to 60 dB (Moelker and Pattynama, 2003). In addition, reengineering MR hardware by incorporating the gradient coils in vacuum enclosures (Katsunuma et al., 2002) combined with a magnet with a non conducting inner bore cryostat (Edelstein et al., 2002) or multiple active gradient shields (Edelstein et al., 2005) effectively interrupt acoustic noise propagation. MR systems with vacuum enclosed gradient and insulators have become commercially available for a decade (Excelart, Toshiba Corporation, Tochigi, Japan, and Signa Twinspeed, General Electric, Milwaukee). Toshiba has implemented a breakthrough technology called pianissimo which leads to an acoustic noise reduction more than 90% of conventional MRI systems (Vantage, Toshiba Corporation, Tochigi, Japan).

#### Chapter 3 The Technology of Functional Neuroimaging and Experimental Paradigms Optimisation

Alternatively, several image acquisition techniques have been studied to overcome the confounding effect of scanner noise such as; sparse temporal sampling (Hall et al., 1999), clustered volume acquisition (Talavage et al., 1999), behaviour-interleaved acquisition (Eden et al., 1999), silent simultaneous multislice excitation (SIMEX) (Loenneker et al., 2001) and silent event related (SER) (Yang et al., 2000, Le et al., 2001, Yetkin et al., 2003). In addition, Schmidt et al. (2008) optimized low-noise EPI for auditory fMRI measurements using a block design. This design produces a narrow-band acoustic frequency spectrum therefore; the authors found that the acoustic noise reduction amounts to up to 20 dB comparing to standard EPI sequence.

Hall et al. (1999) used a long repetition time TR (14 s) with alternating intervals of 14 sec of speech and 14 sec of silence. This method shows a greater MR signal change than continuous scanning. The response of the primary auditory cortex peaked 4-5 s after stimulus onset and decayed after an additional 5-8 s (Hall et al., 1999, Hall et al., 2000). Baumann et al. (2010) studied characterisation of the BOLD response and reported that the BOLD response peaked around 4 s after the onset of 2 s stimuli. In contrast, Belin et al. (1999) found that activation peaked approximately 3s after stimulus onset and lasted for 3s. The authors used shorter effective TR (10 s) and varied the delay between a short auditory stimulus and the MR acquisition. Although sparse imaging shows greater auditory response, continuous imaging could better reveal organizational properties such as aspects of the hierarchical organization of auditory cortex (Petkov et al., 2009).

In clustered volume acquisition (Talavage et al., 1999), slice acquisitions are clustered at the end of TR period to prevent a response from being induced by the volume acquisition itself. Extra gradient readouts are introduced to detect the cortical responses to the scanner noise where acoustic stimulation is presented immediately prior to the acquisition of a cerebral volume. The number of extra gradient readouts controls the duration of acoustic stimulation. This method yields greater measures of dynamic range of the signal change (Talavage et al., 1999).

Eden et al. (1999) used behaviour-interleaved acquisition for non-auditory tasks and showed that it would be useful for the studies involving auditory stimulation. The authors delayed data acquisition to a period immediately after task completion, to utilize the physiological delay in order to disperse between neuronal activity and its resulting hemodynamic lag.

Loenneker et al. (2001) implemented SIMEX pulses to improve the intrinsic low volume coverage of the silent sequence and compared it with a standard noisy technique using auditory and visual stimulation paradigms. The scanner noise was reduced below the range of the ambient noise of the magnet room during the silent part of the experiment, while there was a trend of decreased activated areas for the noisy part.

SER design limits image acquisition to a single volume following an auditory stimulus administered during the 'silent' period. This acquires an image at a point close to the peak of the HRF; 5 s (Le et al., 2001) or 4 s (Yetkin et al., 2003) after stimulus onset or even two time points in the peak period of the stimulus induced

HRF curve and also in the baseline period (Yang et al., 2000). Compared to other designs (block and conventional event related), SER shows larger MR signal change (Yang et al., 2000) and a greater signal to noise ratio (SNR) (Amaro et al., 2002). The main disadvantage of this design that it needs a longer total acquisition time because of its longer TR -at least 20 s for the best optimization- (Amaro et al., 2002, Hall et al., 2000, Le et al., 2001) therefore, it is more susceptible to motion artefacts (Le et al., 2001). It mostly depends on two time points at the peak and baseline of HRF or even one time point at the peak therefore, resultant images will lack temporal information (Amaro et al., 2002). Another disadvantage is that the HRF is excessively affected by medications (Seifritz et al., 2000), age (Yousem et al., 1999), and vascular diseases (Rother et al., 2002), while the peak time for the HRF is estimated by assuming that it will be constant across all subjects and trials. Introducing a jittering scheme overcomes the lack of temporal information. This method permits full sampling of HRF curve by collecting data at different time points on the HRF curve at jittered intervals post stimulus manner (Josephs et al., 1997, Belin et al., 1999, Hall et al., 2000, Amaro et al., 2002), resulting in a total acquisition time which is even longer than that of normal SER. Furthermore, averaging of time points combines the signals of correct and erroneous trials.

#### **3.4 MEG**

MEG is a neuroimaging technique that is extremely sensitive to the minimal changes in the magnetic fields produced by small changes in the electrical activity within the brain. Usually, information is transferred between neurons as electric

currents, which in turn produce magnetic fields, which are detectable outside the head. It is a direct and real time measurement of neural activity (Hansen et al., 2010).

## 3.4.1 Electrophysiological Basis of MEG Signals

A stimulus activates part of the cortex and hence induces time-varying electrical currents at the level of cellular membranes. These currents consist of an action potential and postsynaptic potential (PSP). The contribution of the action potential to MEG signals is minimal to that of compared to that of PSP, due to its short duration ( $\approx$ 1ms) and because action potential attenuates much more strongly than PSP. The PSP can be classified into an excitatory postsynaptic potential (EPSP) and inhibitory postsynaptic potential (IPSP). Detectable MEG signals are produced by EPSP and partially by IPSP (Niedermeyer and Silva, 2005). Here we will consider two types of neurons: pyramidal and stellate. Pyramidal neurons have long apical dendrites that help in generating coherent magnetic fields when activated with a certain degree of synchrony. The cortex is folded into gyri and sulci. Therefore, apical dendrites of some populations of neurons are roughly perpendicular to the overlying skull (gyral neurons), whereas others are roughly parallel to the skull, i.e., (sulcal neurons). Sulcal neurons produce currents that have a component oriented tangentially to the skull, while gyral neurons produce radial currents (Figure 3.3). Tangential currents produce detectable magnetic field by MEG because their magnetic fields are perpendicular to the skull (Hämäläinen

et al., 1993, Vrba and Robinson, 2001, Lu and Kaufman, 2003, Hansen et al., 2010).



Figure 3.3 Origin of MEG signal.

#### 3.4.2 MEG measurement

The magnetic fields that generated by brain are very weak (in the range of several tens to hundreds femto tesla) (Ono and Ishiyama, 2008), which only can be detected by a highly sensitive magnetic detector that named superconducting quantum interference devices (SQUIDs). Environmental noise is about a factor of 1 million to 1 billion larger than the MEG signals, which need very accurate noise cancellation methods. Different approaches have been used to isolate MEG signal. The most effective and important method is employing a passive magnetically shielded room. Another technique is using active noise compensation systems by measuring the interference field and generates a compensating field. Further noise

#### Chapter 3 The Technology of Functional Neuroimaging and Experimental Paradigms Optimisation

reduction techniques have to be applied inside the shielded room; gradiometrization (using multiple magnetometers) is one of them. The gradiometers protects the MEG sensors from far-away magnetic sources. In addition, unwanted magnetic sources can be measured explicitly and then subtracted from the signals. There are different types of magnetic sensors have vary sensitivity to distant sources. Magnetometers are the most sensitive sensor to the depth of sources, followed by first-order axial gradiometers, second-order gradiometers and, finally, planar gradiometers. Planar gradiometers detect less artefacts and have maximum sensitivity to superficial cortical sources (Hämäläinen et al., 1993, Malmivuo and Plonsey, 1995, Vrba and Robinson, 2001, Lu and Kaufman, 2003, Hansen et al., 2010).

#### **3.4.3** Source localisation

Localising electric activity within the brain that induces magnetic fields outside the head is a great challenge in MEG. Source localization can be done by computing what the output of the MEG sensors would be if a certain region of cortex were active "forward problem" or by estimating the location of the activity from MEG signal "inverse problem". The forward problem determines the magnetic fields that result from primary current sources, while the inverse problem estimates the location of these primary current sources. Although, solving the forward problem show diversity of approaches that have been applied, it is crucial for performing source localisation (Yaroslav O. Halchenko et al., 2005). In the inverse problem, there are infinite possible answers, which introduce
the difficulty of defining the best solution. Therefore, prior knowledge of brain activity and anatomy are needed. The source models can be either overdetermined or under-determined. The over-determined estimation assumes that a single current dipole or several current dipoles exist in a head conductor. These models are suitable for a primary sensory response. However, in the case of a complicated neuromagnetic field e.g. spontaneous or high cognitive function, under-determined estimation is required considering multiple or widely distributed cortical areas. The under-determined solution can be divided into norm estimation and spatial filter models (Hämäläinen et al., 1993, Lu and Kaufman, 2003, Hansen et al., 2010). Additionally, time and frequency amplitude analysis, coherence analysis and three-dimensional mapping of source power are considered as solution techniques as well (Vrba and Robinson, 2001).

# 3.5 Combining fMRI and MEG

Although, neuroimaging is a very young science, multimodal neuroimaging has already advanced our understanding of the spatiotemporal pattern of brain activation and connectivity underlying perception, motion and cognition. MEG and fMRI are innovative functional brain imaging techniques. MEG signals are directly coupled to neuronal electrical activity (Hämäläinen et al., 1993, Dale et al., 2000). It reflects the magnetic field resulting primarily from synaptic transmembrane currents in neuronal dendrites. In contrast, fMRI signals are limited by the indirect nature of the coupling between the observed signals and the underlying neuronal electrical activity (Dale and Buckner, 1997). MEG has good

#### Chapter 3 The Technology of Functional Neuroimaging and Experimental Paradigms Optimisation

temporal resolutions with poor spatial resolution whereas fMRI has good spatial resolution with poor temporal resolution due to the limited rate of change in the hemodynamic response. These two modalities present different views of the neural activity; therefore integrating them would improve our understanding of brain activity. It has been shown that BOLD contrast mechanism reflects the synaptic processing of a given area rather than its spiking output (Logothetis et al., 2001, Goense and Logothetis, 2008). Additionally, Arthurs et al. compared the evoked EEG potential and the fMRI response to somatosensory stimulation and found a similar increase in the somatosensory evoked potential amplitude and the BOLD signal with increasing stimulus intensity (Arthurs and Boniface, 2002, Arthurs and Boniface, 2003, Arthurs et al., 2007). Im et al. (2005) investigated spatiotemporal pattern of brain activation of language judgment task and found better localization of the language area on the left hemisphere in the fMRIconstrained data as compared with the distributed sources based on MEG alone. Although, tonotopic mapping is not a complex cognitive task therefore; combination of fMRI and MEG should be truly integrated analyses, it is rather a challenging process to investigate spatiotemporal components of neural activity of frequency organisation as recent studies showed multiple tonotopic fields in time (Liegeois-Chauvel et al., 1991, Liegeois-Chauvel et al., 1994, Pantev et al., 1995, Schneider, 2001, Schneider et al., 2002, Lütkenhöner et al., 2003a) and space (Talavage et al., 2000, Formisano et al., 2003, Seifritz et al., 2006, Woods et al., 2009, Humphries et al., 2010, Da Costa et al., 2011, Langers and van Dijk, 2011).

# 3.6 Experimental Design optimisation

In the pilot study, I investigated and optimised different designs to examine tonotopic mapping in the human auditory cortex using both fMRI and MEG. Also I used different groups of frequencies.

# 3.6.1 Acoustic stimulation

I used different auditory stimuli with nine frequencies (130.8, 164.81, 207.65, 523.25, 659.26, 830.61, 2093, 2637 and 3322.4 Hz) from three notation groups (C, E and G#/A  $\triangleright$ ) in the form of an octave interval. Sound pressure level of the stimulus was set to 70 dB sound pressure level (SPL) for all stimuli and for every subject. Each stimulus was formed in a train of sine waves (rise/fall time 10 ms, plateau 30 ms, interstimulus interval (ISI) 50 ms).

I changed the acoustic stimulations to the following frequencies (247, 440, 784, 1175, 2637 and 4186 Hz) that allocated to the following musical notation groups (B3, A4, G5 D6 E7, C8 respectively) at the same sound level (70 dB SPL). The later group of frequencies was selected depending on the calibration process of sound delivery systems in both modalities (MRI and MEG scanners). 247 and 4186 Hz were the lowest and the highest frequencies that were presented at 60 dB SPL in MEG system. This sound pressure level was characterised and measured by Dr. Guy Lightfoot from department of audiology at the Royal Liverpool University Hospital. This extended the range of frequencies (247-4186 Hz) and reduced the overall fMRI scanning time e.g. in high-resolution stroboscopic event

related (SER), scanning time of each run was reduced from 13 minutes to 9 minutes. Therefore, this design became less susceptible to gross motion artefacts. In MEG design, more stimuli of each frequency were presented.

#### 3.6.2 FMRI experiments

In order to investigate tonotopic mapping of the human auditory cortex, five subjects were tested using four designs of fMRI; low-resolution SER, highresolution SER, low-resolution block and high-resolution block designs. All experiments were performed using a 3T Trio whole body scanner (Siemens, Erlangen, Germany) with an eight-channel phased-array head coil.

#### SER

Stimuli were presented randomly during the silent part of the image acquisition. Acquisition of the EPI data was taken in a randomised manner at 8 distinct time points (0, 3, 4, 5, 6, 7, 8 and 11 s) after the onset of the stimulus to permit sampling at different time-points on the HRF curve (Josephs et al., 1997, Belin et al., 1999). Tone duration was 3s and the scanning time was 9 minutes. This design has been optimised for two types of resolutions. In the low resolution SER, the following scanner parameters have been used; (TR) = 20 s, echo time (TE) = 35 ms; delaying TR = 18196 ms; echo spacing 0.45; imaging resolution =  $3 \times 3 \times$ 3 mm<sup>3</sup>; flip angel= 90°; number of axial slices 28; matrix size 256 × 256; bandwidth = 2605 Hz/px. However in high resolution SER, I used the following scanner parameters: (TR/TE) = 20000/35 ms; delaying TR = 17170 ms; echo spacing 0.7; phase partial Fourier = 6/8; imaging resolution =  $2 \times 2 \times 2$  mm<sup>3</sup>; flip angel= 90°; number of axial slices 28; matrix size  $256 \times 256$ ; bandwidth 1628 Hz/px.

# Block design

This design involves ON-period and OFF-period with duration of 20 sec each period. The duration of the stimulus was 500ms and the duration of rest was 500ms which means each On-period contains a repetition of each stimulus for twenty times. On-periods were repeated twice for each tone. The following scanner parameters have been used for lower resolution design; TR= 2.5 s; TE= 30 ms; matrix size =  $64 \times 64$ ; flip angle =  $90^{\circ}$ ; echo spacing 0.45; field of view  $190 \times 190$ ; number of slices = 36 and bandwidth = 2605 Hz/px with voxel size  $3 \times 3 \times 3$  mm<sup>3</sup>. Scanning session lasted for about 12 min. In contrast, the higher resolution design has the following scanner parameters; number of slices = 30; TR = 3 s; TE= 32 ms; matrix size =  $64 \times 64$ ; flip angle =  $90^{\circ}$ ; field of view  $190 \times 190$ ; number of slices = 36 and bandwidth = 1628 Hz/px; phase partial Fourier = 6/8; echo spacing 0.8; voxel size  $1.5 \times 1.5 \times 2$  mm<sup>3</sup>; matrix size  $128 \times 128$ . This session lasted for about 13 minutes.

## **3.6.3 MEG experiments**

In the first design, I used the same frequencies that have been used in fMRI experiment where every stimulus was presented 200 times with duration of 500 ms, and interstimulus interval ranging from 450 to 550 ms, this session lasts up to 39 min. In the newer design, each frequency was formed in 150 ms sine wave

(rise/fall time 20 ms, plateau 110 ms) with an interstimulus interval randomized between 300 and 500 ms. Every stimulus was presented 1000 times.

**3.7 Results** 

#### 3.7.1 FMRI results

For each subject activated regions were those regions with activated voxels above a significance of adjusted False Discovery Rate (FDR) (Benjamini and Hochberg, 1995, Genovese et al., 2002) P < 0.01. Some subjects had a strong response to the tones, with significance of adjusted FDR P < 0.001. The obtained P values were corrected for multiple comparisons using a cortex-based adjustment. Moreover, maps of the frequency and octave that evoked the highest BOLD response (best tonotopic and octave maps) were represented on folded and morphed inflated polygon meshes representing the subjects' cortical sheet (Figures 3.4 and 3.5) by implementing a three colour coding scheme; high-octave maps were represented in blue, whereas medium octave frequencies maps was represented in a yellow colour, and low octave was illustrated in red colour. I have applied the technique used by Parkes et al. (2009) for colour tuning in the visual cortex to tonotopic mapping in the auditory cortex.

Chapter 3 The Technology of Functional Neuroimaging and Experimental Paradigms Optimisation



**Figure 3.4** Best-octave maps in the auditory cortex (subject C17, P<0.01 FDR adjusted), layout maps are superimposed on a mesh reconstruction of the subject's cortex. a) low-resolution block design, b) high-resolution block design, c) low-resolution SER, d) high-resolution SER.



**Figure 3.5** Best-frequency maps in the auditory cortex (subject C17, P<0.01 FDR adjusted), layout map is superimposed on a mesh reconstruction of the subject's cortex. a) low-resolution block design, b) high-resolution block design, c) low-resolution SER, d) high-resolution SER.

Best-frequency maps of subject C17 with the newest design (6 frequencies and voxel size of  $2 \times 2 \times 2 \text{ mm}^3$ ) are presented in Figure 3.6. The obtained p value was corrected for multi comparisons using a cortex-based Bonferroni adjustment (Formisano et al., 2002).



Figure 3.6 Best-frequency maps in the auditory cortex (subject C17, P<0.01 cortex-based Bonferroni adjustment (Formisano et al., 2002)), layout map is superimposed on a mesh reconstruction of the subject's cortex.

# 3.7.1.1 Defining frequency organization

Further investigation of frequency organisation was applied on subject C17. Best frequencies maps were projected on folded and morphed inflated mesh reconstruction representing the subject's auditory cortices to check the effect of HG orientation, and its degree of convolution on frequencies gradient directions (Figure 3.7). Tonotopic maps were clearly preserved in both convoluted and inflated auditory cortex. Mirror-Symmetric Tonotopic Maps were shown in both hemispheres in the medial part of HG along an axis roughly parallel to the HG axis. First high frequencies cluster (4186 Hz "dark blue" and 2637 Hz "light blue") can be observed in the medial edge of HG. The second high frequencies cluster was located in the middle of HG. A large cluster of low to mid range

Chapter 3 The Technology of Functional Neuroimaging and Experimental Paradigms Optimisation

frequencies (440 Hz "orange", 784 Hz "yellow" and 1175 Hz "green") was located in the middle between these high frequency regions. Another cluster of (247 Hz "red", 440 Hz "orange" and 784 Hz "yellow") was situated in the lateral aspect of HG. In the left HG, there was a third high frequency region at the lateral edge of HG. Tonotopic maps were extended to a single gradient in PT in both hemispheres. In this map, high frequencies cluster was situated in the lateral part of PT and the med to low frequencies cluster was located in the medial part of PT.



Figure 3.7 The cortical best-frequency maps of six tones are displayed on mesh reconstruction and inflated representation of the subject's auditory cortices, HG = Heschl's gyrus, (subject C17, P<0.01 FDR adjusted).

# 3.7.2 MEG results

Figures 3.7 and 3.8 show tonotopic mapping of N19m-P30m and N100m components respectively. Dipoles were colour-coded using six colours (ranging from dark blue for the highest frequency to red for the lowest one). Tonotopic mapping gradients of N100m components were located along the medial-lateral axis of PT posterior to HG. I managed to localize earlier auditory evoked field components (N19m-P30m) in the HG for two frequencies in one subject only.



Figure 3.8 Source dipole localizations of N19m-P30m for the highest frequency (blue) and the lowest (red) are overlaid on fMRI tonotopic maps of same frequencies and displayed on the cortical sheet of subject # 02 in HG and shown on sagittal, coronal and transverse anatomical views.



Figure 3.9 Source dipole localizations of N100m for five frequencies are presented on a mesh reconstruction of the subject's head mesh. All sources are located in the PT (subject # 03).

# 3.8 Discussion

In this pilot study, I implemented various designs using fMRI and MEG. Several methods have been developed to examine tonotopic mapping in the human auditory cortex using various pure tones from different notes and different octaves. Regarding fMRI design, both block designs were superior in statistical power in comparing with stroboscopic event related designs. However, activities were distributed over wide area outside the region of interest. In addition, because of interface between scanner noise and sound tones, some tones were not represented e.g. tones of high frequencies with low-resolutions block designs and low frequencies with high-resolution block designs (Figure 3.5 a-b). I only ran higher-resolution design on 3 subjects because of the load on the scanner acquiring high-resolution images during block design. This produced high noise produced during scanning session. Investigating technical reasons is out of scope for this study. In contrast, stroboscopic SER provided clearer maps (see Figures 3.5 c-d and 3.6). High-resolution stroboscopic SER with six tones produced the best tonotopic maps (Figure 3.6). This could be due to shorter scanning time, which leads to less gross motion artefacts, as number of presented frequencies is less than these of older designs. Multiple tonotopic maps of the optimised design as shown in Figure 3.7 are consistent with some recent studies (Pantev et al., 1995, Schneider, 2001, Formisano et al., 2003, Seifritz et al., 2006).

In contrast, in MEG part of the study, there were clear tonotopic gradients of N100m components in three subjects only along the medial-lateral axis of PT. Tonotopic maps of auditory evoked long latency fields are consistent with some

recent studies (Pantev et al., 1995, Schneider, 2001). I localised the earlier auditory evoked field components (N19m-P30m) in the HG for two frequencies in one subject only. Instability of localising N19m-P30m could be related the fact that middle latency auditory evoked field components require a minimum number of event per stimulus (Pantev et al., 1995, Schneider, 2001).

As a consequence of this optimization study, I decided to investigate influence of musical performance expertise on tonotopic mapping and cortical activation in the human auditory cortex using fMRI and MEG. Approval for the 'musical performance' study was obtained from the Local Research Ethics Committee at University of Liverpool. Number of participants of each study is listed in the table 3.1. Table 3.2 show detailed information of each subject.

Study	Number of participants									
	Musicians				Non-musicians					
	Participated		Excluded		Participated		Excluded			
	Males	Females	Males	Females	Males	Females	Males	Females		
Auditory test	. 7	5	0	0	10	4	0	0		
SMRI	7	5	0	0	10	4	0	0		
FMRI	7	3	4	2	10	3	3	3		
MEG	7	4	3	1	11	4	6	2		

Table 3.1 The number of participants of each study.

•							
Subject's ID	Age	Handedness	Gender	fMRI	MRI	MEG	Psychoacoustic tests
M01	57	R	М	✓*	~	✓+	~
M02	42	L	F	-	~	✓+	~
M03	57	R	М	<b>~</b> *	~	~	~
M04	19	R	F	✓*	~	✓+	~
M05	18	L	М	~	~	✓+	✓
M06	41	R	М	<b>*</b>	~	~	~
M07	23	L	М	✓*	~	~	~
M08	54	R	М	~	~	✓+	~
M09	25	R	F	~	~	✓+	- ~
M10	51	R	М	~	~	~	~
M11	31	R	F	<b>*</b>	~	✓+	~
M12	38	R	F	-	~	-	~
C01	57	R	М	~	~	~	~
C02	35	R	М	-	~	✓+	~
C04	36	R	М	<b>~</b> *	>	. >	~
C05	38	R	М	~	>	✓+	~
C06	21	L	М	~	~	>	~
C07	21	L	М	✓*	~	· 🗸+	~
C08	20	R	F	~	~	>	~
C09	<u>4</u> 7	R	М	~	~	✓+	~
C11	50	R	M	~	~	~	~
C12	22 <sup>.</sup>	R	F	~	~	<b>~</b>	~
C13	44	R	M	✓*	~	~	~
C14	21	R	F	✓*	~	~	~
C15	33	Ĺ	F	✓*	~	✓+	<b>v</b>
C16	56	R	М	-	~	✓+	~
C17	57	R	F	✓*	~	✓+	~

**Table 3.2** Participants' details. M= Musicians, C= non-musicians. Right ticks are for subjects who participated in the studies. Asterisk flag and plus sign are for subjects who have been included in fMRI and MEG analysis respectively.

# Chapter 4 : Investigating of Pitch discrimination ability in musicians and nonmusicians

# 4.1 Aims

Specific aims of this chapter were to:

1- Explore the influence of musical performance expertise on pitch discrimination ability.

2- Investigate laterality effect of pitch discrimination ability.

**3-** Explore the aging effect and influence of musical experience on pitch discrimination ability.

# 4.2 Introduction

Pitch plays an important role in the perception of speech and music and in the segregation of concurrent sound sources (Plack et al., 2005). Pitch discrimination ability is highly variable among the general population. These individuals could either possess inherent abilities and/or they have gained this ability through musical training. It is still unclear what make individuals got good pitch discrimination ability without the benefit of formal musical training. It has been reported that pitch perception and discrimination accuracy are affected by musical education (Kishon-Rabin et al., 2001, Schön et al., 2004).

# 4.3 Pitch perception

Pitch labelling is an important ability in musicianship. Psychologically, pitch plays an important role as a fundamental auditory attribute in auditory communication including music and speech. Generally, it is defined as the perceptual correlation of periodicity in sounds (McDermott and Oxenham, 2008) that allows the ordering of sounds on a frequency related scale. While musically, it was defined by the American Standards Association as "that attribute of auditory sensation in terms of which sounds may be ordered on a musical scale" (ASA, 1960). It corresponds to the frequency for a pure tone and to the fundamental frequency for a periodic complex tone but it cannot be measured directly because it is not a purely objective physical property. For several decades, two basic theories of pitch perception have been proposed to explain how the

human hearing system locates and tracks changes in the fundamental frequency  $(f_0)$  of an input sound; 'place' theory and the 'temporal' theory. The place or resonance theory (Helmholtz and Ellis, 1954) states that each position on the basilar membrane (BM) is tuned with a particular characteristic frequency (CF). The frequency organisation is passed to the auditory cortex through the brainstem. Since the pitch of complex tones induces patterns of excitation along the BM with no single maximum the largest maximum may not be at the CF corresponding to the fundamental component of that pitch (Moore, 2003, Cheveigné, 2005, Warren, 2008). The temporal or frequency theory (Rutherford, 1886) assumes that pitch of a stimulus is determined by the time domain representation of the neural impulses induced by that stimulus (Moore, 2003, Cheveigné, 2005, Warren, 2008). This theory cannot work for a sinusoid tune at high frequencies because nerves do not fire fast enough (Moore, 2003, Cheveigné, 2005, Warren, 2008). Pitch discrimination in the auditory nervous system is dominated by the frequency code at low frequencies and the place code at high frequencies with both principles contributing at middle frequencies. (Wever, 1949, Gulick, 1971).

# 4.4 The frequency discrimination of pure tone

Humans can generally detect frequencies between 20 and 20 kHz, but the identifiable pitch sensation extends from 30 Hz to 5 kHz (the range of the fundamental frequencies of piano strings) (Deutsch, 1982). Frequency discrimination refers to the ability to detect changes of two nearly equal stimuli over time, which is often characterized by a difference limen (DL) or just

noticeable difference (JND) (Moore, 2003, Goldstein et al., 2004, Plack et al., 2005). The JND for pitch depends on the frequency, the sound level, the duration of the tone, and the suddenness of the frequency change (Moore, 2003).

Frequency discrimination are commonly done using two different methods; pitch matching (Goldstein et al., 1978, Moore et al., 1992) where listeners are asked to adjust the frequency of a tone to match the sound of interest, and the most commonly used method; the discrimination threshold (Srulovicz and Goldstein, 1983). The discrimination threshold method can be divided into two types. In the first measure, two successive steady tones with slightly different frequencies are presented randomly in multiple trials and the listener has to show whether the first or the second tone is higher in pitch. When the listener achieves a criterion percentage correct, for example 75%, the frequency DL between the tones will be called the difference limen for frequency (DLF). The second measure, which called the frequency modulation detection limen (FMDL), uses frequencymodulated (FM) tones at a low rate. Usually, two randomly ordered tones are presented successively; one is modulated and the other is un-modulated. The listener is asked to indicate which tone is modulated (Moore, 2003, Goldstein et al., 2004, Plack et al., 2005).

In 1970, Zwicker proposed a model based on space theory for discriminating tones separated by considerably less than a critical band (Moore, 2003, Goldstein et al., 2004, Plack et al., 2005, Warren, 2008). In this model, frequency discrimination and frequency selectivity (usually known as the ability to resolve

the frequency components of a complex sound) are related to each other. Here, frequency discrimination depends on the filtering process on the basilar membrane; hence, alternation of stimulus frequency evokes different excitation patterns (Moore, 2003, Goldstein et al., 2004, Plack et al., 2005). This model has difficulties handling very small JNDs for pure tones especially with frequencies from about 500 Hz to 2k Hz (Goldstein et al., 2004). Furthermore, Moore (1973) tested this model and found that its prediction for variation of the frequency DL with frequency is not accurate especially for short-duration tones below 5 kHz. Moore (1973) suggested an inclusion of temporal model especially for frequencies below 5 kHz. In contrast, the place model shows results that are more accurate for FMDLs especially for relatively high modulation rates (10k Hz and above). In summary, the temporal model determines DLFs for pure tones and FMDLs for very low modulation rates for frequencies up to about 4-5 kHz more accurately than the place model. While DLFs for pure tones of high frequency and FMDLs for medium to high modulation rates and for FMDLs for low modulation rates, when the carrier frequency is above about 5k Hz are determined by place model (Nordmark, 1968, Moore, 1973, Moore, 2003, Goldstein et al., 2004, Plack et al., 2005, Warren, 2008).

### 4.5 Pitch discrimination in musicians and non-musicians

Perception of pitch is evoked by a pure tone and hence pitch variations over time produce perception of melodies. Musical pitch has two dimensions, pitch height and pitch chroma. Pitch height represents the overall pitch level and provides a

basis for segregation of notes into streams and pitch chroma provides a basis for presenting melodies (Warren et al., 2003). Accordingly, musicians show superior auditory performance in tasks that involve discriminating sounds along both pitch dimensions (height and chroma) comparing with non-musicians. This has been reported in different studies reflecting specific aspects of music, such as timbre, rhythm, pitch discrimination, mistuned harmonics and the identification of musical intervals (Spiegel and Watson, 1984, Pitf, 1994, Burns and Houtsma, 1999, Koelsch et al., 1999, Kishon-Rabin et al., 2001, Fujioka et al., 2004, Fujioka et al., 2005, Tervaniemi et al., 2005, Seither-Preisler et al., 2007, Nikjeh et al., 2009, Zendel and Alain, 2009). However in some of the music characteristics, non-musicians with appropriate training (Kishon-Rabin et al., 2001, Amitay et al., 2006, Micheyl et al., 2006, Wright and Sabin, 2007) or repeated exposure to music (Bigand, 2003, Bigand and Poulin-Charronnat, 2006) can nearly obtain the achieved performance by experienced musicians.

In the present study, I attempted to maximize group separation. I included in the group of musicians only professional musicians who had 10 years or more of experience playing an instrument. Non-musicians had not played a musical instrument in the last 10 years and never performed professionally.

# 4.6 Materials and Methods

### 4.6.1 Subjects

Data was collected from 12 musicians (7 males and 5 females) with an age of  $38 \pm$  14.07 (mean  $\pm$  SD) years and 14 age matched non-musicians subjects (10 males

and 4 females) (the mean age: 37 years (SD = 16.62). Volunteers participated in the study after giving written informed consent. They had no history of any neurological or hearing impairment.

# 4.6.2 Stimuli and procedure

Three types of psychoacoustic tests were run; hearing threshold, loudness and pitch discrimination. In all tests, a pure tone of 500 Hz was used, unfortunately larger range of frequencies such as 125 to 15k Hz and 50 to 5k Hz for hearing threshold and pitch discrimination respectively could not be run due to time limitations. Sound stimuli were created and presented using custom-made scripts in MATLAB 7.0 (MathWorks, Natick, USA). Hearing threshold and loudness tests were used to ensure that all listeners are within certain range of hearing threshold and loudness. During all tests, a paradigm of three-alternative forcedchoice was used. In this paradigm, three buttons were marked on the computer screen "1", "2" and "3". In the hearing threshold test, a single tone was presented and the task of the listener was to indicate which of the three buttons flash when the tone was heard. In the loudness test, three tones with different loudness levels were presented. The task of the listener was to indicate which of the three buttons flash when the loudest tone was heard. In the pitch discrimination test, three tones are presented with different pitch levels. The task of the listener was to indicate which of the three buttons flash when the highest pitch tone was heard. In all tests, listeners gave their responses by pressing "1", "2" or "3" on a computer keyboard. Three blocks were run for each test (right ear (RE), left ear (LE) and both ears

(BE) at the same time), each block lasted for about two minutes. The unit of pitch discrimination used to characterise responses to different tones was the "cent": a logarithmic unit used to measure musical intervals equal to a hundredth of a semitone.

#### 4.6.3 Statistical analysis

Statistical analyses were performed using SPSS (for MAC, Rel18.0. 2010. Chicago: SPSS Inc.). Group comparisons of PDP of BE together, the right and left ears were performed using a 2-way ANOVA. A paired (samples) t-test was used to compare the means of PDP of the left and the right ears in each group. Parametric correlations analyses were performed between PDP and age, proficiency period and start of training age.

#### 4.7 Results

ANOVA statistical test showed that there was a significant effect of musicianship on PDP of all tests (BE simultaneously, the RE and the LE independently), (F(1,23) = 11.972, P = .002), Figure 4.1. However, group-ear interaction effect was not statically significant (F(1,23) = .000, P = .991). The means of PDP in musicians were about 8 cents in the LE test, 9 cents in BE simultaneously tests and 11 cents in the RE test, Figure 4.1. In non-musicians, the means of PDP were just above 23 cents, just less than 28 cents and just above 32 during all tests (BE simultaneously, the RE and the LE independently), Figure 4.1.

Chapter 4 Investigating of Pitch Discrimination Ability in Musicians and Non-musicians



Figure 4.1 Estimated means of PDP of both ears (BE) simultaneously, the right ear (RE) and the left ear (LE) independently (Smaller score of pitch discrimination means better performance).

Paired (samples) t-test of the means of PDP showed that performance of the LE was statistically significant compared to the RE in musicians, t (11) = -2.352, P = .041, Figure 4.2. In contrast, non-musicians showed no significant difference in performance between BE, t (11) = .711, P = .490, Figure 4.3.



Figure 4.2 95% confidence interval for PDP in the left and RE in musicians.





Correlations between PDP and age, musical proficiency period and the age at which the person had begun musical training are shown in Table 4.1. There was correlation between age at which musicians begun to practise and PDP of the LE with borderline significance (uncorrected P=.057), (P=.171 Bonferroni adjusted).

		Pitch discrimination performance				
		BE	LE	RE		
Age		.090	.024	.058		
	ed '	.663	.907	.779		
Musical proficiency period	tioi	.239 *	149	.212		
	elai	.455	.662	.509		
Age at which musicians	orr	.205	588	025		
begun to practise	N C P	.523	.057	.938		

**Table 4.1** Correlations between all PDP measurements (PDP of the RE, the LE independently and BE simultaneously) and age, musical proficiency period and age at which musicians begun to practise.

#### 4.8 Discussion

The present study showed a strong relationship between musical proficiency and PDP in all three conditions (BE together, the right and left ears), with no musicianship ear interaction effect. However there was a significant enhancement in PDP of the LE compared to RE in musicians. In contrast, there were no significant different in the performance between the LE and the RE within non-musicians group. These results indicate that musicians had right-hemisphere dominance for pitch discrimination. Mathys et al. (2010) reported that both left and right HG contribute in pitch discrimination, where the right HG might be a stronger contributor. There were no correlations between PDP and age, musical proficiency period or the age at which the person had begun musical training (see Table 4.1). However, as the sample size is small these findings may not be

replicated for larger groups of experienced musicians. Work on larger sample size is needed to properly investigate instrumental effect on PDP as well as testing musicians' listening preferences.

# **4.9 Conclusion**

Musicians showed better performance in pitch discrimination than non-musicians in all three conditions. Musicians performed better their left ear during PDP test. This result indicates that musicians had right-hemisphere dominance for pitch discrimination.

# Chapter 5 : Neuroanatomical correlates of musicianship and PDP

# 5.1 Aims

Specific aims of this chapter were to:

1- Investigate anatomical plasticity of selected brain structures in musicians.

2- Explore the correlations between PDP and some cortical features of selected brain structures.

3- Explore the musical proficiency effect on sulcal and gyral topography.

4- Explore the instrument type effect on sulcal and gyral topography.

# **5.2 Introduction**

The human brain has an excellent ability to reorganize neural pathways and to alter cortical patterns according to the context of the surrounding environment. (Naatanen and Winkler, 1999, Pantev et al., 1999, Tervaniemi et al., 2005, Munro, 2008, Rahne and Sussman, 2009, Tanaka et al., 2009, Trainor et al., 2011) Early intensive musical training affects brain development to a great extent (Shahin et al., 2004). It could change the number of neurons involved in processing, the timing of synchronization and the number and strength of excitatory and inhibitory synaptic connections (Schneider et al., 2002, Jones et al., 2010). Starting at early age, musicians integrate multi functions processes (e.g. auditory, motor, visual, and cognitive tasks) comprehensively during musical performance demanding activation of multiple brain regions. Several studies have reported structural changes in brains of musicians (For details see section 2.4.4.1).

In the current study, a multi-method approach has been used to investigate the neuroanatomical correlates of musicianship and PDP. Different anatomical techniques have been applied to explore plasticity of different cortical features including; GM density, cortical thickness and sulcal depth cortical volume and surface area, and hippocampal volume and shape differences. In addition, I studied the effect of instrument type on sulcal and gyral topography.

Chapteer 5 Neuroanatomical Correlates of Musicianship and Pitch Discrimination Performance

# **5.3 Materials and Methods**

#### 5.3.1 Subjects

Data was collected from 12 musicians (7 males and 5 females) with an age of  $38 \pm 14.07$  (mean  $\pm$  SD) years and 14 matched non-musicians subjects (10 males and 4 females) (the mean age: 37 years (SD = 16.62). Volunteers participated in the study after giving written informed consent. Additionally, the musician group was divided into two groups (string and non-string players) to investigate effect of instrument type on sulcal and gyral topography (Bangert and Schlaug, 2006).

# 5.3.2 Anatomical imaging sessions

Whole head 3D anatomical datasets were obtained for all subjects using a 1.5 T Symphony whole body scanner (Siemens, Erlangen, Germany) (176 slices, TR/TE= 1660/3.05 ms; 2 averages; flip angel = 8°, matrix size 256 x 256; voxel size =  $1 \times 1 \times 1$  mm3). The duration of scanning time was 14 minutes and 11 seconds. The 1.5T scanner was used because it produces T1-weighted images with reduced sensitivity to susceptibility and minimised distortions in anatomical images (McRobbie, 2003).

#### 5.3.3 Anatomical investigation methods

#### 5.3.3.1 Voxel based morphometry

Voxel based morphometry (VBM) is an automated method of identifying structural brain differences depending on voxel-wise comparison of local concentrations of grey and white matter between two or more groups. It provides a comprehensive assessment of variation in anatomical distribution of tissues (Ashburner and Friston, 2000).

The first stage was performing spatial normalization by transforming brains of all subjects into the same stereotactic space (MNI asymmetrical average template of 305 subjects) by registering each image sets with the template image. Images were matched using a 12-parameter affine transformation with accounting for global non-linear shape differences by a linear combination of smooth spatial basis functions. All resultant images were relatively high resolution (1 mm isotropic voxels) to minimise the effects of partial volume. Then spatially normalised images were segmented into GM, WM and CSF. The normalised, segmented images were smoothed by convolving with a Gaussian kernel. Each voxel in the resultant image was an average of the tissue type from the neighbouring voxels in order to represent the local density of the tissue type at that voxel. This also helps in compensating for inter-individual gyral variation in the normalised and segmented images. Different smoothing kernel sizes were applied to provide optimal sensitivity for regions of interest (4 and 10) mm full width at half maximum (FWHM) Gaussian kernel. The final step was to apply voxel-wise parametric statistical tests on the normalised segmented and smoothed images of the groups selected (Ashburner and Friston, 2000). VBM analyses were carried out running MATLAB 7.9.0.529 (R2009b) (Mathworks Inc., Natick, MA, USA; http://www. mathworks.com/products/matlab) and VBM8 (http://dbm.neuro.unijena.de/vbm), implemented as a toolbox in SPM8 software (Wellcome Department of Cognitive Neurology, London; http://www.fil.ion.ucl.ac.uk/spm).

# **Statistical analysis of VBM**

Statistical group analysis was performed using 2-sample *t*-test. Analyses of covariance were performed, with global mean voxel value as a confounding covariate. This covariate corrected for the effect of summed difference in voxel intensity across scans, while preserving regional differences in tissue type. Two contrasts were defined to detect whether each voxel had a greater (or lesser) probability of being GM in the musicians group than had the equivalent voxel in non-musicians group, representing an increase or decrease in local GM density in musicians group. Significant foci above the P<0.01 level were characterised of their peak height (u) and spatial extent (k). Locations of significant clusters were assessed by localising their centre of gravity coordinates on template atlases using FSLView 3.1.8 (http://www.fmrib.ox.ac.uk/fsl) and SPM Anatomy toolbox (http://www2.fz-juelich.de/inm/index.php?index=194) (Eickhoff et al., 2005).

# 5.3.3.2 Cortical thickness and sulcal depth

## Pre-processing

All image data preparation and pre-processing, cortex-based alignment and cortical thickness and sulcal depth analysis were carried out using BrainVoyager QX software package (Brain Innovation, Maastricht, The Netherlands; Version 2.3). The anatomical data (DICOM format) of each subject was converted into

BrainVoyager's internal data format. The data was resampled into 1 mm isotropic voxels, aligned with the ACPC plane and transformed into the Talairach coordinate system. The Talairach transformed image was used for automatic cortical surface reconstruction (Kriegeskorte and Goebel, 2001).

# Advanced segmentation analysis

Talairach transformed image was resampled to 0.5 mm iso-voxels. The brain was segmented from surrounding head tissue using an automatic "brain peeling" tool (advanced segmentation tools). The cerebellum and subcortical structures were removed. Enhancement of tissue contrast and homogeneity was done using a sigma filter, which operates like a standard gaussian smoothing filter but excludes neighbour voxels with intensities too far away from the intensity of the currently considered voxel. The WM–GM border was set automatically using an adaptive region-growing step to separate white from GM voxels. The GM–cerebrospinal fluid (CSF) border was segmented using a dilation process, which controlled by computed gradient fields and histogram analysis of GM–CSF threshold values. The polishing process, which estimate a more correct GM / CSF contour was performed by calculating the magnitude map (Goebel et al., 2006).

#### **Cortical based alignment**

Reconstructed cortical surfaces were aligned using curvature information reflecting the gyral-sulcal folding pattern to improve the spatial correspondence mapping between subjects' brains, which reduces anatomical variability. First step was morphing the reconstructed folded cortical representations of each subject

#### Chapteer 5 Neuroanatomical Correlates of Musicianship and Pitch Discrimination Performance

and hemisphere into a spherical representation. The curvature map created from folded representation was overlaid on the spherical representation where each vertex on the sphere (spherical coordinate system) corresponds to a vertex of the folded cortex (Cartesian coordinate system) and vice versa. The curvature information was smoothed along the surface providing spatially extended gradient information to minimise the mean squared differences between the curvature of a source and a target sphere All brains were aligned using a moving target approach where the goal function is specified as a moving target computed repeatedly during the alignment process as the average curvature across all hemispheres at a given alignment stage. It starts with the coarsest curvature maps then with finer curvature maps (Goebel et al., 2006).

### Cortical thickness and sulcal depth analysis

Cortical thickness identifies structural brain change depending on voxel intensity values. It can be calculated by selecting two different intensity values, one at the white-grey matter boundary and one at the grey-CSF boundary. A smoothed transition of intensities from one boundary to the other is resulted from applying Laplace's equation (Jones et al., 2000). Laplace's equation is a partial differential equation, which is used in many fields of science e.g. electromagnetism and fluid dynamics. Obtained smooth field can be used to calculate a gradient value at each voxel, which results in field lines by integrating along these gradient values. Calculating a cortical thickness value starts at any boundary voxel, and then a small step is performed along the gradient direction after checking the gradient. A re-evaluation of the gradient at the new point is done before performing the next

#### Chapteer 5 Neuroanatomical Correlates of Musicianship and Pitch Discrimination Performance

step along the gradient direction. This procedure is repeated until reaching the other boundary. Cortical thickness value is the sum of the performed small step sizes. Going in both directions (gradient directions) calculates a thickness value for any voxel in between boundary voxels producing cortical thickness volume maps that contain the thickness measures at cortical voxels.

Depth of sulci can be measured in the same way with the only difference that one boundary must be the boundary of the brain and the other boundary must be the white or grey matter. The computed field lines then travel from any WM or GM point through the sulci up to the boundary of the brain and the integrated path length values result in sulcal depth measures. This process produces sulcal depth volume containing the depth measures at sulcus voxels. Both cortical thickness and sulcal depth volume maps were used to efficiently calculate maps on cortical surface meshes producing cortex thickness and sulcal depth surface maps.

Locations of significant clusters were assessed by visual inspection and localising their centre of gravity coordinates on template atlases using BrainVoyager Brain Tutor 2.5 for MAC (http://www.brainvoyager.com/products/braintutor.html) and FSLView 3.1.8 (http://www.fmrib.ox.ac.uk/fsl).

# 5.3.3.3 Cortical volume and surface area

Aligned brains (resultant of cortical based alignment) were used to superimpose several regions of interest; STG, HG and PT. These regions were transformed into volume of interest and corrected manually using macro-anatomically defined anterior, posterior, medial and inferior boundaries as in method previously published in (Rademacher et al., 2001, Hall et al., 2003, Abdul-Kareem and Sluming, 2008). Manual correction was carried out to minimise alignment errors. Next, corrected region and volume of interests were used to measure cortical surface area and volume respectively. Individual cortical surface area and volume values were exported to SPSS 18.0. Statistical analyses were performed using SPSS (for MAC, Rel18.0. 2010. Chicago: SPSS Inc.). Group comparisons of volume and cortical surface area values of auditory areas were performed using a 2-way ANOVA. Parametric correlations were performed between PDPs of all conditions (the LE, the RE and BE together) and cortical surface area or cortical volume of the following regions; PAC, PT, lateral aspect of HG and STG. *P* values of ( $\leq 0.05$ ) were considered statistically significant. Bonferroni multiple comparisons correction were performed on all *p* values (Salkind and Rasmussen, 2007).

#### 5.3.3.4 Hippocampal shape and subfields volume

#### **Segmentation of hippocampus**

The anatomical T1 data (DICOM format) of each subject was converted into compressed NIfTI format using MRIcron (http://www.cabiatl.com/mricro/mricron). Using FSL (http://www.fmrib.ox.ac.uk/fsl) (Smith et al., 2004, Woolrich et al., 2009), the following tools were applied sequentially: brain extraction by the BET tool (Smith, 2002) to clear non-cerebral voxels; automated segmentation by the FIRST tool of the left and right hippocampi (Patenaude et al., 2011). FIRST tool is based on a template originating from manually segmented images, with

#### Chapteer 5 Neuroanatomical Correlates of Musicianship and Pitch Discrimination Performance

subcortical labels parameterized as surface meshes. First brain-extracted image was registered to the template via 12 degrees of freedom affine transformation (Jenkinson and Smith, 2001), and then 12 degrees of freedom registration to a standard MNI 152 template of subcortical mask. Hippocampi were labelled and segmented using subcortical mask depending on T1 image intensity. Registration and segmentation processes of both hippocampi were visually checked at each step to confirm accuracy of the results.

# Analysis of the shape of the hippocampus

A multivariate Gaussian model of vertex location and intensity variation was used to automatically generate surface mesh for both hippocampi of each subject. The same number and labelling of vertices were used to enable point-to-point comparisons across all subjects. Corresponding vertex points of generated surface were aligned to the mean surface of the template hippocampus in MNI152 space. F statistics were then calculated to compare shape differences between musicians and non musicians (Patenaude et al., 2011).

# Subfields volume of the hippocampus

The workflow for creating subcortical segmentations was processed using FreeSurfer version 5.1 (released May 24th, 2011; Martinos Center, Harvard University, Boston, MA) (http://surfer.nmr.mgh.harvard.edu) on a PC running Open SUSE 11.3 Linux. FreeSurfer's built-in mri\_convert command was used to convert T1 images to FreeSurfer format (.mgz). Converted images were registered to Talairach space using 12 degrees of freedom affine transformation. Next,

#### Chapteer 5 Neuroanatomical Correlates of Musicianship and Pitch Discrimination Performance

fluctuations in scan intensity were corrected and scan intensities were adjusted to achieve a mean WM intensity of 110. The skull and meningeal surfaces were removed from the scan, leaving only the brain tissue. Segmentation labels were applied by aligning each image with the FreeSurfer atlas. Volume labels were applied to subcortical structures based on the prior probabilities of voxel identity assigned by the atlas, in addition to the probability of voxel identity based on the tissue class assignment of surrounding voxels, and volumetric statistics were computed (Fischl et al., 2002). Automated segmentation of the subfields of the hippocampus, was carried out using freesurfer models built from manual segmentations of the RHipp (Van Leemput et al., 2009). Individual subfields volume values were exported to SPSS. ANOVA for repeated measurements was used to compare hippocampal subfields volume values in both hemispheres. Parametric correlations were performed between PDPs of all conditions (the LE, the RE and BE) and volume of hippocampus subfields. P values of ( $\leq 0.05$ ) were considered statistically significant. Bonferroni multiple comparisons correction were performed on all *P* values (Salkind and Rasmussen, 2007).
## 5.4 Results

## 5.4.1 Voxel based morphometry

In this study, I investigated differences in GM distribution between musicians and non-musicians on a voxel by voxel basis. 2-sample t-test was computed considering total intracranial volume (TIV) as a global variable at every voxel. Two contrasts were examined: 1) detecting local increases in GM distribution in musicians compared with non-musicians; 2) detecting local increases in GM distribution in non-musicians compared with musicians. Repeating previous tests, PDPs of each ear (chapter 3) were considered as numerical covariates in order to test their effect on local increase in GM density. Each test was carried out on previously smoothed images (with 10 mm smoothing kernel). Statistically significant clusters were corrected for repeated measures on cluster level threshold at p<0.01 and with spatial extent of 20 mm<sup>3</sup> or above. Cluster level correction, was performed using region of interest approach to improve statistic power. Regions of interest were defined using MarsBaR (http://marsbar.sourceforge.net) that implemented as a toolbox in SPM8 software (Wellcome Department of Cognitive Neurology, London; http://www.fil.ion.ucl.ac.uk/spm).

#### 5.4.1.1 ANCOVA test with TIV as global variable

Musicians showed significant increased GM density compared to non-musicians in the left angular and supramarginal gyri (left IPL) and right postcentral as shown in Figure 5.1 and Table 5.1. Non-musicians showed no increased GM density compared with musicians.





**Figure 5.1** Statistical parametric map (extent threshold  $P_{corr}=.05$ ) showing clusters with statistically significant of increased GM density in musicians.

Cluster peak region	Voxels	Zmax	P <sub>corr</sub>	x	у	Z
Left angular gyrus	612	4.34	.006	-44	-51	19
Left supramarginal gyrus	600	4.30	.007	-56	-45	34
Right postcentral gyru	102	3.80	.023	18	-40	63

Table 5.1 P values and MNI coordinates of cluster peak regions.

#### 5.4.1.2 ANCOVA test with PDPs as covariates

Applying PDP of LE as a covariate showed a significant increase in regional GM density in the posterior region of musicians' RHipp (particularly in subiculum) and right angular gyrus as shown in Figure 5.2 and Table 5.2. Non-musicians showed no increased GM density.

Cluster peak region	Voxels	Zmax	P <sub>corr</sub>	x	у	Z
RHipp (subiculum)	38	3.39	.023	18	-32	-5
Right angular gyrus	15	3.45	.027	39	-54	29



Table 5.2 P values and MNI coordinates of cluster peak regions.

Figure 5.2 Statistical parametric map (extent threshold  $P_{corr}=.05$ ) showing clusters with statistically significant of increased GM density in musicians.

# 5.4.2 Cortical thickness, sulcal depth and sulcal and gyral topography

## 5.4.2.1 Musical proficiency effect on sulcal and gyral topography

Figure 5.3 shows the average cortical surfaces generated for the musicians and non-musicians groups. Visual inspection of these representations suggested subtle group differences in sulcal and gyral patterning in different regions e.g. right central sulcus, right dorsal posterior cingulate (Brodmann's area 31), left calcarine sulcus and left and right SMG. For example, the right central sulcus was wider in musicians than in non-musicians. Surface patterns of the STGs were more folded in musicians as seen in Figure 5.3.



Figure 5.3 Superior, medial and lateral views of the average cortical surfaces of musicians and non-musicians. Circles show differences in sulcal and gyral topography.

The posterior ramus of the lateral sulcus was oriented more vertically in the LH of the non-musicians than in musicians (Figure 5.4), and hence, the left SMG was more posterior in non-musicians than musicians as shown clearly in Figure 5.4.



Figure 5.4 Average cortical contours from musicians (blue) and non-musicians (yellow) are superimposed on average target volume for all subjects.

Figures 5.3 and 5.4 show that, the depth of both the right postcentral sulcus and right Brodmann's area 31 were greater in musicians. Additionally, musicians had longer path of calcarine sulcus. The t-statistic maps of sulcal depth difference between both groups are shown on Figure 5.5. The cortex was significantly deeper in musicians in various regions e.g. right postcentral sulcus, anterior portion of right and lift cingulate sulci, right lateral sulcus, left calcarine sulcus and anterior portion of superior temporal sulcus. In contrast, non-musicians significantly showed deeper cortex in posterior part of left superior frontal sulcus, right superior temporal and left inferior temporal sulci.



Figure 5.5 T-statistic maps of group difference in sulcal depth are superimposed on average target surface for all subjects.



Figure 5.6 T-statistic maps of group difference in cortical thickness are superimposed on average target surface for all subjects.

Figure 5.6 shows the t-statistic maps of cortical thickness difference between both groups. Musicians had thicker cortex in SMG and angular gyrus in both hemispheres. Additionally, cortex was thicker in musicians in the posterior portion of left parahippocampal gyrus and left and right lingual gyrus. Musicians also showed thicker cortex in the posterior part of the right postcentral gyrus, medial part of the left postcentral gyrus and inferior frontal gyrus (posterior part of brodmann area 44) in the LH. In contrast, non-musicians had thicker cortex in anterior part of left middle frontal gyrus, and right gyrus rectus.

# 5.4.2.2 Effect of instrument type on sulcal and gyral topography

Figure 5.7 show that, the string players (6 musicians) had prominent omega shape (Yousry et al., 1997) of central sulcus on the RH only, while the non-string players preserve omega shape in both hemispheres (Bangert and Schlaug, 2006). The right precentral sulcus was extended more medially in non-string players (Figure 5.7). The posterior ramus of the lateral sulcus was tilted more anteriorly and posteriorly in the right and left hemispheres respectively in non-string players than those of string players (Figure 5.9). Therefore, the SMG was more anterior and more posterior in the right and left hemispheres respectively in non-string players than string players as shown in Figures 5.7 and 5.8. Additionally, some other regions show differences in shape between both groups such as right dorsal posterior cingulate (Brodmann's area 31) and left calcarine sulcus. In Figure 5.9, the t-statistic maps of group difference in sulcal depth show that non-string players had more regions with deeper cortex than string players.



Figure 5.7 Superior, medial and lateral views of the average cortical surfaces of string and nonstring players (within musicians group). Circles show differences in sulcal and gyral topography.



Figure 5.8 Average cortical contours from string players (blue) and non-string players (yellow) are superimposed on average target volume for all subjects.

Non-string players showed deeper cortex in the following regions; right and left lateral sulcus, right and left occipitotemporal sulcus, left intraparietal sulcus and left calcarine sulcus. While the string players had deeper cortex in the left central, postcentral, superior temporal and inferior temporal sulci. Comparison of cortical thickness maps shows small clusters with significant t-statistic. The string players had thicker cortex in the anterior region of right middle frontal gyrus, middle part of the left inferior temporal gyrus, medial region of orpital gyrus, right dorsal posterior cingulate and medial occipitotemporal gyrus. In contrast, non-string players showed thicker cortex in the following regions; medial part of left precentral gyrus, middle region of cingulate gyrus, right lingual gyrus and middle part of right parahippocampal gyrus Figure 5.10.



Figure 5.9 T-statistic maps of group difference in sulcal depth are superimposed on average target surface for all subjects.



Figure 5.10 T-statistic maps of group difference in cortical thickness are superimposed on average target surface for all subjects.

#### 5.4.3 Cortical volume and surface area of auditory regions

#### 5.4.3.1 Influence of musical performance on cortical surface area

A 2-way ANOVA statistical test showed that there was a significant musicianship effect in the cortical surface area of medial HG (mGH) (F(1,25) = 5.617, P = .026), with no hemisphere-group interaction (F(1,25) = .031, P = .861) Figure 5.11.





## 5.4.3.2 Correlation between cortical surface area of auditory regions and PDP

Performing parametric correlations analysis showed that there were positive and strong correlations between PDP of the RE and cortical surface area of the lateral aspect of left HG (r (26) = -.527, P = .006) and the left STG (r (26) = -.507, P = .008) (Bonferroni adjusted) Figure 5.12.



**Figure 5.12** Correlation between PDP of all conditions and cortical surface area  $(mm^2)$  of the left STG and the lateral aspect of left HG. (Smaller score of pitch discrimination means better performance).

#### 5.4.3.3 Correlation between GM volume of auditory regions and PDP

Figure 5.13 shows the correlation between PDP of the RE and GM volume of the left PT and lateral aspect of left HG. GM volume of the lateral aspect of the left HG was statistically correlated with PDP of the RE (r (26) = -.549, P = .004) (Bonferroni adjusted).



**Figure 5.13** Correlation between PDP of the RE and cortical volume (mm<sup>3</sup>) of the lateral aspect of left HG and left PT. (Smaller score of pitch discrimination means better performance)

# 5.4.4 Hippocampal shape difference and and subfields volume differences



5.4.4.1 Influence of musical performance on hippocampal surface shape

Figure 5.14 Statistic maps of surface and shape difference in the right hippocampus. The colour bar indicates the statistic values; an increase from red to blue is going from lower to higher statistical significance.

The group differences of hippocampi shape were significant only in the right hippocampus. Figure 5.14 shows that there were 3 regions (blue patches) with significant values (5 mm or more). The outward vectors give direction of difference is such that the musicians' RHipp is larger/thicker than in non-musicians (most posterior and the lateral anterior regions of the right hippocampus). While inward vectors give direction of difference is such that the non-musicians' RHipp is larger/thicker (superior anterior region). There were no significant differences of the right and left hippocampi volumes.

#### 5.4.4.2 Influence of musical performance on hippocampal subfields volume

Segmentation process produced nine subfield regions. Five of those regions are considered as part of global GM tissue type; CA1, CA2/3, CA4/DG, subiculum and presubiculum, and one considered as WM tissue type (fimbria) (Van Leemput et al., 2009) Figure 2.2. While the hippocampal fissure, inferior lateral ventricle were considered as CSF tissue type (Van Leemput et al., 2009).

ANOVA for repeated measurements showed there were group-hemisphere interaction effect (F(1,24) = 6.077 (P=.021) and no group-hemisphere-subfields interaction effect (F(1,24) = .010 (P=.921). Post hoc t-test were therefore, conducted between musicians and non-musicians for each subfield separately in Figure 5.15.



#### 5.4.4.3 Correlation between volume of right hippocampal subfields and PDP

There was positive correlation between PDP of the LE and volume of fimbria region in the RHipp (r (25) = -.497, P = .012) (Bonferroni adjusted), Figure 5.16.



Figure 5.16 Correlation between PDP of all conditions and volume of fimbria (Smaller score of pitch discrimination means better performance).

5.4.5 Correlation between musicians' characteristics and both volume of hippocampal subfields and cortical surface area and volume in the auditory regions

There were no correlations between cortical surface area and volume in the auditory regions and musical proficiency period or the age of commencing musical training. As well as the correlations between volume of hippocampal subfields and both musical proficiency period and the age of commencing musical training were not statistically significant.

# **5.5 Discussion**

This multi-methods study consisted of four different parts; i) VBM, ii) sulcal and gyral topography including cortical thickness and sulcal depth, iii) cortical surface area and volume of the auditory regions and iv) right hippocampal shape. Musicians showed significantly increased volume in various brain regions; left angular gyrus, right postcentral, and right hippocampus. Also there were statistically significant shape differences in sulcal and gyral anatomy. Additionally, there were significant brain-behavioural correlations in different regions e.g. lateral aspect of HG and fimbria of the right hippocampus.

VBM study showed that GM density in the left IPL and right postcentral gyrus of musicians were significantly increased compared to non-musicians while there was borderline significant increased GM distribution in the right angular gyrus in musicians compared to non-musicians as shown in Figure 5.1. Those regions are part of the auditory-motor coupling network (Gaab and Schlaug, 2003, Bangert et al., 2006, Gaab et al., 2006). VBM results had been further investigated using PDP of the LE as covariates showing significant large GM density in the right angular gyrus in musicians compared to non-musicians (Figure 5.2). IPL is considered as part of the pitch short memory network (Binder et al., 1997, Rinne et al., 2009).

Comparison of the topographic patterns of sulci and gyri in musicians and nonmusicians showed that musicians had deeper cortex in the following sulci: right

postcentral right and left cingulate, right lateral and left calcarine sulci. Musicians also showed thicker cortex in multiple brain regions including the left parahippocampal gyrus, the left and right lingual gyrus, the posterior part of right postcentral gyrus, medial part of left postcentral gyrus, posterior part of the left Broca's area, the left and the right IPL (see Figures 5.3–5.6). Cortical thickness results closely agree with VBM results (particularly in the left IPL and the right postcentral gyrus). Sulcal and gyral topographic difference in musicians reflect multi-modal integration of auditory, motor, and cognitive tasks in response to daily extensive requirements of multi functions process; e.g. pitch perception, playing on instruments, remembering and musical sight-reading. In addition, sulcal and gyral topographic patterns indicate considerable within-musician differences (see Figures 5.7-5.10). These differences suggest an association between acquired sensory-motor skills and features of brain topographic patterns especially in the primary and pre-motors area and some brain regions that associated with motor tasks. Association of changes of the brain's external features with motor training has been reported previously in animals (Kleim et al., 1996, Anderson et al., 2002) and humans (Bengtsson et al., 2005, Bangert and Schlaug, 2006, Hyde et al., 2009, Schlaug et al., 2009a). Therefore, I interpret my findings of the differences in sulcal and gyral topography patterns (including sulcal depth and cortical thickness) between musicians and non-musicians to reflect use-dependent structural adaptation associated with expert musical skill acquisition.

Musicians had larger cortical surface area in the anteromedial portion of the left HG Figure 5.11. Additionally, this study showed positive brain-behavioural correlations in the auditory regions (STG, PT and lateral aspect of HG) in the LH in both groups and irrespective to musical proficiency (Figures 5.12–5.14). The strongest correlations were found between PDP of the RE and both cortical surface area and GM volume of the lateral aspect of left HG. The lateral part of HG is considered as the pitch-processing centre (Griffiths, 2003). Although the musical proficiency only influence anteromedial portion of the left HG the brain-behavioural correlations in non-primary auditory areas are irrespective to musical proficiency. Schneider et al. (2005) observed a strong relation between absolute magnitudes of structure and function at the early automatic processing level in anteromedial HG only. This suggests that genetic predisposition could influence the anatomical differences particularly in non-core auditory areas, which would be enhanced by training.

Of importance, musicianship affected the surface shape of some regions of the RHipp but not the whole hippocampus. Musicians' RHipp was thicker in the lateral anterior and the most posterior regions compared to non-musicians' RHipp Figure 5.14. No significance difference was found between musicians and non-musicians in the subfields volume in both hippocampi. But there were group-hemisphere interaction and a significant difference between two hemispheres Figure 5.15. Post hoc t-test showed significant difference between musicians and non-musicians in some subfield regions (fimbria, subiculum and presubiculum in the RHipp and CA4/DG in the left hippocampus). The findings of hippocampus

study and the VBM analysis including PDPs of the LE as a covariate strongly agree with each other. In VBM, musicians had increased GM density in the posterior portion of hippocampus compared to non-musicians using PDP of the LE as a covariate Figure 5.2 and Table 5.2. It has been reported that the posterior part of subiculum is relatively more concerned with space and memory (O'Mara et al., 2009). In addition, volume of fimbria region correlated positively with PDP of the LE (Figure 5.16). Hippocampus plays an important role in pitch memory and auditory temporal information in animal (Sakurai, 2002) and human (Watanabe et al., 2008). I propose that the correlation between volume of fimbria of the RHipp and PDP of the LE reflect an overall internal/external reorganization of hippocampal connections in response to increasingly demanding tasks and skills required to perform pitch discrimination tasks. The present study has not fully reported previous data (Sluming et al., 2007b) about the positive correlation between hippocampal volume and musical proficiency in male musicians under 50 years of age. This was possibly related to the small sample size.

## 5.6 Limitations

Each method has its own limitation. VBM method has a few limitations (Ashburner and Friston, 2000, Bookstein, 2001, Mechelli et al., 2005) including sensitivity to methodological choices in normalization, smoothing kernel and template. To assure the best results I used default setting of VBM8 toolbox during pre-processing step. Additionally, the quality of normalised and segmented images was visually inspected. I got similar results using two different smoothing

kernels (4 and 10 mm) providing optimal sensitivity for regions of interest. Another limitation is that size, position, and morphology are concurrently linked in VBM results (Bermudez et al., 2009). VBM results were verified using other anatomical methods e.g. cortical thickness, sulcal depth and cortical surface area. These methods have clearly comprehensible biological meanings than GM density.

Using BrainVoyager OX to calculate cortical thickness and sulcal depth. I performed surface-based registration, which offers an accurate gyral/sulcal topology and allows the preservation of the cortical anatomy of the original surface model (Fischl et al., 1999, Desai et al., 2005). All brains were aligned using a moving target approach (Goebel et al., 2006) without user intervention. In regions of high shape variability, the algorithm can force alignment process leading to a misalignment of the corresponding regions (Van Essen, 2005). It has been shown that BrainVoyager QX is not successful in aligning the curves (Pantazis et al., 2010). In addition, some regions e.g. frontal lobes are not performed accurately due to technical limitations of the cortical thickness algorithm (Geuze et al., 2008). Furthermore, there is an interindividual variation in cortical thickness in healthy control subjects of up to 15% (Haller et al., 2009). Although, misalignment of the cortical regions affects the measurements of cortical surface area and volume, manual correction would overcome any inadequate results. Correction procedure is an excruciatingly time consuming process.

For Hippocampus measurements a few data set have been excluded from further analysis due to major defects during automatic segmentation of hippocampus by both techniques; Freesurfer and FIRST. This can be explained by over-inclusion of part of boundary voxels, neighbour regions e.g. parahippocampal gyrus and some CSF. Segmentation algorithms of both techniques are sensitive to differences in contrast in boundary voxels especially in FIRST (Morey et al., 2009). Therefore quality of segmented images was carefully inspected and major defects were manually corrected. However, there is an underestimation of the volume of the hippocampal fissure by this method (Van Leemput et al., 2009). Therefore I excluded hippocampal fissure from further analysis. Additionally, the sequence that used in this study might be not optimum for the automated segmentation of hippocampal subfields because freesurfer models for both hippocampi were built from manual segmentations of the RHipp in  $0.38 \times 0.38 \times$  $0.8 \text{ mm}^3$  (Van Leemput et al., 2009).

Additionally, the sample size was underpowered for calculation of correlation between some structure measurements and the age at which the person had begun musical training and musical proficiency period. For example, sample size was underpowered for the calculation of the correlation between age of commencing musical training and cortical surface area of the anteromedial portion of the left HG (Power = .62) (Faul et al., 2007, Faul et al., 2009).

## **5.7** Conclusion

Taken together, the results of the current study found that musicians showed increased cortical volume in the right postcentral gyrus and in the left IPL. These regions are part of the auditory-motor coupling network, which indicates that long term practicing may lead to GM adaptation in auditory-motor related regions. Additionally, there were changes in the RHipp surface shape, which indicates change of hippocampal volume according to musical proficiency. Further investigation showed brain-behavioural correlation of fimbria of the RHipp reflecting the importance of internal circuitry of the right hippocampal during PDP. Additionally, brain-behavioural correlations in the RHipp and non-primary auditory areas are irrespective to musical proficiency. Although, musical training could enhance shape and volume of these areas, these changes could be related to predisposition. Only well genetic designed future functional/structural experiments can determine the relative contribution of predisposition and practice.

## 6.1 Aims

Specific aims of this chapter were to:

1- Investigate tonotopic mapping in the human auditory cortex using fMRI.

2- Explore the influence of musical performance expertise and pitch discrimination ability on frequency organization.

**3-** Explore the influence of musical performance expertise and pitch discrimination ability on cortical activation of the human auditory cortex.

# **6.2 Introduction**

The main advantages of using fMRI that brain activation and anatomy can be related directly in individual subjects, and it provides the best spatial resolution. The current study was undertaken to investigate tonotopic mapping in the human auditory cortex using fMRI and to explore the influence of pitch discrimination ability and musical proficiency on frequency organization and cortical activation of the human auditory cortex. Here, I used a stroboscopic event related design that optimised in the pilot study (For details see chapter 3).

## 6.3 Materials and Methods

## 6.3.1 Subjects

Data was collected from 10 musicians (7 males and 3 females) (the mean age: 37.6 years (SD = 16.19) and 13 matched non-musicians subjects (10 males and 3 females) with an age of  $36 \pm 13.59$  (mean  $\pm$  SD) years. Volunteers participated in the study after giving written informed consent. They had no history of any neurological or hearing impairment.

#### 6.3.2 Stimuli

Subjects were binaurally presented with six sine tones varying in frequency (247, 440, 784, 1175, 2637 and 4186 Hz) from different musical notation groups (B3, A4, G5 D6 E7, C8 respectively). Each frequency was formed in a train of sine

waves (rise/fall time 10 ms, plateau 30 ms, interstimulus interval (ISI) 50 ms) as shown in Figure 6.1.



Figure 6.1 Waveform of one of the sine tones (440 Hz).

Tasks were programmed using Eprime (Psychology Software Tools, PST, Pittsburgh, PA). Stimuli were delivered through electrodynamic headphones using a system that included a personal computer, and a commercial MR-compatible device (MR Confon; MR confon GmbH, Magdeburg, Germany). This type of headphone has a high degree of acoustic attenuation to reduce scanner acoustic noise. The system accurately preserves tone frequencies in the range of the stimulation frequencies used (247–4186 Hz). The intensity of the output stimulus was set to 70 dB SPL for all subjects and for all the frequencies to produce maximum neuronal activation (Formisano et al., 2003, Woods et al., 2009).

#### **6.3.3 Experiment Procedure**

All subjects participated in four functional runs, each lasting 9 min, to increase the total number of trials in order to obtain the full sampling of HRF curve and to increase the statistical power in single subject analyses. During the fMRI sessions, subjects were instructed to lie still, listen to the tones and focus on a visually presented cross-hair. The cross-hair was of white colour and presented on a black

background. To maintain a stable level of alertness, subjects were asked to press a button when they saw a circle shape that was presented randomly for 12 times in each run. The circles were presented 6 times during an fMRI session.

#### 6.3.3.1 Functional imaging sessions

Functional imaging was performed using a 3T Trio whole body scanner (Siemens, Erlangen, Germany) with an eight-channel phased-array head coil. FMRI parameters: pulse sequence = EPI; (TR/TE) = 20000/35 ms; delaying TR = 17170 ms; echo spacing 0.7; phase partial Fourier = 6/8; imaging resolution =  $2 \times 2 \times 2$  mm<sup>3</sup>; flip angel= 90°; number of axial slices 28; matrix size 128 × 128; bandwidth 1628 Hz/px. Each EPI measurement was collected at regular 20s intervals with an auditory stimulus being presented at variable offsets (0, 3, 4, 5, 6, 7, 8, and 11 s) before each acquisition to obtain the full sampling of HRF curve (Josephs et al., 1997, Belin et al., 1999). Figure 6.2 shows presentations of (3 s) stimulus at three different time points (11, 7 and 5 s) before each acquisition. This acquisition scheme permits full sampling of HRF for each frequency.





## 6.3.4 Statistical analysis

In the 2nd level analysis of musicianship effect, qualitative comparison was applied by creating percent overlay maps within subjects, then checking the overlay of "fixed effects" within group maps and overlaying single subject maps. Then beta weights were extracted from demarcated regions of interest (HG and PT) by averaging across all voxels in those regions. Alternative methods such as random effects analysis (McNamee and Lazar, 2004, Penny and Holmes, 2007) are not suitable for exploratory investigations in my data because of the small sample size (Seghier et al., 2008). Statistical analyses were performed SPSS (for MAC, Rel18.0. 2010. Chicago: SPSS Inc.) Group comparisons of the estimated beta values were performed using a 2-way ANOVA. P values of ( $\leq 0.05$ ) were considered statistically significant.

#### 6.3.5 FMRI and Anatomical Data Analysis

BrainVoyager QX software package was utilized for single subject fMRI and anatomical image analysis (Brain Innovation, Maastricht, The Netherlands; Version 2.3). Anatomical images were transformed into the Talairach coordinate system. The anatomical data sets were used for automatic cortical surface reconstruction (Kriegeskorte and Goebel, 2001), and inflation. Reconstructed cortical surfaces were then aligned for better match between corresponding brain regions (For details see section 5.3.3.2). The reconstructed cortical surface of the

HG and PT of each subject (For details see section 5.3.3.3) was used for the generation of cortical masks that were used as a constraint for the analysis of functional data sets (Formisano et al., 2002) in order to reduce the number of considered comparisons. Functional data sets were pre-processed with a 3D head motion correction procedure. The functional data sets for each subject were then co-registered to individual 3D anatomical images and inflated aligned brains. This allowed for the viewing of best frequency or tonotopic maps on anatomical data sets. Then the functional data was transformed into Talairach space producing volume time course. Statistical activation maps were obtained by computing a general linear model (GLM), with defining six predictors, one selected for each of those varying pure tones presented during a scanning session. Best frequency maps for each individual were computed by colouring each voxel according to the perceptual frequency using six colours (ranging from dark blue for the highest frequency to red for the lowest one) to which the voxel showed the largest BOLD response (Parkes et al., 2009).

## 6.3.5.1 The exclusion criteria

Eleven out of twenty three subjects (four musicians and seven non-musicians) were excluded from further analysis for different reasons. Three subjects had artefactual responses over large regions Figure (6.3). This exclusion criterion was performed in each subject before applying cortical masks on functional data to exclude subjects whose estimates were heavily biased by non-cognitive effects. I also excluded four subjects whose overall BOLD responses were very weak and

not statistically significant. For each subject activated regions were those regions with activated voxels above a significance of adjusted FDR (Benjamini and Hochberg, 1995, Genovese et al., 2002) P < 0.01.



Figure 6.3 Artefactual responses over large regions due to non-cognitive effects (subject C12).

The FDR method is less conservative than Bonferroni correction, yet it is rigorous enough, especially when the data shows sparse activations. The Bonferroni correction, operating on voxel level, was employed in pilot computations and proved to be too stringent. Some subjects had a very strong response to the tones, with significance of adjusted FDR P < 0.001 across all BOLD responses as shown in Figure 6.4. However, the significance level across frequencies was set to FDR

P < 0.01 for all subjects. Unfortunately, the signal to noise ratio was very weak for some frequencies for four more subjects. Therefore, tonotopic maps were investigated only in six subjects in each group.



**Figure 6.4** Statistical map of the overall auditory cortex response to six frequencies (subject C17) is superimposed on sagittal (LH), coronal and transverse anatomical slices in standard Talairach space, and on a mesh reconstruction of the subject's cortex.

## **6.4 Results**

For each subject, a statistical map of the response of the auditory cortex to all the tones was computed. The activation was mainly located in the HG and secondary auditory cortex (PT).



#### 6.4.1 Influence of musical proficiency on overall activation

Figure 6.5 Means of the effect sizes (beta values) of overall activations in the medial (primary auditory cortex) (mHG) and the lateral (lHG) portions of HG in the LH.

Repeated measures ANOVA showed no significant effect of musicianship on beta values of overall activations in the medial and the lateral parts of HG (F(1,10) = .267, P = .617). Additionally there were hemisphere-group interaction (F(1,10) = 8.155, P = .017) and hemisphere-beta values interaction (F(1,10) = 19.288, P = .001), but no group-beta values interaction (F(1,10) = .599, P = .457) or hemisphere-group-beta values interaction (F(1,10) = 1.258, P = .288), Figure 6.7.

## 6.4.3 Tonotopic Organization



b

**Figure 6.6** Individual frequencies organisation maps of right auditory cortex. Coloured regions are areas that showed significant differences across six tones in the auditory cortex (P<0.01 FDR adjusted). a) Non-musicians, b) Musicians.

Tonotopic maps of a participant from each group are shown in figures 6.6 and 6.7. Frequency gradient maps vary in location, orientations and sizes across all subjects. At least 2 prominent high-frequency areas with a maximal response to frequencies at 4186 and 2637 Hz (dark blue and light blue colours) can be observed medially and laterally or in the middle of HG along an axis roughly parallel to the HG axis in subjects (C04, C07, C13, C14, C15, C17, M3, M6 M07 and M11). With a broad cluster of (440, 784 and 1175 Hz) - orange, yellow and green colours respectively - located over a large portion in the middle of the HG.
CHAPTER 6 Tonotopic Mapping of The Human Auditory Cortex and its Relationship with musical proficiency: an fMRI Study Using a Stroboscopic Event Related Design

Area with a maximal response to frequencies at 247 and 440 Hz (red and orange colours respectively) was situated in the middle of the right HG in subjects (C04, C17, M03 and M07) and the left HG in subjects (C04, C13, C15, M01 and M04). This low frequency region was located in the lateral part of the right HG in subject (C14) and the left HG in subjects (C07, C13, C15, C17 and M01). In contrast, there were single gradient maps in the right PT in subjects (C04, C07, C17, M03 and M11) and the left PT in subjects (C14, C15, C17, M03, M04, M06 and M07).



# b

**Figure 6.7** Individual frequencies organisation maps of left auditory cortex. Coloured regions are areas that showed significant differences across six tones in the auditory cortex (P<0.01 FDR adjusted). a) Non-musicians, b) Musicians.

# **6.5 Discussion**

Although ANOVA test showed no significant effect of musicianship on beta values of overall activations in the medial and the lateral parts of HG, there were hemisphere-group interaction and hemisphere-beta values interaction. In contrast, there were no interaction relations either group-beta values or hemisphere-group-beta values.

In the current study, inter-individual variability was large in frequency gradient maps in location, orientations and sizes. These considerable variations have been reported in most previous studies (Formisano et al., 2003, Talavage et al., 2004, Seifritz et al., 2006, Woods et al., 2009, Humphries et al., 2010, Da Costa et al., 2011, Langers and van Dijk, 2011). This inconsistency of maps can be attributed to attributed to the individual gyral pattern of the position and the anatomy of HG and PT (Penhune et al., 1996, Hackett et al., 2001, Rademacher et al., 2001). Another possible source of variance are the differences in the physiologic state changes.

In some subjects (C07, C13, C14, C15, C17, M03, M06, M07 and M11 "right HG") and (C04, C13, C15, C17, M01, M06, M07 and M11 "left HG"), mirrorsymmetric frequency gradients extending from a region in the middle of the HG that was most sensitive to lower frequencies. These gradients extended posteriormedial and anterior-lateral along the HG axis are similar to those observed by Formisano et al., (2003). These multiple frequency gradients could help explain the first stage of parallel processing of auditory information (Formisano et al., CHAPTER 6 Tonotopic Mapping of The Human Auditory Cortex and its Relationship with musical proficiency: an fMRI Study Using a Stroboscopic Event Related Design

2003). I propose that there are multiple auditory fields in the HG and the PT which overlap in time representing a combination of results from two MEG studies (Pantev et al., 1995, Schneider, 2001) and supported by Liegeois-Chauvel et al., 1994, Gutschalk et al., 1999 and Lütkenhöner et al., 2003a.

Specifically, frequency maps of subject C17 is very similar to those described in macaque monkeys (Petkov et al., 2006, Petkov et al., 2009). It is clearly shown in Figure 3.7 (page 61). In their data, a low frequency band was located in the centre of the core area, similar to low frequencies areas roughly occupying the middle part of HG of subject C17. High frequency regions present immediately anterior and posterior to low frequency band.

### **6.6 Limitations**

The total acquisition time of each run was long, taking into consideration that each scanning slot consisted of four runs. Therefore inter-runs and inter-subjects variability was seen in the number of significantly activated voxels and their preferred frequencies. Therefore, the maps in this study did show a high degree of variability across subjects and across hemispheres and hence, in the analysis of tonotopic organization of auditory cortex, no group comparison was done. There are some possible factors could cause those variances such as amount of head movement and alterations in attention and alertness. The latter factors affected results of this study to a great extent. Additionally, stroboscopic event related design has a lower statistical power compared with other designs (Rosen et al., 1998, Amaro et al., 2002). This design also needs longer time and multiple runs to

CHAPTER 6 Tonotopic Mapping of The Human Auditory Cortex and its Relationship with musical proficiency: an fMRI Study Using a Stroboscopic Event Related Design

obtain the full sampling of HRF curve.

A number of datasets were excluded because of either their artefactual responses over large regions (Figure 6.3) or their response estimates were very weak and not statistically significant. Insignificant statistic response estimates could be related to reduction of attention and alertness level due to running multiple scans on the same subject in the same day (running fMRI experiments directly after a MEG scan). Taking in consideration, that MEG part lasted for more than two hours.

# 6.7 Conclusion

Although, ANOVA test of Influence of musical proficiency on overall activation showed no significant effect in the medial and the lateral parts of HG, there were hemisphere-group interaction and hemisphere-beta values interaction. Due to inter-individual variability of the tonotopic mapping of this study, group comparison was not possible. In most subjects, I observed gradients along the axis of HG similar to those produced by Formisano et al., (2003) but most recent studies (Woods et al., 2009, Humphries et al., 2010, Da Costa et al., 2011, Langers and van Dijk, 2011) showed gradients perpendicular to the long axis of HG. This discrepancy with the most recent studies could be partly due to differences in the stimuli and range of frequencies used. It could be related as well to the different methods that have been used. I used stroboscopic event related design which has been used by Formisano et al., (2003) while Humphries et al. (2010), Langers and van Dijk (2011) and Woods et al. (2009) used sparse temporal sampling method.

# Chapter 7: Tonotopic Mapping of the Human Auditory Cortex and its Relationship with Musical Proficiency: an MEG Study

## 7.1 Aims

Specific aims of this chapter were to:

1- Investigate tonotopic mapping in the human auditory cortex using MEG.

2- Explore the influence of musical performance expertise and pitch discrimination ability on frequency organization of the human auditory cortex.

**3-** Explore the influence of musical performance expertise and pitch discrimination ability on cortical activation of the human auditory cortex.

# 7.2 Introduction

MEG has been used to investigate tonotopic mapping of the auditory cortex in humans since 1982 (Romani et al., 1982a, Romani et al., 1982b). Most studies used different experimental designs based on the type of neuromagnetic response (For details see section 2.4.1.3).

#### 7.2.1 Auditory evoked fields in musicians and non-musicians

Several studies showed that some components of evoked auditory potentials can be enhanced in musicians or by musical training. Musicians manifested larger N1c (peak at 138 ms) and P2 (peak at 185 ms) responses to all types of tonal stimuli (violin tones, piano tones, and pure tones) (Shahin et al., 2003). Pantev et al. (1998) investigated the evoked magnetic fields of a pseudorandom sequence of piano and pure tones matched in frequency and loudness in musicians and non-musicians. The only difference between musicians and non-musicians was in the strength of cortical activation for piano tones, which was 25% greater in musicians. Musicians also showed a strong correlation with the age of commencement of musical practice. Schneider et al. (2002) investigated the processing of sinusoidal tones in the auditory cortex of non-musicians, professional musicians and amateur musicians. The authors found neurophysiological differences between groups. Comparing professional musicians and non-musicians, the activity evoked in primary auditory cortex (AEMLF) was 102% larger in professional

Chapter 7 Tonotopic Mapping of the Human Auditory Cortex and its Relationship with Pitch Discrimination Ability: A Magnetoencephalography Study

musicians. Pantev et al. (2001) showed that musical education and training is reflected in the organization of auditory and somatosensory representational cortex in musicians. In addition, musicians had larger N2b (peak at 175 ms) and P300 (peak at 300 ms) amplitude during attentive listening (Tervaniemi et al., 2005).

The effect of auditory discrimination has been studied by Bosnyak et al. (2007) using single carrier frequency for 10 training sessions, alternating training and passive control blocks to explore whether competitive interactions among frequency representations in both primary and secondary auditory cortices. P2 (peak at 200 ms) amplitude increased across sessions on both passive listening and training blocks. In contrast, steady-state response amplitude was larger in training compared to passive blocks. Fujioka et al. (2006) reported that musically trained children had larger and earlier N250m peak in the LH in response to the violin sound compared with untrained children.

The current study was undertaken to investigate tonotopic mapping in the human auditory cortex using MEG and to explore the influence of pitch discrimination ability and musical experience on frequency organization and cortical activation of the human auditory. An additional aim was to investigate any hemispheric difference of tonotopic mapping of the human auditory cortex.

### 7.3 Materials and Methods

#### 7.3.1 Subjects

Data was collected from 11 musicians (7 males and 4 females) with an average age of an age of  $35 \pm 14.09$  (mean  $\pm$  SD) years and 14 matched nonmusicians subjects (10 males and 4 females) (the mean age: 38 years (SD = 11.59). Volunteers participated in the study after giving written informed consent. They had no history of any neurological or hearing impairment.

#### 7.3.2 Stimuli

Six sine tones of varying frequencies (247, 440, 784, 1175, 2637 and 4186 Hz) were applied binaurally. The tones belonged to different musical notation groups (B3, A4, G5 D6 E7, C8 respectively). Tasks were programmed using Eprime (Psychology Software Tools, PST, Pittsburgh, PA). Stimuli were delivered through plastic tubes attached to ear tips using a system that included a personal computer, a multi-channel analogue amplifier. The system accurately preserves tone frequencies in the range of the stimulation frequencies used (247–4186 Hz). The intensity of the output stimulus was set to 60 dB SPL for all subjects and for all the frequencies. Each frequency was formed in 150 ms sine wave (rise/fall time 20 ms, plateau 110 ms) with an interstimulus interval randomized between 300 and 500 ms.

Chapter 7 Tonotopic Mapping of the Human Auditory Cortex and its Relationship with Pitch Discrimination Ability: A Magnetoencephalography Study

#### 7.3.3 Experimental procedure

All subjects participated in 6 blocks of 10 min duration each. In every block, 200 tones of each frequency were presented. During the sessions subjects were instructed to lie still, listen to the tones and focus on a visually presented cross-hair. The cross-hair was of white colour and shown on black background. There was a 2 min break between successive blocks.

Whole- head MEG was recorded using 148 axial magnetometers encased in a plastic helmet (Magnes 2400, 4D Neuroimaging, San Diego, USA). The bandpass filter was 0–200 Hz, and the sampling rate was 640.78 Hz. Before the experiment, the 3-D positions of three fiducials (nasion and the left and right pre-auricular points) and five head coils were measured using Polhemus system (Polhemus Navigation, Inc., Colchester, USA). Vertical electrooculogram was recorded using silver electrodes filled with conductive gel and placed above and below the right eye. The electrocardiogram was recorded by two silver cup electrodes placed on the left and right shoulder.

#### 7.3.5 Statistical analysis

Statistical analyses were performed using SPSS (for MAC, Rel18.0. 2010. Chicago: SPSS Inc.) Group comparison of the dipole amplitude of N100m response was performed using repeated measures analyses of variance (ANOVA). Parametric correlations analysis were performed between frequency gradient and x, y and z axes of Talairach coordinates in both hemispheres. P values of ( $\leq 0.05$ ) were considered statistically significant. Bonferroni multiple comparisons correction were performed on all p values (Salkind and Rasmussen, 2007).

### 7.3.6 MEG Data Analysis

MEG data were analysed using the Brain Electrical Source Analysis v. 5.1.8 program (BESA, Megis, GmbH, Germany). MEG signals were visually inspected to identify head motion artefacts and dead channels. Electrocardiographic and electrooculographic artefacts were removed by identifying their patterns and averaging all events of these patterns in MEG data. Clean MEG data were cut into epochs corresponding to 100 ms of the pre-stimulus and 400 ms of the post-stimulus timeframe. Triggers identifying the occurrences of individual stimuli were given order numbers so as to allow selective averaging of epochs. MEG data were averaged for all stimuli encompassing up to 1200 events for each frequency, then the averaged MEG waveforms were analysed using source dipole analysis in BESA program. Source reconstruction was based on a spherical head model. The digitized head shape of the individual was used to adjust the size of the sphere in every subject. The main source activity of N100m was analysed after applying zero phase shift band-pass filter from 3 Hz (6 dB/octave) to 30 Hz (12 dB/octave) (Schneider, 2001) then modelled with two equivalent dipoles -one in each hemisphere- located in the PT posterior to HG. Additionally, I analysed

Chapter 7 Tonotopic Mapping of the Human Auditory Cortex and its Relationship with Pitch Discrimination Ability: A Magnetoencephalography Study

earlier components of auditory evoked fields, such as N19m–P30m, which situated in HG using different filter values (20–120 Hz zero phase bandpass). Different patterns of equivalent dipole source (e.g. locations and orientations) have been estimated using source- fitting method. I exclude earlier components from further analysis because of difficulty of localising dipoles within HG in most subjects. This issue is related to limited number of events for each condition. The resultant dipole sources were transformed into Talairach space using BrainVoyager QX software package (Brain Innovation, Maastricht, The Netherlands; Version 2.3) as explained in section 5.3.3.2. Flux maps were used to visually view the spatial pattern of the activity observing apparent sources over the epoch of interest.

#### 7.4 Results

# 7.4.1 Influence of musical performance on cortical activation

The N100m component was enhanced in musicians compared to nonmusicians as shown in flux maps of grand average evoked fields (Figure 7.1). Visual inspections of flux maps within each group shows that middle range frequencies were greater than the highest and lowest frequencies. Activation of 1175 Hz is the greatest in both groups (Figure 7.1). Flux maps were larger in the LH than in the RH in both groups. Musicians had larger flux maps of 440, 784 and 1175 Hz than non-musicians while non-musicians had larger map of 2637 Hz.



Chapter 7 Tonotopic Mapping of the Human Auditory Cortex and its Relationship with Pitch Discrimination Ability: A Magnetoencephalography Study

Figure 7.1 Magnetic flux contour maps at the peak latency of N100m of grand average evoked fields in musicians and non-musicians.

#### 7.4.1.1 Dipole magnitude of cortical responses

The individual N100m baseline to peak dipole amplitudes are shown in Figure 7.3. There is a large variance in each group making it difficult to analyse the data, therefore I analysed the grand average of each group. Overall, source activity was largest around 1175 Hz in the LH as shown in Figure 7.4. In both groups, source activity was larger in left hemisphere within med range frequencies (440, 1175 and 2637 Hz).



Chapter 7 Tonotopic Mapping of the Human Auditory Cortex and its Relationship with Pitch Discrimination Ability: A Magnetoencephalography Study

Figure 7.2 The baseline to peak dipole amplitude of the N100m response of both RH and LH of individual, (7 musicians, 7 non-musicians).

Figures 7.2–7.4 show that, musicians had larger amplitude of the N100m component at most frequencies in both hemispheres compared to nonmusicians. However, ANOVA for repeated measurements showed no significance effect of musicianship (P=.07). Additionally there were no interaction effects either frequency-group interaction (P=.953), hemispheregroup interaction (P=.909), or even frequency-hemisphere interaction (P=.977).





Figure 7.3 The baseline to peak dipole amplitude of the N100m response of the grand average of each group. The error bars indicate the standard error.



Figure 7.4 Means of the dipole amplitude of N100m response of the grand average of each group.

# 7.4.2 Latency

Figure 7.5 shows that tonotopic mapping was reflected in the frequency dependence of the latency (time from stimulus onset to peak) of the N100m peak. The N100m peak occurred at about 120 ms with the lowest frequency and at around 95 ms at 2637 Hz.



Figure 7.5 The latencies of N100m peaks of musicians and non-musicians are strongly frequency dependent. The error bars indicate the standard error.

### 7.4.3 Tonotopic organisation

Figures 7.6 shows tonotopic mapping of non-musicians and musicians groups. Each individual in both groups showed a strong tonotopic gradient of

N100m components along the medial-lateral axis of the PT in both hemispheres with no left-right asymmetry. This gradient was located in the lateral part of PT in most cases. I observed that the source of N100m components shifted laterally by increasing frequencies with an average about 1.5 mm per octave. Dipoles were colour-coded using six colours (ranging from dark blue for the highest frequency to red for the lowest one).

Additionally, N100m components of grand average of each group have been investigated. Tonotopic gradient was preserved in both groups as shown in (Figures 7.7–7.10) Visual inspections of tonotopic maps showed no differences between musicians and non-musicians. Performing parametric correlations analysis showed that both groups demonstrated positive and strong correlations between frequency gradient and the gradient along the medial-lateral axis of the PT only in both hemispheres (Table 7.1).

Group		Hemisphere	
		Right	Left
Musicians	Pearson Correlation	.940	920
	Sig. (2-tailed) N (6)	.005	.010
Non-musicians	Pearson Correlation	.945	909
	Sig. (2-tailed) N (6)	.004	.012

 Table 7.1 Correlations between frequency gradient and the gradient along the medial-lateral axis of the PT, P values were corrected using Bonferroni adjustment.



Figure 7.6 Individual tonotopic mapping of the N100m components in non-musicians (letter C) and musicians (letter M). Dipole loclisations are superimposed on coronal and transverse slices in standard Talairach space.



Figure 7.7 Tonotopic mapping of the N100m components in musicians. Dipole localisations are superimposed on coronal and transverse slices in standard Talairach space of an average volume of all participants.



Figure 7.8 Tonotopic mapping of the N100m components in non-musicians. Dipole localisations are superimposed on coronal and transverse slices in standard Talairach space of an average volume of all participants.



Figure 7.9 Tonotopic mapping of the N100m components in the PT are presented along x, y and z axis of Talairach coordinates in both hemispheres. The error bars indicate the standard error.

Chapter 7 Tonotopic Mapping of the Human Auditory Cortex and its Relationship with Pitch Discrimination Ability: A Magnetoencephalography Study



Figure 7.10 Tonotopic mapping of the N100m components of grand average of each group are displayed on mesh reconstructions of the right and left auditory cortex of one subject's (subject C17).

Chapter 7 Tonotopic Mapping of the Human Auditory Cortex and its Relationship with Pitch Discrimination Ability: A Magnetoencephalography Study

#### 7.5 Discussion

Results showed that musicians had a greater neuromagnetic activation of most frequencies than non-musicians as shown in flux maps (Figure 7.1). Visual inspection of Figure 7.4 showed that musicians had greater amplitude of N100m components at 1175 Hz in the LH. However, the ANOVA for repeated measurements showed insignificant group effect (p=.07) with no interaction effects either frequency-group interaction, hemisphere-group interaction or even frequency-hemisphere interaction (Figure 7.4). This finding is consistent with some previous studies (Pantev et al., 1998, Menning et al., 2000, Schneider et al., 2002). Menning et al. (2000) showed that N100m source strength of a pure tone of 1kHz increased over intensive frequency discrimination training period. Additionally, Pantev et al. (1998) reported an increase of cortical response of N100m of piano tones but not pure tones in musicians comparing with non-musicians with a significant correlation between the mean N100m dipole moment and the age where musical training started.

The tonotopic mapping was reflected in the frequency dependence of the latency of the N100m peak. The N100m peak occurred at about 120 ms with the lowest frequency and at around 95 ms at 2637 Hz. This could be related to the running time of the travelling wave along the basilar membrane (Schneider, 2001).

In this study, I could not obtain stable response of middle latency auditory evoked fields therefore; cortical activation was mainly in the PT. I observed a tonotopic gradient of N100m components in the lateral part of the PT with no significant Chapter 7 Tonotopic Mapping of the Human Auditory Cortex and its Relationship with Pitch Discrimination Ability: A Magnetoencephalography Study

group effect, where the highest frequency was located more laterally and the lowest one was situated more medially with a gradient average of 1.5 mm per octave. Results of this study are consistent with the tonotopic mapping of N100m response of (Schneider, 2001).

The gradient of N100m components is inconsistent with frequency gradients in the PT that resulted from fMRI part of this thesis. It might be explained by methodological differences or due to the large inter-individual variability especially in fMRI study and the comparatively small number of subjects precluded both averaging over subjects and comparison of fMRI and MEG tonotopic maps. Additionally, systematic error related to co-registration of averaged MEG data to an averaged brain could contribute in that variation.

# 7.6 Limitations

There are two noteworthy limitations regarding the present study. Firstly, the auditory evoked middle latency fields were very weak in most subjects because the total number of stimuli presented was low (in the range of 900 to 1000 per condition). Previous studies presented larger number of stimuli per condition - 3500 (Lütkenhöner and Steinstrater, 1998), 1500 (Pantev et al., 1995) and 12000 (Schneider, 2001)- to gain an enhanced response of the primary components. Primary components are very important because they are originated within the medial portion of the HG.

Another limitation was excluding data sets (4 musicians and 8 non-musicians)

Chapter 7 Tonotopic Mapping of the Human Auditory Cortex and its Relationship with Pitch Discrimination Ability: A Magnetoencephalography Study

from further analysis due to severe tilting of sensor locations as shown in Figure 7.11. In addition, technical abilities of the MEG system at the University of Liverpool limited the number and range of frequencies used.



Figure 7.11 Illustration of sensor tilting problem.

**Tilted sensors** 

# 7.7 Conclusion

In conclusion, musicians had greater amplitude of N100m component especially at 1175 Hz with no interaction effect. This shows that N100m components for pure tone can be used as a comparison tool for the neural response but only at a limited range of frequencies. Both groups demonstrated positive and strong correlations between frequency gradient and the gradient along the medial-lateral axis of the PT in both hemispheres The gradient of N100m components is inconsistence with fMRI part of the main study.

# Chapter 8 : Summary, discussion and conclusion

# 8.1 Introduction

In this thesis, the main aim was to investigate the effect of musical proficiency on functional neural substrates of sine tone frequency processing and anatomical morphology in the human auditory cortex using multimodal neuroimaging methods (structural MRI, fMRI and MEG). I also explored neuroanatomical features in brain regions that are associated with pitch memory (IPL and hippocampus) with taking in consideration the effect of type of instrument on sulcal and gyral topography.

I assessed the possible links between pitch discrimination ability test scores (as defined in Chapter 4) and neuroanatomical features e.g. GM density, cortical surface area and volume (see Chapter 5).

#### 8.2 Results and Interpretations

The results that I obtained for each study were described and interpreted separately within each chapter. The following subsections were related to the interpretation of results for analyses based around individual objectives.

# 1- Investigating PDP in musicians and non-musicians (chapter 4):

Musicians had better performance in pitch discrimination than non-musicians in all three conditions (both ears together, the right and left ears). In musicians, a statistically significant difference in PDP was observed with the left ear performing better than the right ear. This result might suggest that musicians had right-hemisphere dominance for pitch discrimination. Additionally, I may consider that musicians' performance of the LE was related to the duration of musical proficiency.

# 2- Frequency organisation of the human auditory cortex (chapters 6 and 7):

In the fMRI study, group comparison was not possible because of inter-individual variability of the tonotopic mapping. In most subjects, I observed gradients along the axis of HG similar to those produced by Formisano et al., (2003). Formisano et al. (2003) suggested three different explanations for mirror-symmetric of frequency organisation of the human auditory cortex; a) it could be a presentation of two primary cortical fields that located closely to each other, b) tonotopic gradient of this mapping is oriented in opposite directions, and c) There is an overlapping between two fields in time. The authors supported first and second suggestions. Additionally, gradient of N100m components which occupied the lateral part of PT. In this study, I suggested that there were multiple auditory fields overlapping in time and perhaps representing combination of results from two MEG studies (Pantev et al., 1995, Schneider, 2001) and supported by (Liegeois-Chauvel et al., 1994, Gutschalk et al., 1999, Lütkenhöner et al., 2003a).

# 3- Neuroanatomical/functional correlates of musicianship in the human auditory cortex (chapters 5, 6 and 7):

Musicians had larger cortical surface area in the anteromedial portion of the left HG compared to non-musicians. Functionally, ANOVA of repeated measures of the amplitude of hemodynamic responses to neural activity of the medial and the lateral parts of HG showed hemisphere-group interaction and hemisphere-beta

values interaction. Functional correlates of musicianship in anteromedial portion of HG might be explained by the structural correlate of musicianship and hence the number of neurons that contributed in processing auditory stimuli. Furthermore, neuromagnetic activation showed musicianship effect of borderline significance on the amplitude of the N100m component. These results indicate the importance of the morphology and neurophysiology of auditory cortex in musical proficiency (Schneider et al., 2002).

# 4- Neuroanatomical correlates of PDP in the human auditory cortex (chapters 5 and 6):

There were positive correlations between PDP of the right ear and both cortical surface area in the left STG and the lateral aspect of left HG. PDP of the right ear was also significantly related to GM volume of the lateral aspect of left HG. The anatomical correlates of PDP of the right ear in the lateral part of left HG might suggest that increasing number of synchronised neurons that involved in pitch processing may improve PDP and reflecting its importance in pitch processing (Griffiths et al., 1998a, Griffiths et al., 2001, Patterson et al., 2002, Griffiths, 2003, Krumbholz et al., 2003, Warren and Griffiths, 2003).

# 5- Neuroanatomical correlates of musicianship and PDP in IPL and postcentral gyrus (chapter 5):

Musicians had larger GM distributions than non-musicians in the right postcentral gyrus and in the left IPL. Differences of sulcal and gyral topographic patterns indicate that musicians had deeper and thicker cortex than non-musicians in various sulci and gyri respectively such as right postcentral and the right and left

cingulate sulci, and the left parahippocampal gyrus, the posterior part of the left Broca's area and the left and the right IPL. In contrast, non-musicians had deeper cortex than musicians in the following sulci: posterior part of the left superior frontal, right superior temporal and left inferior temporal sulci. Non-musicians also had thicker cortex than musicians in the following gyri: the anterior part of left middle frontal gyrus, and the right gyrus rectus.

Sulcal and gyral topographic features in musicians may reflect multi-modal integration of auditory, motor, and cognitive processes in response to daily extensive requirements to perform music at professional level. Limiting my results to those regions that had significant difference in both anatomical methods (VBM and sulcal and gyral topographic patterns), GM distributions and sulcal and gyral topographic patterns), GM distributions and sulcal and gyral topographic patterns results agree with each other in the left IPL and the right postcentral gyrus. These regions are considered as part of the auditory-motor coupling network (Gaab and Schlaug, 2003, Bangert et al., 2006, Gaab et al., 2006).

Larger GM density in the right angular gyrus in musicians compared to nonmusicians were further explored by covarying with PDP of the left ear. Angular gyrus, which takes part in pitch short memory (Binder et al., 1997, Rinne et al., 2009), is considered as part of the semantic memory network (Platel et al., 2003, Platel, 2005, Groussard et al., 2010b).

# 6- Neuroanatomical correlates of musicianship and PDP in the right and left hippocampi (chapter 5):

It has been shown that there is a relation between musicianship and the surface shape of the RHipp indicating that musicians had thicker lateral anterior and the most posterior regions of the RHipp compared to non-musicians. Further investigation showed no significance difference between musicians and nonmusicians in the subfields volume in both hippocampi with group-hemisphere interaction and a significant difference between two hemispheres. Post hoc t-test showed that musicians had significantly larger volume than non-musicians in fimbria, subiculum and presubiculum of the right hippocampus. In contrast, nonmusicians had significantly increased volume of CA4/DG in the left hippocampus compared to musicians.

When PDP was included as a covariate in the VBM analysis, musicians had larger GM density compared to non-musicians in the posterior portion of hippocampus (particularly subiculum), which relatively more concerned with space and memory (O'Mara et al., 2009). In addition, volume of fimbria region in the RHipp correlated positively with PDP of the left ear reflecting the importance of hippocampal internal/external circuitry in pitch memory and auditory temporal information processing. It has been shown that the RHipp is associated with pitch memory and processing of auditory temporal information in animal (Sakurai, 2002) and human (Watanabe et al., 2008).

Covarying with PDP of the left and right ear in VBM study showed an associated increase in GM density in the RHipp and the right and left angular gyrus

respectively may suggest a network supporting the spatial processing required for pitch memory.

## 7- Brain effects of type of instrument (chapter 5):

Sulcal and gyral topographic patterns indicate considerable within-musician differences. String players (6 musicians) had prominent omega shape (Yousry et al., 1997) of central sulcus in the right hemisphere\_only, while the non-string players (6 musicians) preserved omega shape in both hemispheres (Bangert and Schlaug, 2006). These results reveal an association between acquired sensory-motor skills and features of brain topographic patterns. This has been reported previously in animals (Kleim et al., 1996, Anderson et al., 2002) and humans (Bengtsson et al., 2005, Bangert and Schlaug, 2006, Hyde et al., 2009).

#### 8.3 General discussion

# 8.3.1 Multimodal imaging approach

Although, neuroimaging is a very young science, multimodal neuroimaging has already advanced our understanding of the spatiotemporal pattern of brain activation and connectivity underlying perception, motion and cognition. Combining fMRI and MEG offers a great tool for spatiotemporal mapping of brain activity (Dale et al., 2000, Dale and Halgren, 2001, He and Liu, 2008). In this thesis, I implemented both methods to investigate spatiotemporal excitation patterns of frequency organisation and simple tone pitch perception in the auditory cortex and to understand the neural basis of mirror-symmetric of frequency organizations of the human auditory cortex. In MEG part, unfortunately I could

not obtain stable response of middle latency auditory evoked fields, which situated in the HG; therefore cortical activation was mainly in the PT. Therefore it was not possible to examine spatiotemporal excitation patterns of frequency organization in the auditory cortex. I found that frequency organizations of N100m components occupied lateral part of frequency gradient that resulted from fMRI, which extended along the lateral-medial axis of the PT. This might be explained by assuming that there is an involvement of multiple areas in the PT in the generation of auditory fields, which is supported by previous MEG studies (Pantev et al., 1995, Lütkenhöner and Steinstrater, 1998, Schneider, 2001). However, the inconsistency in MEG and fMRI results could be also explained by methodological differences that related to the differences in physiological and physical rationales of both imaging modalities.

The detectable change of BOLD signal should last at least thousands of milliseconds to become observed by MRI, while MEG and EEG could detect shorter cortical activity. Exciting a small percentage of neurons could produce detectable electromagnetic signal and could lead to an unobserved change in metabolic consumption by fMRI. Spike timing-dependent synaptic plasticity is considered as linking patterns of pre- and postsynaptic activity to changes in synaptic strength (Froemke and Dan, 2002, Dan and Poo, 2006). Synaptic strength modifications alter representational patterns or reorganise neural connections in morphological or functional domains. MEG and EEG may detect early plasticity changes related to strengthening of synaptic but MRI could explore long-term brain plasticity reveal significant volumetric/regional changes. However diffusion

MRI can detect structural changes in cell morphology induced by plasticity within few hours (Le Bihan et al., 2006, Blumenfeld-Katzir et al., 2011). In this thesis, both fMRI and MEG revealed sort of asymmetric functional differences in the auditory cortex between musicians and non-musicians. Various anatomical techniques revealed significant difference in various cortical features in different brain regions between musicians and non-musicians. Results of anatomical studies strongly agree with each other.

#### 8.3.2 Is it genetic predisposition for music?

Genetic and environmental factors influence brain structure and function to a great extent (Thompson et al., 2001, Brun et al., 2009, Jahanshad et al., 2010, Thompson et al., 2010, Kochunov et al., 2011, Yoon et al., 2011). Several studies showed the environmental and personal experience effect on structural characteristic of some brain regions (Maguire et al., 2000, Mechelli et al., 2004, Lazar et al., 2005, Aydin et al., 2007).

The main question remains whether structural and functional changes in musicians' brains are associated with musical training (Hyde et al., 2009, Herdener et al., 2010) or represent a genetic predisposition (Thompson et al., 2001, Jahanshad et al., 2010) for music?. The first factor is supported by a considerable literature. The structural and functional correlates of musicianship are highly correlated with the duration of musical proficiency and intensity of musical practice (Sluming et al., 2002, Hutchinson et al., 2003, Aydin et al., 2005, Abdul-Kareem et al., 2011b), with the age at which the person had begun musical

training (Elbert et al., 1995, Schlaug et al., 1995a, Monaghan et al., 1998, Pantev et al., 1998, Ohnishi et al., 2001, Gaser and Schlaug, 2003, Lotze et al., 2003, Bengtsson et al., 2005) and musical aptitude (Schneider et al., 2002, Gaser and Schlaug, 2003, Lotze et al., 2003). Additionally, the literature and this study showed that structural differences are directly related to type of instrument (Elbert et al., 1995, Schneider et al., 2005, Bangert and Schlaug, 2006). Pantev et al. (2003) reported functional correlates of the type of instrument. Investigation of the functional effect of type of instrument was beyond this thesis as I used simple tone stimuli during functional neuroimaging tasks. Furthermore, induced structural and functional plasticity were observed in several longitudinal studies (Jäncke et al., 2001, Pascual-Leone, 2001, Gaab et al., 2006, Lappe et al., 2008, Hyde et al., 2009, Herdener et al., 2010). The results of this thesis are consistent with some of sited studies above. Although, relation between structural differences and type of instrument has been observed effect of the duration of musical proficiency, intensity of musical practice and age of commencing musical training were not statically significant. This could be related to the lack of power and intersubject variability and will be considered further in the limitation section.

In this thesis, structural correlates of PDP, which were irrespective to musical proficiency, were found in various brain regions. This finding could support the assumption of the influence of environmental and genetic factors on particular brain regions (Lenroot et al., 2009, Deary et al., 2010, Davies et al., 2011). Exploring environmental and gentic effect is beyond the aims of this thesis.

#### 8.4 Limitations

One of the limitations is that data size is too few to accurately account for measurements of some factors such as the age at which the person had begun musical training and musical proficiency period. For example, sample size was underpowered for the calculation of the correlation between age at which musicians begun to practise and PDP of the LE (Power = .52) (Faul et al., 2007, Faul et al., 2009). In contrast, power of the data was fairly enough in some other measurements e.g. Power of fMRI study was 0.9 (Desmond and Glover, 2002, Hayasaka et al., 2007). Additionally, this study might be affected by intersubject variability e.g. within the whole sample there are five and four females in musicians and non-musicians respectively and one left handed participant in each group. That may explain why this study is not fully consistent with some previous studies. Specific limitations of each method were discussed in sections (5.5, 6.5 and 7.5).

#### 8.5 Conclusion

In this thesis, I investigated PDP and brain structure and function in a group of musicians in comparison with age, sex and handedness matched non-musicians. Different neuroimaging methods and image analyses approaches have been employed to study structural and functional difference in the auditory cortex and some brain regions that associated with pitch memory specifically hippocampus and IPL.

A statistically significant region of increased GM distribution was found in the right postcentral gyrus and in the right and the left IPL of musicians. IPL forms triangle-like functional structure with Broca's and Wernicke's areas (Catani et al., 2007). These regions are important for music and language (Zatorre et al., 1992, Binder et al., 1997, Koelsch et al., 2002a, Sluming et al., 2002, Tillmann et al., 2003) with taking in consideration the evidences of neural specializations for music and speech (Peretz et al., 1994, Tervaniemi et al., 1999, Maess et al., 2001, Zatorre et al., 2002a, Abrams et al., 2011, Rogalsky et al., 2011).

My findings of the difference in some brain structures between both groups might reflect multi-modal integration demands of auditory, motor, and cognitive tasks in response to daily extensive requirements for multisensory process; e.g. pitch perception, playing on instruments, remembering and musical sight-reading. In contrast, the functional and structural behavioural correlates in some brain regions that associated with pitch processing and memory, which were present irrespective of musical proficiency. Structure and function of these regions play essential role in musical proficiency, however these correlates could be related to innate predisposition (Thompson et al., 2001, Carmelli et al., 2002, Toga and Thompson, 2005).

Significant differences in brain structure were identified in regions that have functional relevance to musical proficiency and pitch discrimination ability with all techniques of image analyses employed. These differences suggest an indication of use-dependent adaptation of cortical tissue in reaction to the lifestyle
of professional musicians. Knowledge of training-induced anatomical and functional neuroplasticity might be applied to neurorehabilitation strategies in neurologically impaired patients (Jenkins, 2001, Flor et al., 2004, Schneider et al., 2007, Herraiz et al., 2009, Schlaug et al., 2009b). Early musical training may have transferable benefits which contribute towards achievement in verbal memory (Chan et al., 1998, Ho et al., 2003), verbal intelligence (Moreno et al., 2011), mathematical enhancement (Rauscher et al., 1997, Graziano et al., 1999) and fine motor skills (Forgeard et al., 2008).

## 8.6 Future work

Additional work on larger sample size is needed including performing longitudinal, multimodal and behavioural studies to separate apart the genetic effects from those of musical training. Designing specific spectral/temporal pitch processing/memory paradigms using fMRI and MEG in combination with ultra high resolution DTI and MRI would provide clear image about role of brain regions and the brain internal circuitry that involved in pitch processing and pitch memory.

This thesis could be considered as the first implementation of brain topographic patterns and shape change of subcortical regions in studying neuroanatomical correlates of musicianship. Combination of these methods is important to find out at which extent they can be used as disease indications such as Alzheimer senile dementia (Li et al., 2007, Tondelli et al., 2011) and schizophrenia (Csernansky et al., 2008).

Language and music share many common features (Patel et al., 1998, Patel, 2008, Fedorenko et al., 2009, Shahin, 2011). Therefore musical perception and processing may be relevant for understanding the neural bases of language only in specific cases such as the higher-order cognitive processes (Rogalsky et al., 2011) and complex sound structures (phonology) (Brown et al., 2006). Also musical training could help in extracting prosodic information from speech (Thompson et al., 2003, Thompson et al., 2004).

Distinctive neural processing is applied in human brain to process different languages. During reading tasks, native speakers of the Arabic language showed specific right hemisphere disability for Arabic, but not for Hebrew or English (Ibrahim and Eviatar, 2009). The authors used the divided visual field technique on native Arabic speakers who use Hebrew and English as second language. Arabic and English languages differ in reading direction, orthographic complexity (extensive use of dots and many letters with an identical structure in Arabic script) and morphological structure, which could cause the specific difficulty of the right hemisphere with Arabic. Boudelaa et al. (2010) investigated Arabic morphology using MMN. The authors reported that topographically, the root MMN has a symmetric fronto-central distribution, whereas the word pattern MMN lateralizes significantly to the left. It is really important to apply neuroimaging methods extensively in studying Arabic language, as neural basis of Arabic language is still not clear. This could help in knowing how a stroke affects language recovery in Arabic speakers and in diagnosing some learning and language disorders in

Arabic language e.g. aphasia, dyslexia and dysgraphia. Additionally, understanding neural basis of Arabic language would be essential in defining learning strategies especially for children.

In Arabic language, sounds are closely fused with meanings. This strong connection between sounds and meanings is one of the beauty and affectivity of Classical Arabic texts including Holy Qur'an. Arabic poetry is the main form of Arabic literature especially during pre-Islamic era. At the beginning of Islamic period, the Holy Qur'an reinforced phonologic and prosodic patterns in Arabic language. Al-Khalīl ibn Ahmad al-Farāhīdī (d.ca. A.H. 175/A.D.791) invented science of "Arūd" (prosody) depending on rhythmic nature of pre-Islamic poetry (Ryding, 1998). He wrote fifteen "Buhūr" (verse types) to measure the poems' structure imitating musical "iqa" (rhythmic mode). In most cases, Arabic poets start producing poems at the early age and continue production process until they die. Professional readers of the Holy Qur'an usually memorise the whole book (more than 600 page) before the age of ten. They read it from memory in the daily basis in a particular phonetic way "Tajweed". Implementing neuroimaging methods in studying neural structure and function of some brain regions in professional readers of the Holy Qur'an and poets may extend our knowledge about neural basis of language processing and production at least in the Arabic language and may be the Semitic languages. Additionally, language production and processing in poets might be more appropriate to be used as investigation tools for neural bases of language comparing with musical perception and processing.

163

It has been reported that memory training (Belleville et al., 2011) and bilingualism (Bialystok et al., 2007, Craik et al., 2010, Schweizer et al., 2011) may delay onset of dementia including Alzheimer diseases. Studying brain structure and function of people with impressive memory ability and extensive language use e.g. poets could help improving neurorehabilitation or delaying strategies of dementia.

## References

- ABDUL-KAREEM, I. A. & SLUMING, V. 2008. Heschl gyrus and its included primary auditory cortex: structural MRI studies in healthy and diseased subjects. *J Magn Reson Imaging*, 28, 287-99.
- ABDUL-KAREEM, I. A., STANCAK, A., PARKES, L. M., AL-AMEEN, M., ALGHAMDI, J., ALDHAFEERI, F. M., EMBLETON, K., MORRIS, D. & SLUMING, V. 2011a. Plasticity of the Superior and Middle Cerebellar Peduncles in Musicians Revealed by Quantitative Analysis of Volume and Number of Streamlines Based on Diffusion Tensor Tractography. *Cerebellum*.
- ABDUL-KAREEM, I. A., STANCAK, A., PARKES, L. M. & SLUMING, V. 2011b. Increased gray matter volume of left pars opercularis in male orchestral musicians correlate positively with years of musical performance. J Magn Reson Imaging, 33, 24-32.
- ABRAMS, D. A., BHATARA, A., RYALI, S., BALABAN, E., LEVITIN, D. J. & MENON, V. 2011. Decoding temporal structure in music and speech relies on shared brain resources but elicits different fine-scale spatial patterns. *Cereb Cortex*, 21, 1507-18.
- ALTENMÜLLER, E. O. 2001. How Many Music Centers Are in the Brain? Ann NY Acad Sci, 930, 273-280.
- AMARO, E., JR., WILLIAMS, S. C., SHERGILL, S. S., FU, C. H., MACSWEENEY, M., PICCHIONI, M. M., BRAMMER, M. J. & MCGUIRE, P. K. 2002. Acoustic noise and functional magnetic resonance imaging: current strategies and future prospects. J Magn Reson Imaging, 16, 497-510.
- AMARO, E., JR. & BARKER, G. J. 2006. Study design in fMRI: basic principles. Brain Cogn, 60, 220-32.
- AMITAY, S., IRWIN, A. & MOORE, D. R. 2006. Discrimination learning induced by training with identical stimuli. *Nat Neurosci*, 9, 1446-8.
- AMUNTS, K., SCHLAUG, G., JANCKE, L., STEINMETZ, H., SCHLEICHER, A., DABRINGHAUS, A. & ZILLES, K. 1997. Motor cortex and hand motor skills: structural compliance in the human brain. *Hum Brain Mapp*, 5, 206-15.
- ANDERSEN, P. 2007. *The hippocampus book*, New York ; Oxford, Oxford University Press.

- ANDERSON, B. J., ECKBURG, P. B. & RELUCIO, K. I. 2002. Alterations in the thickness of motor cortical subregions after motor-skill learning and exercise. *Learn Mem*, 9, 1-9.
- ARTHURS, O., DONOVAN, T., SPIEGELHALTER, D., PICKARD, J. & BONIFACE, S. 2007. Intracortically Distributed Neurovascular Coupling Relationships within and between Human Somatosensory Cortices. *Cereb Cortex*, 17, 661-668.
- ARTHURS, O. J. & BONIFACE, S. 2002. How well do we understand the neural origins of the fMRI BOLD signal? *Trends in Neurosciences*, 25, 27-31.
- ARTHURS, O. J. & BONIFACE, S. J. 2003. What aspect of the fMRI BOLD signal best reflects the underlying electrophysiology in human somatosensory cortex? *Clinical neurophysiology : official journal of the International Federation of Clinical Neurophysiology*, 114, 1203-1209.
- ASA 1960. American standard acoustical terminology: (including mechanical shock and vibration), American Standards Association.
- ASHBURNER, J. & FRISTON, K. J. 2000. Voxel-based morphometry--the methods. *Neuroimage*, 11, 805-21.
- AYDIN, K., CIFTCI, K., TERZIBASIOGLU, E., OZKAN, M., DEMIRTAS, A., SENCER, S. & MINARECI, O. 2005. Quantitative proton MR spectroscopic findings of cortical reorganization in the auditory cortex of musicians. AJNR - American Journal of Neuroradiology, 26, 128-36.
- AYDIN, K., UCAR, A., OGUZ, K. K., OKUR, O. O., AGAYEV, A., UNAL, Z., YILMAZ, S. & OZTURK, C. 2007. Increased gray matter density in the parietal cortex of mathematicians: a voxel-based morphometry study. AJNR - American Journal of Neuroradiology, 28, 1859-64.
- BANDETTINI, P. A., JESMANOWICZ, A., VAN KYLEN, J., BIRN, R. M. & HYDE, J. S. 1998. Functional MRI of brain activation induced by scanner acoustic noise. *Magn Reson Med*, 39, 410-6.
- BANGERT, M., PESCHEL, T., SCHLAUG, G., ROTTE, M., DRESCHER, D., HINRICHS, H., HEINZE, H. J. & ALTENMULLER, E. 2006. Shared networks for auditory and motor processing in professional pianists: evidence from fMRI conjunction. *Neuroimage*, 30, 917-26.
- BANGERT, M. & SCHLAUG, G. 2006. Specialization of the specialized in features of external human brain morphology. *Eur J Neurosci*, 24, 1832-4.
- BARKER, D., PLACK, C. J. & HALL, D. A. 2011. Reexamining the Evidence <sup>°</sup> for a Pitch-Sensitive Region: A Human fMRI Study Using Iterated Ripple Noise. *Cereb Cortex*.

- BAUMANN, S., GRIFFITHS, T. D., REES, A., HUNTER, D., SUN, L. & THIELE, A. 2010. Characterisation of the BOLD response time course at different levels of the auditory pathway in non-human primates. *Neuroimage*, 50, 1099-108.
- BAZWINSKY, I., HILBIG, H., BIDMON, H. J. & RUBSAMEN, R. 2003. Characterization of the human superior olivary complex by calcium binding proteins and neurofilament H (SMI-32). J Comp Neurol, 456, 292-303.
- BELIN, P., ZATORRE, R. J., HOGE, R., EVANS, A. C. & PIKE, B. 1999. Event-related fMRI of the auditory cortex. *Neuroimage*, 10, 417-29.
- BELLEVILLE, S., CLÉMENT, F., MELLAH, S., GILBERT, B., FONTAINE, F. & GAUTHIER, S. 2011. Training-related brain plasticity in subjects at risk of developing Alzheimer's disease. *Brain*.
- BENDOR, D. & WANG, X. 2005. The neuronal representation of pitch in primate auditory cortex. *Nature*, 436, 1161-5.
- BENDOR, D. & WANG, X. 2008. Neural response properties of primary, rostral, and rostrotemporal core fields in the auditory cortex of marmoset monkeys. *J Neurophysiol*, 100, 888-906.
- BENDOR, D. 2011. Understanding how neural circuits measure pitch. *J Neurosci*, 31, 3141-2.
- BENGTSSON, S. L., NAGY, Z., SKARE, S., FORSMAN, L., FORSSBERG, H. & ULLEN, F. 2005. Extensive piano practicing has regionally specific effects on white matter development. *Nat Neurosci*, 8, 1148-50.
- BENJAMINI, Y. & HOCHBERG, Y. 1995. Controlling the False Discovery Rate: A Practical and Powerful Approach to Multiple Testing. *J R Stat Soc Ser B Stat Methodol*, 57, 289-300.
- BERMUDEZ, P., LERCH, J. P., EVANS, A. C. & ZATORRE, R. J. 2009. Neuroanatomical correlates of musicianship as revealed by cortical thickness and voxel-based morphometry. *Cereb Cortex*, 19, 1583-96.
- BERNSTEIN, M. A., KING, K. F. & ZHOU, Z. J. 2004. Handbook of MRI pulse sequences, Amsterdam; Boston, Academic Press.
- BIALYSTOK, E., CRAIK, F. I. M. & FREEDMAN, M. 2007. Bilingualism as a protection against the onset of symptoms of dementia. *Neurophysiology*, 45, 459-464.
- BIGAND, E. 2003. More About the Musical Expertise of Musically Untrained Listeners. Ann N Y Acad Sci, 999, 304-312.

- BIGAND, E. & POULIN-CHARRONNAT, B. 2006. Are we "experienced listeners"? A review of the musical capacities that do not depend on formal musical training. *Cognition*, 100, 100-130.
- BILECEN, D., SEIFRITZ, E., SCHEFFLER, K., HENNING, J. & SCHULTE, A. C. 2002. Amplitopicity of the human auditory cortex: an fMRI study. *Neuroimage*, 17, 710-8.
- BINDER, J. R., FROST, J. A., HAMMEKE, T. A., COX, R. W., RAO, S. M. & PRIETO, T. 1997. Human brain language areas identified by functional magnetic resonance imaging. *J Neurosci*, 17, 353-62.
- BLUMENFELD-KATZIR, T., PASTERNAK, O., DAGAN, M. & ASSAF, Y. 2011. Diffusion MRI of structural brain plasticity induced by a learning and memory task. *PLoS ONE*, 6, e20678.
- BOOKSTEIN, F. L. 2001. "Voxel-based morphometry" should not be used with imperfectly registered images. *Neuroimage*, 14, 1454-62.
- BOSNYAK, D. J., GANDER, P. E. & ROBERTS, L. E. 2007. Does auditory discrimination training modify representations in both primary and secondary auditory cortex? *International Congress Series*, 1300, 25-28.
- BOUDELAA, S., PULVERMULLER, F., HAUK, O., SHTYROV, Y. & MARSLEN-WILSON, W. 2010. Arabic morphology in the neural language system. *J Cogn Neurosci*, 22, 998-1010.
- BRATTICO, E., NÄÄTÄNEN, R. & TERVANIEMI, M. 2001. Context effects on pitch perception in musicians and non-musicians: Evidence from ERP recordings. *Music Perception*, 19, 199–222.
- BROWN, M., IRVINE, D. R. & PARK, V. N. 2004. Perceptual learning on an auditory frequency discrimination task by cats: association with changes in primary auditory cortex. *Cereb Cortex*, 14, 952-65.
- BROWN, S., MARTINEZ, M. J. & PARSONS, L. M. 2006. Music and language side by side in the brain: a PET study of the generation of melodies and sentences. *Eur J Neurosci*, 23, 2791-803.
- BRUN, C. C., LEPORE, N., PENNEC, X., LEE, A. D., BARYSHEVA, M., MADSEN, S. K., AVEDISSIAN, C., CHOU, Y. Y., DE ZUBICARAY, G. I., MCMAHON, K. L., WRIGHT, M. J., TOGA, A. W. & THOMPSON, P. M. 2009. Mapping the regional influence of genetics on brain structure variability--a tensor-based morphometry study. *Neuroimage*, 48, 37-49.

- BURNS, E. M. & HOUTSMA, A. J. 1999. The influence of musical training on the perception of sequentially presented mistuned harmonics. *J Acoust Soc Am*, 106, 3564-70.
- CARMELLI, D., SWAN, G. E., DECARLI, C. & REED, T. 2002. Quantitative genetic modeling of regional brain volumes and cognitive performance in older male twins. *Biol Psychol*, 61, 139-55.
- CATANI, M., ALLIN, M. P., HUSAIN, M., PUGLIESE, L., MESULAM, M. M., MURRAY, R. M. & JONES, D. K. 2007. Symmetries in human brain language pathways correlate with verbal recall. *Proc Natl Acad Sci U S A*, 104, 17163-8.
- CEDOLIN, L. & DELGUTTE, B. 2010. Spatiotemporal representation of the pitch of harmonic complex tones in the auditory nerve. *J Neurosci*, 30, 12712-24.
- CELSIS, P., BOULANOUAR, K., DOYON, B., RANJEVA, J. P., BERRY, I., NESPOULOUS, J. L. & CHOLLET, F. 1999. Differential fMRI responses in the left posterior superior temporal gyrus and left supramarginal gyrus to habituation and change detection in syllables and tones. *Neuroimage*, 9, 135-44.
- CHAN, A. S., HO, Y. C. & CHEUNG, M. C. 1998. Music training improves verbal memory. *Nature*, 396, 128.
- CHEVEIGNÉ, A. 2005. Pitch Perception Models. In: PLACK, C., FAY, R., OXENHAM, A. & POPPER, A. (eds.) Pitch. Springer New York.
- CRAIK, F. I., BIALYSTOK, E. & FREEDMAN, M. 2010. Delaying the onset of Alzheimer disease: bilingualism as a form of cognitive reserve. *Neurology*, 75, 1726-9.
- CSERNANSKY, J. G., GILLESPIE, S. K., DIERKER, D. L., ANTICEVIC, A., WANG, L., BARCH, D. M. & VAN ESSEN, D. C. 2008. Symmetric abnormalities in sulcal patterning in schizophrenia. *Neuroimage*, 43, 440-6.
- DA COSTA, S., VAN DER ZWAAG, W., MARQUES, J. P., FRACKOWIAK, R. S., CLARKE, S. & SAENZ, M. 2011. Human Primary Auditory Cortex Follows the Shape of Heschl's Gyrus. *J Neurosci*, 31, 14067-75.
- DALE, A. M. & BUCKNER, R. L. 1997. Selective averaging of rapidly presented individual trials using fMRI. *Hum Brain Mapp*, 5, 329-40.
- DALE, A. M., LIU, A. K., FISCHL, B. R., BUCKNER, R. L., BELLIVEAU, J. W., LEWINE, J. D. & HALGREN, E. 2000. Dynamic statistical

parametric mapping: combining fMRI and MEG for high-resolution imaging of cortical activity. *Neuron*, 26, 55-67.

- DALE, A. M. & HALGREN, E. 2001. Spatiotemporal mapping of brain activity by integration of multiple imaging modalities. *Current Opinion in Neurobiology*, 11, 202-8.
- DAN, Y. & POO, M. M. 2006. Spike timing-dependent plasticity: from synapse to perception. *Physiol Rev*, 86, 1033-48.
- DAVIES, G., TENESA, A., PAYTON, A., YANG, J., HARRIS, S. E., LIEWALD, D., KE, X., LE HELLARD, S., CHRISTOFOROU, A., LUCIANO, M., MCGHEE, K., LOPEZ, L., GOW, A. J., CORLEY, J., REDMOND, P., FOX, H. C., HAGGARTY, P., WHALLEY, L. J., MCNEILL, G., GODDARD, M. E., ESPESETH, T., LUNDERVOLD, A. J., REINVANG, I., PICKLES, A., STEEN, V. M., OLLIER, W., PORTEOUS, D. J., HORAN, M., STARR, J. M., PENDLETON, N., VISSCHER, P. M. & DEARY, I. J. 2011. Genome-wide association studies establish that human intelligence is highly heritable and polygenic. *Mol Psychiatry*, 16, 996-1005.
- DEARY, I. J., PENKE, L. & JOHNSON, W. 2010. The neuroscience of human intelligence differences. *Nat Rev Neurosci*, 11, 201-11.
- DESAI, R., LIEBENTHAL, E., POSSING, E. T., WALDRON, E. & BINDER, J. R. 2005. Volumetric vs. surface-based alignment for localization of auditory cortex activation. *Neuroimage*, 26, 1019-29.
- DESMOND, J. E. & GLOVER, G. H. 2002. Estimating sample size in functional MRI (fMRI) neuroimaging studies: statistical power analyses. *J Neurosci Methods*, 118, 115-28.

DEUTSCH, D. 1982. The Psychology of music, Academic Press.

DIETRICH, V., NIESCHALK, M., STOLL, W., RAJAN, R. & PANTEV, C. 2001. Cortical reorganization in patients with high frequency cochlear hearing loss. *Hearing Research*, 158, 95-101.

DUVERNOY, H. M. 2005. The human hippocampus : functional anatomy, vascularization, and serial sections with MRI, Berlin Berlin; New York, Springer.

- ECONOMO, C. & TRIARHOU, L. C. 2009. Cellular structure of the human cerebral cortex, Basel; New York, Karger.
- EDELSTEIN, W. A., HEDEEN, R. A., MALLOZZI, R. P., EL-HAMAMSY, S. A., ACKERMANN, R. A. & HAVENS, T. J. 2002. Making MRI quieter. *Magn Reson Imaging*, 20, 155-63.

- EDELSTEIN, W. A., KIDANE, T. K., TARACILA, V., BAIG, T. N., EAGAN, T. P., CHENG, Y. C., BROWN, R. W. & MALLICK, J. A. 2005. Activepassive gradient shielding for MRI acoustic noise reduction. *Magn Reson Med*, 53, 1013-7.
- EDEN, G. F., JOSEPH, J. E., BROWN, H. E., BROWN, C. P. & ZEFFIRO, T. A. 1999. Utilizing hemodynamic delay and dispersion to detect fMRI signal change without auditory interference: the behavior interleaved gradients technique. *Magn Reson Med*, 41, 13-20.
- EGGERMONT, J. J. & ROBERTS, L. E. 2004. The neuroscience of tinnitus. *Trends in Neurosciences*, 27, 676-82.
- EGGERMONT, J. J. 2005. Plasticity of Tonotopic and Correlation Maps in Cat Primary Auditory Cortex. In: SYKA, J. & MERZENICH, M. M. (eds.) Plasticity and Signal Representation in the Auditory System. Springer US.
- EICKHOFF, S. B., STEPHAN, K. E., MOHLBERG, H., GREFKES, C., FINK, G. R., AMUNTS, K. & ZILLES, K. 2005. A new SPM toolbox for combining probabilistic cytoarchitectonic maps and functional imaging data. *Neuroimage*, 25, 1325-1335.
- ELBERT, T., PANTEV, C., WIENBRUCH, C., ROCKSTROH, B. & TAUB, E. 1995. Increased Cortical Representation of the Fingers of the Left Hand in String Players. *Science*, 270, 305-307.
- ELBERT, T., STERR, A., ROCKSTROH, B., PANTEV, C., MULLER, M. M. & TAUB, E. 2002. Expansion of the tonotopic area in the auditory cortex of the blind. *J Neurosci*, 22, 9941-4.
- FAUL, F., ERDFELDER, E., LANG, A. G. & BUCHNER, A. 2007. G\*Power 3: a flexible statistical power analysis program for the social, behavioral, and biomedical sciences. *Behavior Research Methods*, 39, 175-91.
- FAUL, F., ERDFELDER, E., BUCHNER, A. & LANG, A. G. 2009. Statistical power analyses using G\*Power 3.1: tests for correlation and regression analyses. *Behavior Research Methods*, 41, 1149-60.
- FEDORENKO, E., PATEL, A., CASASANTO, D., WINAWER, J. & GIBSON, E. 2009. Structural integration in language and music: evidence for a shared system. *Mem Cognit*, 37, 1-9.
- FISCHL, B., SERENO, M. I. & DALE, A. M. 1999. Cortical surface-based analysis. II: Inflation, flattening, and a surface-based coordinate system. *Neuroimage*, 9, 195-207.
- FISCHL, B., SALAT, D. H., BUSA, E., ALBERT, M., DIETERICH, M., HASELGROVE, C., VAN DER KOUWE, A., KILLIANY, R.,

KENNEDY, D., KLAVENESS, S., MONTILLO, A., MAKRIS, N., ROSEN, B. & DALE, A. M. 2002. Whole brain segmentation: automated labeling of neuroanatomical structures in the human brain. *Neuron*, 33, 341-55.

- FLOR, H., HOFFMANN, D., STRUVE, M. & DIESCH, E. 2004. Auditory discrimination training for the treatment of tinnitus. *Appl Psychophysiol Biofeedback*, 29, 113-20.
- FORGEARD, M., WINNER, E., NORTON, A. & SCHLAUG, G. 2008. Practicing a musical instrument in childhood is associated with enhanced verbal ability and nonverbal reasoning. *PLoS ONE*, 3, e3566.
- FORMISANO, E., LINDEN, D. E., DI SALLE, F., TROJANO, L., ESPOSITO, F., SACK, A. T., GROSSI, D., ZANELLA, F. E. & GOEBEL, R. 2002. Tracking the mind's image in the brain I: time-resolved fMRI during visuospatial mental imagery. *Neuron*, 35, 185-94.
- FORMISANO, E., KIM, D. S., DI SALLE, F., VAN DE MOORTELE, P. F., UGURBIL, K. & GOEBEL, R. 2003. Mirror-symmetric tonotopic maps in human primary auditory cortex. *Neuron*, 40, 859-69.
- FROEMKE, R. C. & DAN, Y. 2002. Spike-timing-dependent synaptic modification induced by natural spike trains. *Nature*, 416, 433-8.
- FUJIOKA, T., TRAINOR, L. J., ROSS, B., KAKIGI, R. & PANTEV, C. 2004. Musical training enhances automatic encoding of melodic contour and interval structure. J Cogn Neurosci, 16, 1010-21.
- FUJIOKA, T., TRAINOR, L. J., ROSS, B., KAKIGI, R. & PANTEV, C. 2005. Automatic encoding of polyphonic melodies in musicians and nonmusicians. J Cogn Neurosci, 17, 1578-92.
- FUJIOKA, T., ROSS, B., KAKIGI, R., PANTEV, C. & TRAINOR, L. J. 2006. One year of musical training affects development of auditory corticalevoked fields in young children. *Brain*, 129, 2593-608.
- GAAB, N., GASER, C., ZAEHLE, T., JANCKE, L. & SCHLAUG, G. 2003. Functional anatomy of pitch memory--an fMRI study with sparse temporal sampling. *Neuroimage*, 19, 1417-26.
- GAAB, N. & SCHLAUG, G. 2003. The effect of musicianship on pitch memory in performance matched groups. *Neuroreport*, 14, 2291-5.
- GAAB, N., GASER, C. & SCHLAUG, G. 2006. Improvement-related functional plasticity following pitch memory training. *Neuroimage*, 31, 255-63.

- GALAZYUK, A. V. 1990. Tonotopic organization of the ventrorostral zone of the auditory cortex region AII in cats. *Neurophysiology*, 22, 139-144.
- GASER, C. & SCHLAUG, G. 2003. Brain structures differ between musicians and non-musicians. *J Neurosci*, 23, 9240-5.
- GENOVESE, C. R., LAZAR, N. A. & NICHOLS, T. 2002. Thresholding of statistical maps in functional neuroimaging using the false discovery rate. *Neuroimage*, 15, 870-8.
- GEUZE, E., WESTENBERG, H. G. M., HEINECKE, A., DE KLOET, C. S., GOEBEL, R. & VERMETTEN, E. 2008. Thinner prefrontal cortex in veterans with posttraumatic stress disorder. *Neuroimage*, 41, 675-681.
- GOEBEL, R., ESPOSITO, F. & FORMISANO, E. 2006. Analysis of functional image analysis contest (FIAC) data with brainvoyager QX: From singlesubject to cortically aligned group general linear model analysis and selforganizing group independent component analysis. *Hum Brain Mapp*, 27, 392-401.
- GOENSE, J. B. & LOGOTHETIS, N. K. 2008. Neurophysiology of the BOLD fMRI signal in awake monkeys. *Curr Biol*, 18, 631-40.
- GOLDSTEIN, E. B., HUMPHREYS, G. W., SHIFFRAR, M. & YOST, W. A. 2004. Blackwell handbook of sensation and perception, Blackwell Pub.
- GOLDSTEIN, E. B., HUMPHREYS, G. W., SHIFFRAR, M. & YOST, W. A. 2005. *Blackwell handbook of sensation and perception*, Oxford, UK ; Malden, MA, Blackwell Pub.
- GOLDSTEIN, J. L., GERSON, A., SRULOVICZ, P. & FURST, M. 1978. Verification of the optimal probabilistic basis of aural processing in pitch of complex tones. J Acoust Soc Am, 63, 486-497.
- GRAZIANO, A. B., PETERSON, M. & SHAW, G. L. 1999. Enhanced learning of proportional math through music training and spatial-temporal training. *Neurol Res*, 21, 139-52.
- GRIFFITHS, T. D., BUCHEL, C., FRACKOWIAK, R. S. & PATTERSON, R. D. 1998a. Analysis of temporal structure in sound by the human brain. *Nat Neurosci*, 1, 422-7.
- GRIFFITHS, T. D., REES, G., REES, A., GREEN, G. G., WITTON, C., ROWE, D., BUCHEL, C., TURNER, R. & FRACKOWIAK, R. S. 1998b. Right parietal cortex is involved in the perception of sound movement in humans. *Nat Neurosci*, 1, 74-9.

- GRIFFITHS, T. D., UPPENKAMP, S., JOHNSRUDE, I., JOSEPHS, O. & PATTERSON, R. D. 2001. Encoding of the temporal regularity of sound in the human brainstem. *Nat Neurosci*, 4, 633-7.
- GRIFFITHS, T. D. 2003. Functional Imaging of Pitch Analysis. Ann N Y Acad Sci, 999, 40-49.
- GROUSSARD, M., LA JOIE, R., RAUCHS, G., LANDEAU, B., CHETELAT, G., VIADER, F., DESGRANGES, B., EUSTACHE, F. & PLATEL, H. 2010a. When music and long-term memory interact: effects of musical expertise on functional and structural plasticity in the hippocampus. *PLoS ONE*, 5.
- GROUSSARD, M., VIADER, F., HUBERT, V., LANDEAU, B., ABBAS, A., DESGRANGES, B., EUSTACHE, F. & PLATEL, H. 2010b. Musical and verbal semantic memory: two distinct neural networks? *Neuroimage*, 49, 2764-73.
- GULICK, W. L. 1971. *Hearing: physiology and psychophysics*, Oxford University Press.
- HACKETT, T. A., STEPNIEWSKA, I. & KAAS, J. H. 1998. Subdivisions of auditory cortex and ipsilateral cortical connections of the parabelt auditory cortex in macaque monkeys. *The Journal of comparative neurology*, 394, 475-95.
- HACKETT, T. A., PREUSS, T. M. & KAAS, J. H. 2001. Architectonic identification of the core region in auditory cortex of macaques, chimpanzees, and humans. *The Journal of comparative neurology*, 441, 197-222.
- HALL, D. A., HAGGARD, M. P., AKEROYD, M. A., PALMER, A. R.,
  SUMMERFIELD, A. Q., ELLIOTT, M. R., GURNEY, E. M. & BOWTELL, R. W. 1999. "Sparse" temporal sampling in auditory fMRI. *Hum Brain Mapp*, 7, 213-23.
- HALL, D. A., SUMMERFIELD, A. Q., GONCALVES, M. S., FOSTER, J. R., PALMER, A. R. & BOWTELL, R. W. 2000. Time-course of the auditory BOLD response to scanner noise. *Magn Reson Med*, 43, 601-6.
- HALL, D. A., HART, H. C. & JOHNSRUDE, I. S. 2003. Relationships between human auditory cortical structure and function. *Audiol Neurootol*, 8, 1-18.
- HALL, D. A. & PLACK, C. J. 2009. Pitch processing sites in the human auditory brain. *Cereb Cortex*, 19, 576-85.
- HALLER, S., BORGWARDT, S. J., SCHINDLER, C., ASTON, J., RADUE, E. W. & RIECHER-RöSSLER, A. 2009. Can Cortical Thickness

Asymmetry Analysis Contribute to Detection of At-Risk Mental State and First-Episode Psychosis?: A Pilot Study1. *Radiology*, 250, 212-221.

- HALWANI, G. F., LOUI, P., RUEBER, T. & SCHLAUG, G. 2011. Effects of practice and experience on the arcuate fasciculus: comparing singers, instrumentalists, and non-musicians. *Front Psychol*, 2.
- HÄMÄLÄINEN, M., HARI, R., ILMONIEMI, R. J., KNUUTILA, J. & LOUNASMAA, O. V. 1993. Magnetoencephalography -- theory, instrumentation, and applications to noninvasive studies of the working human brain. *Reviews of Modern Physics*, 65, 413.
- HAN, Y., YANG, H., LV, Y.-T., ZHU, C.-Z., HE, Y., TANG, H.-H., GONG, Q.-Y., LUO, Y.-J., ZANG, Y.-F. & DONG, Q. 2009. Gray matter density and white matter integrity in pianists' brain: A combined structural and diffusion tensor MRI study. *Neurosci Lett*, 459, 3-6.
- HANSEN, P. C., KRINGELBACH, M. L. & SALMELIN, R. 2010. *MEG* : an *introduction to methods*, New York, Oxford University Press.
- HARRISON, R. V., STANTON, S. G., NAGASAWA, A., IBRAHIM, D. & MOUNT, R. J. 1993. The effects of long-term cochlear hearing loss on the functional organization of central auditory pathways. *J Otolaryngol*, 22, 4-11.
- HAYASAKA, S., PEIFFER, A. M., HUGENSCHMIDT, C. E. & LAURIENTI, P. J. 2007. Power and sample size calculation for neuroimaging studies by non-central random field theory. *Neuroimage*, 37, 721-30.
- HE, B. & LIU, Z. 2008. Multimodal Functional Neuroimaging: Integrating Functional MRI and EEG/MEG. *IEEE reviews in biomedical engineering*, 1, 23-40.
- HEDEEN, R. A. & EDELSTEIN, W. A. 1997. Characterization and prediction of gradient acoustic noise in MR imagers. *Magn Reson Med*, 37, 7-10.
- HELMHOLTZ, H. L. F. & ELLIS, A. J. 1954. On the sensations of tone as a physiological basis for the theory of music, NY, [s.n.].
- HERDENER, M., ESPOSITO, F., DI SALLE, F., BOLLER, C., HILTI, C. C., HABERMEYER, B., SCHEFFLER, K., WETZEL, S., SEIFRITZ, E. & CATTAPAN-LUDEWIG, K. 2010. Musical training induces functional plasticity in human hippocampus. *J Neurosci*, 30, 1377-84.
- HERRAIZ, C., DIGES, I. & COBO, P. 2007. Auditory discrimination therapy (ADT) for tinnitus management. *Progress in Brain Research*, 166, 467-71.

- HERRAIZ, C., DIGES, I., COBO, P. & APARICIO, J. M. 2009. Cortical reorganisation and tinnitus: principles of auditory discrimination training for tinnitus management. *Eur Arch Otorhinolaryngol*, 266, 9-16.
- HO, Y. C., CHEUNG, M. C. & CHAN, A. S. 2003. Music training improves verbal but not visual memory: cross-sectional and longitudinal explorations in children. *Neuropsychology*, 17, 439-50.
- HOENIG, K., MULLER, C., HERRNBERGER, B., SIM, E. J., SPITZER, M., EHRET, G. & KIEFER, M. 2011. Neuroplasticity of semantic representations for musical instruments in professional musicians. *Neuroimage*, 56, 1714-25.
- HOWARD, M. A., 3RD, VOLKOV, I. O., ABBAS, P. J., DAMASIO, H., OLLENDIECK, M. C. & GRANNER, M. A. 1996. A chronic microelectrode investigation of the tonotopic organization of human auditory cortex. *Brain Res*, 724, 260-4.
- HUMPHRIES, C., LIEBENTHAL, E. & BINDER, J. R. 2010. Tonotopic organization of human auditory cortex. *Neuroimage*, 50, 1202-11.
- HUTCHINSON, S., LEE, L. H., GAAB, N. & SCHLAUG, G. 2003. Cerebellar volume of musicians. *Cereb Cortex*, 13, 943-9.
- HYDE, K. L., LERCH, J., NORTON, A., FORGEARD, M., WINNER, E., EVANS, A. C. & SCHLAUG, G. 2009. Musical training shapes structural brain development. J Neurosci, 29, 3019-25.
- IBRAHIM, R. & EVIATAR, Z. 2009. Language status and hemispheric involvement in reading: evidence from trilingual Arabic speakers tested in Arabic, Hebrew, and English. *Neuropsychology*, 23, 240-54.
- IM, C.-H., JUNG, H.-K. & FUJIMAKI, N. 2005. fMRI-constrained MEG source imaging and consideration of fMRI invisible sources. *Hum Brain Mapp*, 26, 110-118.
- JAHANSHAD, N., LEE, A. D., BARYSHEVA, M., MCMAHON, K. L., DE ZUBICARAY, G. I., MARTIN, N. G., WRIGHT, M. J., TOGA, A. W. & THOMPSON, P. M. 2010. Genetic influences on brain asymmetry: a DTI study of 374 twins and siblings. *Neuroimage*, 52, 455-69.
- JÄNCKE, L., GAAB, N., WUSTENBERG, T., SCHEICH, H. & HEINZE, H. J. 2001. Short-term functional plasticity in the human auditory cortex: an fMRI study. *Brain Res Cogn Brain Res*, 12, 479-85.

JENKINS, J. S. 2001. The Mozart effect. J R Soc Med, 94, 170-2.

- JENKINSON, M. & SMITH, S. 2001. A global optimisation method for robust affine registration of brain images. *Med Image Anal*, 5, 143-56.
- JONES, M. R., FAY, R. R. & POPPER, A. N. 2010. *Music perception*, New York, Springer.
- JONES, S. E., BUCHBINDER, B. R. & AHARON, I. 2000. Three-dimensional mapping of cortical thickness using Laplace's equation. *Hum Brain Mapp*, 11, 12-32.
- JOSEPHS, O., TURNER, R. & FRISTON, K. 1997. Event-related f MRI. Hum Brain Mapp, 5, 243-8.
- KAAS, J. H. & HACKETT, T. A. 2000. Subdivisions of auditory cortex and processing streams in primates. *Proc Natl Acad Sci U S A*, 97, 11793-9.
- KALATSKY, V. A., POLLEY, D. B., MERZENICH, M. M., SCHREINER, C. E.
   & STRYKER, M. P. 2005. Fine functional organization of auditory cortex revealed by Fourier optical imaging. *Proc Natl Acad Sci U S A*, 102, 13325-30.
- KATSUNUMA, A., TAKAMORI, H., SAKAKURA, Y., HAMAMURA, Y., OGO, Y. & KATAYAMA, R. 2002. Quiet MRI with novel acoustic noise reduction. *MAGMA*, 13, 139-44.
- KENET, T., FROEMKE, R. C., SCHREINER, C. E., PESSAH, I. N. & MERZENICH, M. M. 2007. Perinatal exposure to a noncoplanar polychlorinated biphenyl alters tonotopy, receptive fields, and plasticity in rat primary auditory cortex. *Proc Natl Acad Sci US A*, 104, 7646-51.
- KISHON-RABIN, L., AMIR, O., VEXLER, Y. & ZALTZ, Y. 2001. Pitch Discrimination: Are Professional Musicians Better than Non-Musicians? J Basic Clin Physiol Pharmacol, 12, 125-144.
- KLEIM, J. A., LUSSNIG, E., SCHWARZ, E. R., COMERY, T. A. & GREENOUGH, W. T. 1996. Synaptogenesis and Fos expression in the motor cortex of the adult rat after motor skill learning. *J Neurosci*, 16, 4529-35.
- KOCHUNOV, P., GLAHN, D. C., NICHOLS, T. E., WINKLER, A. M., HONG,
  E. L., HOLCOMB, H. H., STEIN, J. L., THOMPSON, P. M., CURRAN,
  J. E., CARLESS, M. A., OLVERA, R. L., JOHNSON, M. P., COLE, S.
  A., KOCHUNOV, V., KENT, J. & BLANGERO, J. 2011. Genetic analysis of cortical thickness and fractional anisotropy of water diffusion in the brain. *Front Neurosci*, 5, 120.
- KOELSCH, S., SCHROGER, E. & TERVANIEMI, M. 1999. Superior preattentive auditory processing in musicians. *Neuroreport*, 10, 1309-13.

- KOELSCH, S., GUNTER, T. C., V. CRAMON, D. Y., ZYSSET, S., LOHMANN, G. & FRIEDERICI, A. D. 2002a. Bach Speaks: A Cortical "Language-Network" Serves the Processing of Music. *Neuroimage*, 17, 956-966.
- KOELSCH, S., SCHMIDT, B.-H. & KANSOK, J. 2002b. Effects of musical expertise on the early right anterior negativity: An event-related brain potential study. *Psychophysiology*, 39, 657-663.
- KOELSCH, S., FRITZ, T., SCHULZE, K., ALSOP, D. & SCHLAUG, G. 2005. Adults and children processing music: an fMRI study. *Neuroimage*, 25, 1068-76.
- KOSAKI, H., HASHIKAWA, T., HE, J. & JONES, E. G. 1997. Tonotopic organization of auditory cortical fields delineated by parvalbumin immunoreactivity in macaque monkeys. *J Comp Neurol*, 386, 304-16.
- KRIEGESKORTE, N. & GOEBEL, R. 2001. An efficient algorithm for topologically correct segmentation of the cortical sheet in anatomical mr volumes. *Neuroimage*, 14, 329-46.
- KRUMBHOLZ, K., PATTERSON, R. D., SEITHER-PREISLER, A., LAMMERTMANN, C. & LUTKENHONER, B. 2003. Neuromagnetic evidence for a pitch processing center in Heschl's gyrus. *Cereb Cortex*, 13, 765-72.
- KUSMIEREK, P. & RAUSCHECKER, J. P. 2009. Functional specialization of medial auditory belt cortex in the alert rhesus monkey. *J Neurophysiol*, 102, 1606-22.
- LANGHEIM, F. J., CALLICOTT, J. H., MATTAY, V. S., DUYN, J. H. & WEINBERGER, D. R. 2002. Cortical systems associated with covert music rehearsal. *Neuroimage*, 16, 901-8.
- LAPPE, C., HERHOLZ, S. C., TRAINOR, L. J. & PANTEV, C. 2008. Cortical plasticity induced by short-term unimodal and multimodal musical training. *J Neurosci*, 28, 9632-9.
- LAUTER, J. L., HERSCOVITCH, P., FORMBY, C. & RAICHLE, M. E. 1985. Tonotopic organization in human auditory cortex revealed by positron emission tomography. *Hearing Research*, 20, 199-205.
- LAZAR, S. W., KERR, C. E., WASSERMAN, R. H., GRAY, J. R., GREVE, D. N., TREADWAY, M. T., MCGARVEY, M., QUINN, B. T., DUSEK, J.

A., BENSON, H., RAUCH, S. L., MOORE, C. I. & FISCHL, B. 2005. Meditation experience is associated with increased cortical thickness. *Neuroreport*, 16, 1893-1897.

- LE BIHAN, D., URAYAMA, S., ASO, T., HANAKAWA, T. & FUKUYAMA, H. 2006. Direct and fast detection of neuronal activation in the human brain with diffusion MRI. *Proc Natl Acad Sci U S A*, 103, 8263-8.
- LE, T. H., PATEL, S. & ROBERTS, T. P. 2001. Functional MRI of human auditory cortex using block and event-related designs. *Magn Reson Med*, 45, 254-60.
- LEE, D. J., CHEN, Y. & SCHLAUG, G. 2003. Corpus callosum: musician and gender effects. *Neuroreport*, 14, 205-9.
- LENROOT, R. K., SCHMITT, J. E., ORDAZ, S. J., WALLACE, G. L., NEALE, M. C., LERCH, J. P., KENDLER, K. S., EVANS, A. C. & GIEDD, J. N. 2009. Differences in genetic and environmental influences on the human cerebral cortex associated with development during childhood and adolescence. *Hum Brain Mapp*, 30, 163-74.
- LI, S., SHI, F., PU, F., LI, X., JIANG, T., XIE, S. & WANG, Y. 2007. Hippocampal shape analysis of Alzheimer disease based on machine learning methods. AJNR - American Journal of Neuroradiology, 28, 1339-45.
- LIEGEOIS-CHAUVEL, C., MUSOLINO, A. & CHAUVEL, P. 1991. Localization of the primary auditory area in man. *Brain*, 114 (Pt 1A), 139-51.
- LIEGEOIS-CHAUVEL, C., MUSOLINO, A., BADIER, J. M., MARQUIS, P. & CHAUVEL, P. 1994. Evoked potentials recorded from the auditory cortex
- in man: evaluation and topography of the middle latency components. *Electroencephalogr Clin Neurophysiol*, 92, 204-14.
- LOENNEKER, T., HENNEL, F., LUDWIG, U. & HENNIG, J. 2001. Silent BOLD imaging. *MAGMA*, 13, 76-81.
- LOGOTHETIS, N. K., PAULS, J., AUGATH, M., TRINATH, T. & OELTERMANN, A. 2001. Neurophysiological investigation of the basis of the fMRI signal. *Nature*, 412, 150-7.
- LOPES DA SILVA, F. H. & ARNOLDS, D. E. 1978. Physiology of the hippocampus and related structures. *Annu Rev Physiol*, 40, 185-216.
- LOTZE, M., SCHELER, G., TAN, H. R., BRAUN, C. & BIRBAUMER, N. 2003. The musician's brain: functional imaging of amateurs and professionals during performance and imagery. *Neuroimage*, 20, 1817-29.

- LOUI, P., LI, H. C., HOHMANN, A. & SCHLAUG, G. 2011. Enhanced cortical connectivity in absolute pitch musicians: a model for local hyperconnectivity. *J Cogn Neurosci*, 23, 1015-26.
- LU, Z.-L. & KAUFMAN, L. 2003. *Magnetic source imaging of the human brain,* Mahwah, N.J.; London, Lawrence Erlbaum.
- LUDERS, E., GASER, C., JANCKE, L. & SCHLAUG, G. 2004. A voxel-based approach to gray matter asymmetries. *Neuroimage*, 22, 656-64.
- LÜTKENHÖNER, B. & STEINSTRATER, O. 1998. High-precision neuromagnetic study of the functional organization of the human auditory cortex. *Audiol Neurootol*, 3, 191-213.
- LÜTKENHÖNER, B., KRUMBHOLZ, K., LAMMERTMANN, C., SEITHER-PREISLER, A., STEINSTRATER, O. & PATTERSON, R. D. 2003a. Localization of primary auditory cortex in humans by magnetoencephalography. *Neuroimage*, 18, 58-66.
- LÜTKENHÖNER, B., KRUMBHOLZ, K. & SEITHER-PREISLER, A. 2003b. Studies of tonotopy based on wave N100 of the auditory evoked field are problematic. *Neuroimage*, 19, 935-49.
- MAESS, B., KOELSCH, S., GUNTER, T. C. & FRIEDERICI, A. D. 2001. Musical syntax is processed in Broca's area: an MEG study. *Nat Neurosci*, 4, 540-5.
- MAGUIRE, E. A., GADIAN, D. G., JOHNSRUDE, I. S., GOOD, C. D., ASHBURNER, J., FRACKOWIAK, R. S. & FRITH, C. D. 2000. Navigation-related structural change in the hippocampi of taxi drivers. *Proc Natl Acad Sci U S A*, 97, 4398-403.
- MALMIVUO, J. & PLONSEY, R. 1995. Bioelectromagnetism : principles and applications of bioelectric and biomagnetic fields, New York ; Oxford, Oxford University Press.
- MANSFIELD, P., GLOVER, P. M. & BEAUMONT, J. 1998. Sound generation in gradient coil structures for MRI. *Magn Reson Med*, 39, 539-50.
- MATHYS, C., LOUI, P., ZHENG, X. & SCHLAUG, G. 2010. Non-invasive brain stimulation applied to Heschl's gyrus modulates pitch discrimination. *Front Psychol*, 1, 193.
- MCDERMOTT, J. H. & OXENHAM, A. J. 2008. Music perception, pitch, and the auditory system. *Current Opinion in Neurobiology*, 18, 452-63.
- MCMULLEN, N. T. & GLASER, E. M. 1982. Tonotopic organization of rabbit auditory cortex. *Exp Neurol*, 75, 208-20.

- MCNAMEE, R. L. & LAZAR, N. A. 2004. Assessing the sensitivity of fMRI group maps. *Neuroimage*, 22, 920-31.
- MCROBBIE, D. W. 2003. MRI from picture to proton, Cambridge, Cambridge University Press.
- MECHELLI, A., CRINION, J. T., NOPPENEY, U., O'DOHERTY, J., ASHBURNER, J., FRACKOWIAK, R. S. & PRICE, C. J. 2004. Neurolinguistics: structural plasticity in the bilingual brain. *Nature*, 431, 757.
- MECHELLI, A., PRICE, C. J., FRISTON, K. J. & ASHBURNER, J. 2005. Voxel-Based Morphometry of the Human Brain: Methods and Applications. *Current Medical Imaging Reviews*, 1, 105-113.
- MEISTER, I. G., KRINGS, T., FOLTYS, H., BOROOJERDI, B., MULLER, M., TOPPER, R. & THRON, A. 2004. Playing piano in the mind--an fMRI study on music imagery and performance in pianists. *Brain Res Cogn Brain Res*, 19, 219-28.
- MENNING, H., ROBERTS, L. E. & PANTEV, C. 2000. Plastic changes in the auditory cortex induced by intensive frequency discrimination training. *Neuroreport*, 11, 817-22.
- MERZENICH, M. M. & BRUGGE, J. F. 1973. Representation of the cochlear partition of the superior temporal plane of the macaque monkey. *Brain Res*, 50, 275-96.
- MERZENICH, M. M., KNIGHT, P. L. & ROTH, G. L. 1975. Representation of cochlea within primary auditory cortex in the cat. *J Neurophysiol*, 38, 231-249.
- MICHEYL, C., DELHOMMEAU, K., PERROT, X. & OXENHAM, A. J. 2006. Influence of musical and psychoacoustical training on pitch discrimination. *Hear Res*, 219, 36-47.
- MOELKER, A. & PATTYNAMA, P. M. 2003. Acoustic noise concerns in functional magnetic resonance imaging. *Hum Brain Mapp*, 20, 123-41.
- MONAGHAN, P., METCALFE, N. B. & RUXTON, G. D. 1998. Does practice shape the brain? *Nature*, 394, 434.
- MOORE, B. C. 1973. Frequency difference limens for short-duration tones. J Acoust Soc Am, 54, 610-9.
- MOORE, B. C. J., GLASBERG, B. R. & PROCTOR, G. M. 1992. Accuracy of pitch matching for pure tones and for complex tones with overlapping or nonoverlapping harmonics. *J Acoust Soc Am*, 91, 3443-3450.

- MOORE, B. C. J. 2003. An introduction to the psychology of hearing, Amsterdam ; London, Academic Press.
- MORENO, S., BIALYSTOK, E., BARAC, R., SCHELLENBERG, E. G., CEPEDA, N. J. & CHAU, T. 2011. Short-term music training enhances verbal intelligence and executive function. *Psychol Sci*, 22, 1425-33.
- MOREY, R. A., PETTY, C. M., XU, Y., PANNU HAYES, J., WAGNER II, H. R., LEWIS, D. V., LABAR, K. S., STYNER, M. & MCCARTHY, G. 2009. A comparison of automated segmentation and manual tracing for quantifying hippocampal and amygdala volumes. *Neuroimage*, 45, 855-866.
- MOROSAN, P., RADEMACHER, J., SCHLEICHER, A., AMUNTS, K., SCHORMANN, T. & ZILLES, K. 2001. Human primary auditory cortex: cytoarchitectonic subdivisions and mapping into a spatial reference system. *Neuroimage*, 13, 684-701.
- MUHLNICKEL, W., ELBERT, T., TAUB, E. & FLOR, H. 1998. Reorganization of auditory cortex in tinnitus. *Proc Natl Acad Sci U S A*, 95, 10340-3.
- MUNRO, K. J. 2008. Reorganization of the adult auditory system: perceptual and physiological evidence from monaural fitting of hearing AIDS. *Trends Amplif*, 12, 85-102.
- MUNTE, T. F., ALTENMULLER, E. & JANCKE, L. 2002. The musician's brain as a model of neuroplasticity. *Nat Rev Neurosci*, 3, 473-8.
- MUSACCHIA, G., SAMS, M., SKOE, E. & KRAUS, N. 2007. Musicians have enhanced subcortical auditory and audiovisual processing of speech and music. *Proceedings of the National Academy of Sciences*, 104, 15894-15898.
- NAATANEN, R. & WINKLER, I. 1999. The concept of auditory stimulus representation in cognitive neuroscience. *Psychological Bulletin*, 125, 826-59.
- NAKAHARA, H., ZHANG, L. I. & MERZENICH, M. M. 2004. Specialization of primary auditory cortex processing by sound exposure in the "critical period". *Proc Natl Acad Sci U S A*, 101, 7170-4.
- NIEDERMEYER, E. & SILVA, F. H. L. 2005. *Electroencephalography: basic principles, clinical applications, and related fields*, Lippincott Williams & Wilkins.
- NIKJEH, D. A., LISTER, J. J. & FRISCH, S. A. 2009. Preattentive corticalevoked responses to pure tones, harmonic tones, and speech: influence of music training. *Ear and Hearing*, 30, 432-46.

- NORDMARK, J. O. 1968. Mechanisms of frequency discrimination. J Acoust Soc Am, 44, 1533-40.
- O'MARA, S. M., SANCHEZ-VIVES, M. V., BROTONS-MAS, J. R. & O'HARE, E. 2009. Roles for the subiculum in spatial information processing, memory, motivation and the temporal control of behaviour. *Progress in Neuro-Psychopharmacology and Biological Psychiatry*, 33, 782-90.
- OHNISHI, T., MATSUDA, H., ASADA, T., ARUGA, M., HIRAKATA, M., NISHIKAWA, M., KATOH, A. & IMABAYASHI, E. 2001. Functional anatomy of musical perception in musicians. *Cereb Cortex*, 11, 754-60.
- ONO, Y. & ISHIYAMA, A. 2008. Magnetoencephalography: Basic Theory and Estimation Techniques of Working Brain Activity.
- OTTAVIANI, F., DI GIROLAMO, S., BRIGLIA, G., DE ROSSI, G., DI GIUDA, D. & DI NARDO, W. 1997. Tonotopic organization of human auditory cortex analyzed by SPET. *Audiology*, 36, 241-8.
- OXENHAM, A. J., BERNSTEIN, J. G. & PENAGOS, H. 2004. Correct tonotopic representation is necessary for complex pitch perception. *Proc Natl Acad Sci U S A*, 101, 1421-5.
- OZAKI, I., JIN, C. Y., SUZUKI, Y., BABA, M., MATSUNAGA, M. & HASHIMOTO, I. 2004. Rapid change of tonotopic maps in the human auditory cortex during pitch discrimination. *Clin Neurophysiol*, 115, 1592-604.
- PANTAZIS, D., JOSHI, A., JIANG, J., SHATTUCK, D. W., BERNSTEIN, L. E., DAMASIO, H. & LEAHY, R. M. 2010. Comparison of landmark-based and automatic methods for cortical surface registration. *Neuroimage*, 49, 2479-93.
- PANTEV, C., HOKE, M., LEHNERTZ, K., LUTKENHONER, B., ANOGIANAKIS, G. & WITTKOWSKI, W. 1988. Tonotopic organization of the human auditory cortex revealed by transient auditory evoked magnetic fields. *Electroencephalogr Clin Neurophysiol*, 69, 160-70.
- PANTEV, C., HOKE, M., LEHNERTZ, K. & LUTKENHONER, B. 1989. Neuromagnetic evidence of an amplitopic organization of the human auditory cortex. *Electroencephalogr Clin Neurophysiol*, 72, 225-31.
- PANTEV, C., BERTRAND, O., EULITZ, C., VERKINDT, C., HAMPSON, S., SCHUIERER, G. & ELBERT, T. 1995. Specific tonotopic organizations of different areas of the human auditory cortex revealed by simultaneous magnetic and electric recordings. *Electroencephalogr Clin Neurophysiol*, 94, 26-40.

- PANTEV, C., ROBERTS, L. E., ELBERT, T., ROSS, B. & WIENBRUCH, C. 1996. Tonotopic organization of the sources of human auditory steadystate responses. *Hearing Research*, 101, 62-74.
- PANTEV, C., OOSTENVELD, R., ENGELIEN, A., ROSS, B., ROBERTS, L. E. & HOKE, M. 1998. Increased auditory cortical representation in musicians. *Nature*, 392, 811-4.
- PANTEV, C., WOLLBRINK, A., ROBERTS, L. E., ENGELIEN, A. & LUTKENHONER, B. 1999. Short-term plasticity of the human auditory cortex. *Brain Res*, 842, 192-9.
- PANTEV, C., ENGELIEN, A., CANDIA, V. & ELBERT, T. 2001. Representational cortex in musicians. Plastic alterations in response to musical practice. Ann N Y Acad Sci, 930, 300-14.
- PANTEV, C., ROSS, B., FUJIOKA, T., TRAINOR, L. J., SCHULTE, M. & SCHULZ, M. 2003. Music and learning-induced cortical plasticity. Ann N Y Acad Sci, 999, 438-50.
- PARKES, L. M., MARSMAN, J. B., OXLEY, D. C., GOULERMAS, J. Y. & WUERGER, S. M. 2009. Multivoxel fMRI analysis of color tuning in human primary visual cortex. J Vis, 9, 1 1-13.
- PASCUAL-LEONE, A. 2001. The brain that plays music and is changed by it. Ann N Y Acad Sci, 930, 315-29.
- PATEL, A. D., GIBSON, E., RATNER, J., BESSON, M. & HOLCOMB, P. J. 1998. Processing syntactic relations in language and music: an eventrelated potential study. *J Cogn Neurosci*, 10, 717-33.
- PATEL, A. D. 2008. *Music, language, and the brain*, New York ; Oxford, Oxford University Press.
- PATENAUDE, B., SMITH, S. M., KENNEDY, D. N. & JENKINSON, M. 2011. A Bayesian model of shape and appearance for subcortical brain segmentation. *Neuroimage*, 56, 907-22.
- PATTERSON, R. D., UPPENKAMP, S., JOHNSRUDE, I. S. & GRIFFITHS, T. D. 2002. The processing of temporal pitch and melody information in auditory cortex. *Neuron*, 36, 767-76.
- PAULESU, E., FRITH, C. D. & FRACKOWIAK, R. S. 1993. The neural correlates of the verbal component of working memory. *Nature*, 362, 342-5.
- PENHUNE, V. B., ZATORRE, R. J., MACDONALD, J. D. & EVANS, A. C. 1996. Interhemispheric anatomical differences in human primary auditory

cortex: probabilistic mapping and volume measurement from magnetic resonance scans. *Cereb Cortex*, 6, 661-72.

- PENNY, W. D. & HOLMES, A. J. 2007. Chapter 12 Random Effects Analysis. In: KARL, F., JOHN, A., STEFAN, K., THOMAS, N., WILLIAM PENNYA2 - KARL FRISTON, J. A. S. K. T. N. & WILLIAM, P. (eds.) Statistical Parametric Mapping. London: Academic Press.
- PERETZ, I., KOLINSKY, R., TRAMO, M., LABRECQUE, R., HUBLET, C., DEMEURISSE, G. & BELLEVILLE, S. 1994. Functional dissociations following bilateral lesions of auditory cortex. *Brain*, 117 (Pt 6), 1283-301.
- PETKOV, C. I., KAYSER, C., AUGATH, M. & LOGOTHETIS, N. K. 2006. Functional imaging reveals numerous fields in the monkey auditory cortex. *PLoS Biol*, 4, e215.
- PETKOV, C. I., KAYSER, C., AUGATH, M. & LOGOTHETIS, N. K. 2009. Optimizing the imaging of the monkey auditory cortex: sparse vs. continuous fMRI. *Magn Reson Imaging*, 27, 1065-73.
- PIENKOWSKI, M. & EGGERMONT, J. J. 2011. Cortical tonotopic map plasticity and behavior. *Neurosci Biobehav Rev*.
- PITT, M. A. 1994. Perception of pitch and timbre by musically trained and untrained listeners. J Exp Psychol Hum Percept Perform, 20, 976-86.
- PLACK, C. J., OXENHAM, A. J. & FAY, R. R. 2005. Pitch: neural coding and perception, Springer.
- PLATEL, H., BARON, J. C., DESGRANGES, B., BERNARD, F. & EUSTACHE, F. 2003. Semantic and episodic memory of music are subserved by distinct neural networks. *Neuroimage*, 20, 244-56.
- PLATEL, H. 2005. Functional neuroimaging of semantic and episodic musical memory. Ann N Y Acad Sci, 1060, 136-47.
- RADEMACHER, J., MOROSAN, P., SCHORMANN, T., SCHLEICHER, A., WERNER, C., FREUND, H. J. & ZILLES, K. 2001. Probabilistic mapping and volume measurement of human primary auditory cortex. *Neuroimage*, 13, 669-83.
- RAHNE, T. & SUSSMAN, E. 2009. Neural representations of auditory input accommodate to the context in a dynamically changing acoustic environment. *Eur J Neurosci*, 29, 205-11.
- RAUSCHECKER, J. P. & SCOTT, S. K. 2009. Maps and streams in the auditory cortex: nonhuman primates illuminate human speech processing. *Nat Neurosci*, 12, 718-24.

- RAUSCHER, F. H., SHAW, G. L., LEVINE, L. J., WRIGHT, E. L., DENNIS, W. R. & NEWCOMB, R. L. 1997. Music training causes long-term enhancement of preschool children's spatial-temporal reasoning. *Neurol Res*, 19, 2-8.
- RECANZONE, G. H., SCHREINER, C. E. & MERZENICH, M. M. 1993. Plasticity in the frequency representation of primary auditory cortex following discrimination training in adult owl monkeys. *J Neurosci*, 13, 87-103.
- RECANZONE, G. H., SCHREINER, C. E., SUTTER, M. L., BEITEL, R. E. & MERZENICH, M. M. 1999. Functional organization of spectral receptive fields in the primary auditory cortex of the owl monkey. *J Comp Neurol*, 415, 460-81.
- RESER, D. H., FISHMAN, Y. I., AREZZO, J. C. & STEINSCHNEIDER, M. 2000. Binaural interactions in primary auditory cortex of the awake macaque. *Cereb Cortex*, 10, 574-84.
- RINNE, T., KOISTINEN, S., SALONEN, O. & ALHO, K. 2009. Task-dependent activations of human auditory cortex during pitch discrimination and pitch memory tasks. *J Neurosci*, 29, 13338-43.
- ROBERTSON, D. & IRVINE, D. R. 1989. Plasticity of frequency organization in auditory cortex of guinea pigs with partial unilateral deafness. *J Comp Neurol*, 282, 456-71.
- ROGALSKY, C., RONG, F., SABERI, K. & HICKOK, G. 2011. Functional anatomy of language and music perception: temporal and structural factors investigated using functional magnetic resonance imaging. *J Neurosci*, 31, 3843-52.
- ROJAS, D. C., BAWN, S. D., CARLSON, J. P., ARCINIEGAS, D. B., TEALE, P. D. & REITE, M. L. 2002. Alterations in tonotopy and auditory cerebral asymmetry in schizophrenia. *Biol Psychiatry*, 52, 32-9.
- ROMANI, G. L., WILLIAMSON, S. J. & KAUFMAN, L. 1982a. Tonotopic organization of the human auditory cortex. *Science*, 216, 1339-40.
- ROMANI, G. L., WILLIAMSON, S. J., KAUFMAN, L. & BRENNER, D. 1982b. Characterization of 'the human auditory cortex by the neuromagnetic method. *Exp Brain Res*, 47, 381-93.
- ROMERO, L., WALSH, V. & PAPAGNO, C. 2006. The neural correlates of phonological short-term memory: a repetitive transcranial magnetic stimulation study. *J Cogn Neurosci*, 18, 1147-55.

- ROSEN, B. R., BUCKNER, R. L. & DALE, A. M. 1998. Event-related functional MRI: past, present, and future. *Proc Natl Acad Sci U S A*, 95, 773-80.
- ROSENKRANZ, K., WILLIAMON, A. & ROTHWELL, J. C. 2007. Motorcortical excitability and synaptic plasticity is enhanced in professional musicians. *J Neurosci*, 27, 5200-6.
- ROSS, B., BORGMANN, C., DRAGANOVA, R., ROBERTS, L. E. & PANTEV, C. 2000. A high-precision magnetoencephalographic study of human auditory steady-state responses to amplitude-modulated tones. J Acoust Soc Am, 108, 679-91.
- ROTHER, J., KNAB, R., HAMZEI, F., FIEHLER, J., REICHENBACH, J. R., BUCHEL, C. & WEILLER, C. 2002. Negative dip in BOLD fMRI is caused by blood flow--oxygen consumption uncoupling in humans. *Neuroimage*, 15, 98-102.
- RÜSSELER, J., ALTENM, LLER, E., NAGER, W., KOHLMETZ, C. & M, NTE, T. F. 2001. Event-related brain potentials to sound omissions differ in musicians and non-musicians. *Neurosci Lett*, 308, 33-36.
- RUTHERFORD, W. 1886. A New Theory of Hearing. J Anat Physiol, 21, 166-8.
- RUYTJENS, L., GEORGIADIS, J. R., HOLSTEGE, G., WIT, H. P., ALBERS, F.
   W. & WILLEMSEN, A. T. 2007. Functional sex differences in human primary auditory cortex. *Eur J Nucl Med Mol Imaging*, 34, 2073-81.
- RYDING, K. C. 1998. Early medieval Arabic : studies on al-Khali\0304l ibn Ah\0323mad, Washington, D.C., Georgetown University Press.
- SAKURAI, Y. 2002. Coding of auditory temporal and pitch information by hippocampal individual cells and cell assemblies in the rat. *Neuroscience*, 115, 1153-1163.
- SALKIND, N. J. & RASMUSSEN, K. 2007. Encyclopedia of measurement and statistics, Thousand Oaks, Calif., SAGE Publications.
- SALMON, E., VAN DER LINDEN, M., COLLETTE, F., DELFIORE, G., MAQUET, P., DEGUELDRE, C., LUXEN, A. & FRANCK, G. 1996. Regional brain activity during working memory tasks. *Brain*, 119 (Pt 5), 1617-25.
- SCHLAUG, G., JANCKE, L., HUANG, Y., STAIGER, J. F. & STEINMETZ, H. 1995a. Increased corpus callosum size in musicians. *Neurophysiology*, 33, 1047-55.

- SCHLAUG, G., JANCKE, L., HUANG, Y. & STEINMETZ, H. 1995b. In vivo evidence of structural brain asymmetry in musicians. *Science*, 267, 699-701.
- SCHLAUG, G. 2001. The brain of musicians. A model for functional and structural adaptation. *Ann N Y Acad Sci*, 930, 281-99.
- SCHLAUG, G., FORGEARD, M., ZHU, L., NORTON, A. & WINNER, E. 2009a. Training-induced neuroplasticity in young children. Ann N Y Acad Sci, 1169, 205-8.
- SCHLAUG, G., MARCHINA, S. & NORTON, A. 2009b. Evidence for plasticity in white-matter tracts of patients with chronic Broca's aphasia undergoing intense intonation-based speech therapy. *Ann N Y Acad Sci*, 1169, 385-94.
- SCHMIDT, C. F., ZAEHLE, T., MEYER, M., GEISER, E., BOESIGER, P. & JANCKE, L. 2008. Silent and continuous fMRI scanning differentially modulate activation in an auditory language comprehension task. *Hum Brain Mapp*, 29, 46-56.
- SCHMITHORST, V. J. & HOLLAND, S. K. 2003. The effect of musical training on music processing: a functional magnetic resonance imaging study in humans. *Neurosci Lett*, 348, 65-68.
- SCHNEIDER, P. 2001. Source activity and tonotopic organization of the auditory cortex in musicians and non-musicians. University of Heidelberg.
- SCHNEIDER, P., SCHERG, M., DOSCH, H. G., SPECHT, H. J., GUTSCHALK, A. & RUPP, A. 2002. Morphology of Heschl's gyrus reflects enhanced activation in the auditory cortex of musicians. *Nat Neurosci*, 5, 688-94.
- SCHNEIDER, P., SLUMING, V., ROBERTS, N., SCHERG, M., GOEBEL, R.,
   SPECHT, H. J., DOSCH, H. G., BLEECK, S., STIPPICH, C. & RUPP, A. 2005. Structural and functional asymmetry of lateral Heschl's gyrus reflects pitch perception preference. *Nat Neurosci*, 8, 1241-7.
- SCHNEIDER, S., SCHONLE, P. W., ALTENMULLER, E. & MUNTE, T. F. 2007. Using musical instruments to improve motor skill recovery following a stroke. *J Neurol*, 254, 1339-46.
- SCHÖN, D., MAGNE, C. & BESSON, M. 2004. The music of speech: Music training facilitates pitch processing in both music and language. *Psychophysiology*, 41, 341-349.
- SCHREINER, C. E. 1991. Functional topographies in the primary auditory cortex of the cat. *Acta Otolaryngol Suppl*, 491, 7-15; discussion 16.

ę

- SCHULZ, M., ROSS, B. & PANTEV, C. 2003. Evidence for training-induced crossmodal reorganization of cortical functions in trumpet players. *Neuroreport*, 14, 157-61.
- SCHWEIZER, T. A., WARE, J., FISCHER, C. E., CRAIK, F. I. & BIALYSTOK, E. 2011. Bilingualism as a contributor to cognitive reserve: Evidence from brain atrophy in Alzheimer's disease. *Cortex*.
- SEGHIER, M. L., LAZEYRAS, F., PEGNA, A. J., ANNONI, J. M. & KHATEB, A. 2008. Group analysis and the subject factor in functional magnetic resonance imaging: analysis of fifty right-handed healthy subjects in a semantic language task. *Hum Brain Mapp*, 29, 461-77.
- SEIFRITZ, E., BILECEN, D., HANGGI, D., HASELHORST, R., RADU, E. W., WETZEL, S., SEELIG, J. & SCHEFFLER, K. 2000. Effect of ethanol on BOLD response to acoustic stimulation: implications for neuropharmacological fMRI. *Psychiatry Research*, 99, 1-13.
- SEIFRITZ, E., DI SALLE, F., ESPOSITO, F., HERDENER, M., NEUHOFF, J. G. & SCHEFFLER, K. 2006. Enhancing BOLD response in the auditory system by neurophysiologically tuned fMRI sequence. *Neuroimage*, 29, 1013-22.
- SEITHER-PREISLER, A., JOHNSON, L., KRUMBHOLZ, K., NOBBE, A., PATTERSON, R., SEITHER, S. & LUTKENHONER, B. 2007. Tone sequences with conflicting fundamental pitch and timbre changes are heard differently by musicians and nonmusicians. J Exp Psychol Hum Percept Perform, 33, 743-51.
- SHAH, N. J., JÄNCKE, L., GROSSE-RUYKEN, M. L. & MÜLLER-GÄRTNER,
   H. W. 1999. Influence of acoustic masking noise in fMRI of the auditory cortex during phonetic discrimination. *J Magn Reson Imaging*, 9, 19-25.
- SHAH, N. J., STEINHOFF, S., MIRZAZADE, S., ZAFIRIS, O., GROSSE-RUYKEN, M. L., JANCKE, L. & ZILLES, K. 2000. The effect of sequence repeat time on auditory cortex stimulation during phonetic discrimination. *Neuroimage*, 12, 100-8.
- SHAHIN, A., BOSNYAK, D. J., TRAINOR, L. J. & ROBERTS, L. E. 2003. Enhancement of neuroplastic P2 and N1c auditory evoked potentials in musicians. J Neurosci, 23, 5545-52.
- SHAHIN, A., ROBERTS, L. E. & TRAINOR, L. J. 2004. Enhancement of auditory cortical development by musical experience in children. *Neuroreport*, 15, 1917-21.
- SHAHIN, A. J. 2011. Neurophysiological influence of musical training on speech perception. *Front Psychol*, 2, 126.

- SHELLOCK, F. G., ZIARATI, M., ATKINSON, D. & CHEN, D. Y. 1998. Determination of gradient magnetic field-induced acoustic noise associated with the use of echo planar and three-dimensional, fast spin echo techniques. *J Magn Reson Imaging*, 8, 1154-7.
- SLUMING, V., BARRICK, T., HOWARD, M., CEZAYIRLI, E., MAYES, A. & ROBERTS, N. 2002. Voxel-Based Morphometry Reveals Increased Gray Matter Density in Broca's Area in Male Symphony Orchestra Musicians. *Neuroimage*, 17, 1613-1622.
- SLUMING, V., BROOKS, J., HOWARD, M., DOWNES, J. J. & ROBERTS, N. 2007a. Broca's area supports enhanced visuospatial cognition in orchestral musicians. J Neurosci, 27, 3799-806.
- SLUMING, V., PAGE, D., DOWNES, J., MAYES, A., DENBY, C. & ROBERTS, N. Hippocampal Volumes and Visuospatial Memory in Musicians. European Congress of Radiology, 2007b Vienna 1.
- SMITH, S. M. 2002. Fast robust automated brain extraction. *Hum Brain Mapp*, 17, 143-55.
- SMITH, S. M., JENKINSON, M., WOOLRICH, M. W., BECKMANN, C. F., BEHRENS, T. E., JOHANSEN-BERG, H., BANNISTER, P. R., DE LUCA, M., DROBNJAK, I., FLITNEY, D. E., NIAZY, R. K., SAUNDERS, J., VICKERS, J., ZHANG, Y., DE STEFANO, N., BRADY, J. M. & MATTHEWS, P. M. 2004. Advances in functional and structural MR image analysis and implementation as FSL. *Neuroimage*, 23 Suppl 1, S208-19.
- SPIEGEL, M. F. & WATSON, C. S. 1984. Performance on frequencydiscrimination tasks by musicians and nonmusicians. J Acoust Soc Am, 76, 1690-1695.
- SRULOVICZ, P. & GOLDSTEIN, J. L. 1983. A central spectrum model: A synthesis of auditory-nerve timing and place cues in monaural communication of frequency spectrum. *J Acoust Soc Am*, 73, 1266-1276.
- STIPPICH, C., BLATOW, M. & DELMAIRE, C. 2007. Clinical Functional MRI: Presurgical Functional Neuroimaging, Springer.
- STOECKEL, C., GOUGH, P. M., WATKINS, K. E. & DEVLIN, J. T. 2009. Supramarginal gyrus involvement in visual word recognition. *Cortex*, 45, 1091-6.
- STRAINER, J. C., ULMER, J. L., YETKIN, F. Z., HAUGHTON, V. M., DANIELS, D. L. & MILLEN, S. J. 1997. Functional MR of the primary auditory cortex: an analysis of pure tone activation and tone discrimination. *AJNR - American Journal of Neuroradiology*, 18, 601-10.

- STRIEM-AMIT, E., HERTZ, U. & AMEDI, A. 2011. Extensive Cochleotopic Mapping of Human Auditory Cortical Fields Obtained with Phase-Encoding fMRI. *PLoS ONE*, 6, e17832.
- SUGA, N. & MA, X. 2003. Multiparametric corticofugal modulation and plasticity in the auditory system. *Nat Rev Neurosci*, 4, 783-94.
- TALAVAGE, T. M., EDMISTER, W. B., LEDDEN, P. J. & WEISSKOFF, R. M. 1999. Quantitative assessment of auditory cortex responses induced by imager acoustic noise. *Hum Brain Mapp*, 7, 79-88.
- TALAVAGE, T. M., LEDDEN, P. J., BENSON, R. R., ROSEN, B. R. & MELCHER, J. R. 2000. Frequency-dependent responses exhibited by multiple regions in human auditory cortex. *Hearing Research*, 150, 225-44.
- TALAVAGE, T. M., SERENO, M. I., MELCHER, J. R., LEDDEN, P. J., ROSEN, B. R. & DALE, A. M. 2004. Tonotopic organization in human auditory cortex revealed by progressions of frequency sensitivity. J Neurophysiol, 91, 1282-96.
- TAMRAZ, J. C. & COMAIR, Y. G. 2000. Atlas of regional anatomy of the brain using MRI with functional correlations, Berlin; London, Springer.
- TANAKA, E., KIDA, T., INUI, K. & KAKIGI, R. 2009. Change-driven cortical activation in multisensory environments: an MEG study. *Neuroimage*, 48, 464-74.
- TECCHIO, F., BICCIOLO, G., DE CAMPORA, E., PASQUALETTI, P., PIZZELLA, V., INDOVINA, I., CASSETTA, E., ROMANI, G. L. & ROSSINI, P. M. 2000. Tonotopic cortical changes following stapes substitution in otosclerotic patients: a magnetoencephalographic study. *Hum Brain Mapp*, 10, 28-38.
- TERVANIEMI, M., KUJALA, A., ALHO, K., VIRTANEN, J., ILMONIEMI, R. J. & NAATANEN, R. 1999. Functional specialization of the human auditory cortex in processing phonetic and musical sounds: A magnetoencephalographic (MEG) study. *Neuroimage*, 9, 330-6.
- TERVANIEMI, M., JUST, V., KOELSCH, S., WIDMANN, A. & SCHROGER, E. 2005. Pitch discrimination accuracy in musicians vs nonmusicians: an event-related potential and behavioral study. *Exp Brain Res*, 161, 1-10.
- THOMPSON, P. M., CANNON, T. D., NARR, K. L., VAN ERP, T., POUTANEN, V. P., HUTTUNEN, M., LONNQVIST, J., STANDERTSKJOLD-NORDENSTAM, C. G., KAPRIO, J., KHALEDY, M., DAIL, R., ZOUMALAN, C. I. & TOGA, A. W. 2001. Genetic influences on brain structure. *Nat Neurosci*, 4, 1253-8.

- THOMPSON, P. M., MARTIN, N. G. & WRIGHT, M. J. 2010. Imaging genomics. Curr Opin Neurol, 23, 368-73.
- THOMPSON, W. F., SCHELLENBERG, E. G. & HUSAIN, G. 2003. Perceiving prosody in speech. Effects of music lessons. *Ann N Y Acad Sci*, 999, 530-2.
- THOMPSON, W. F., SCHELLENBERG, E. G. & HUSAIN, G. 2004. Decoding speech prosody: do music lessons help? *Emotion*, 4, 46-64.
- TILLMANN, B., JANATA, P. & BHARUCHA, J. J. 2003. Activation of the inferior frontal cortex in musical priming. *Brain Res Cogn Brain Res*, 16, 145-61.
- TOGA, A. W. & THOMPSON, P. M. 2005. Genetics of brain structure and intelligence. Annu Rev Neurosci, 28, 1-23.
- TONDELLI, M., WILCOCK, G. K., NICHELLI, P., DE JAGER, C. A., JENKINSON, M. & ZAMBONI, G. 2011. Structural MRI changes detectable up to ten years before clinical Alzheimer's disease. *Neurobiol Aging*.
- TRAINOR, L. J., LEE, K. & BOSNYAK, D. J. 2011. Cortical Plasticity in 4-Month-Old Infants: Specific Effects of Experience with Musical Timbres. *Brain Topogr.*
- TRAMO, M. J., SHAH, G. D. & BRAIDA, L. D. 2002. Functional role of auditory cortex in frequency processing and pitch perception. J Neurophysiol, 87, 122-39.
- ULMER, J. L., BISWAL, B. B., YETKIN, F. Z., MARK, L. P., MATHEWS, V. P., PROST, R. W., ESTKOWSKI, L. D., MCAULIFFE, T. L., HAUGHTON, V. M. & DANIELS, D. L. 1998. Cortical activation response to acoustic echo planar scanner noise. J Comput Assist Tomogr, 22, 111-9.
- UPADHYAY, J., DUCROS, M., KNAUS, T. A., LINDGREN, K. A., SILVER, A., TAGER-FLUSBERG, H. & KIM, D. S. 2007. Function and connectivity in human primary auditory cortex: a combined fMRI and DTI study at 3 Tesla. *Cereb Cortex*, 17, 2420-32.
- VALENTINE, P. A. & EGGERMONT, J. J. 2003. Intracortical microstimulation induced changes in spectral and temporal response properties in cat auditory cortex. *Hearing Research*, 183, 109-125.
- VAN ESSEN, D. 2005. A Population-Average, Landmark- and Surface-based (PALS) atlas of human cerebral cortex. *Neuroimage*, 28, 635-662.

£

192

- VAN LEEMPUT, K., BAKKOUR, A., BENNER, T., WIGGINS, G., WALD, L. L., AUGUSTINACK, J., DICKERSON, B. C., GOLLAND, P. & FISCHL, B. 2009. Automated segmentation of hippocampal subfields from ultra-high resolution in vivo MRI. *Hippocampus*, 19, 549-57.
- VINES, B. W., SCHNIDER, N. M. & SCHLAUG, G. 2006. Testing for causality with transcranial direct current stimulation: pitch memory and the left supramarginal gyrus. *Neuroreport*, 17, 1047-50.
- VRBA, J. & ROBINSON, S. E. 2001. Signal processing in magnetoencephalography. *Methods*, 25, 249-71.
- WAN, C. Y. & SCHLAUG, G. 2010. Music making as a tool for promoting brain plasticity across the life span. *The Neuroscientist*, 16, 566-77.
- WARREN, J. D. & GRIFFITHS, T. D. 2003. Distinct mechanisms for processing spatial sequences and pitch sequences in the human auditory brain. J Neurosci, 23, 5799-804.
- WARREN, J. D., UPPENKAMP, S., PATTERSON, R. D. & GRIFFITHS, T. D. 2003. Analyzing Pitch Chroma and Pitch Height in the Human Brain. Ann N Y Acad Sci, 999, 212-214.
- WARREN, R. M. 2008. Auditory perception: an analysis and synthesis, Cambridge University Press.
- WATANABE, T., YAGISHITA, S. & KIKYO, H. 2008. Memory of music: Roles of right hippocampus and left inferior frontal gyrus. *Neuroimage*, 39, 483-491.
- WEISZ, N., KEIL, A., WIENBRUCH, C., HOFFMEISTER, S. & ELBERT, T. 2004. One set of sounds, two tonotopic maps: exploring auditory cortex with amplitude-modulated tones. *Clin Neurophysiol*, 115, 1249-58.
- WESSINGER, C. M., BUONOCORE, M. H., KUSSMAUL, C. L. & MANGUN, G. R. 1997. Tonotopy in human auditory cortex examined with functional magnetic resonance imaging. *Hum Brain Mapp*, 5, 18-25.

WESTBROOK, C. 2002. MRI at a glance, Oxford, Blackwell Science.

WEVER, E. G. 1949. Theory of hearing, Wiley.

ŧ

WOODS, D. L., STECKER, G. C., RINNE, T., HERRON, T. J., CATE, A. D., YUND, E. W., LIAO, I. & KANG, X. 2009. Functional maps of human auditory cortex: effects of acoustic features and attention. *PLoS ONE*, 4, e5183.

- WOOLRICH, M. W., JBABDI, S., PATENAUDE, B., CHAPPELL, M., MAKNI, S., BEHRENS, T., BECKMANN, C., JENKINSON, M. & SMITH, S. M. 2009. Bayesian analysis of neuroimaging data in FSL. *Neuroimage*, 45, S173-86.
- WRIGHT, B. & SABIN, A. 2007. Perceptual learning: how much daily training is enough? *Exp Brain Res*, 180, 727-736.
- YAMAMOTO, T., WILLIAMSON, S. J., KAUFMAN, L., NICHOLSON, C. & LLINAS, R. 1988. Magnetic localization of neuronal activity in the human brain. *Proc Natl Acad Sci U S A*, 85, 8732-6.
- YANG, Y., ENGELIEN, A., ENGELIEN, W., XU, S., STERN, E. & SILBERSWEIG, D. A. 2000. A silent event-related functional MRI technique for brain activation studies without interference of scanner acoustic noise. *Magn Reson Med*, 43, 185-90.
- YAROSLAV O. HALCHENKO, STEPHEN JOSE HANSON & PEARLMUTTER, B. A. 2005. Advanced image processing in magnetic resonance imaging. In: LANDINI, L., POSITANO, V. & SANTARELLI, M. F. (eds.) Signal processing and communications. Boca Raton, FL: CRC/Taylor & Francis.
- YETKIN, F. Z., ROLAND, P. S., PURDY, P. D. & CHRISTENSEN, W. F. 2003. Evaluation of auditory cortex activation by using silent FMRI. Am J Otolaryngol, 24, 281-9.
- YOON, U., PERUSSE, D., LEE, J. M. & EVANS, A. C. 2011. Genetic and environmental influences on structural variability of the brain in pediatric twin: deformation based morphometry. *Neurosci Lett*, 493, 8-13.
- YOUSEM, D. M., MALDJIAN, J. A., HUMMEL, T., ALSOP, D. C., GECKLE, R. J., KRAUT, M. A. & DOTY, R. L. 1999. The effect of age on odorstimulated functional MR imaging. AJNR - American Journal of Neuroradiology, 20, 600-8.
- YOUSRY, T. A., SCHMID, U. D., ALKADHI, H., SCHMIDT, D., PERAUD, A., BUETTNER, A. & WINKLER, P. 1997. Localization of the motor hand area to a knob on the precentral gyrus. A new landmark. *Brain*, 120 (Pt 1), 141-57.
- ZATORRE, R. & MCGILL, J. 2005. Music, the food of neuroscience? *Nature*, 434, 312-5.
- ZATORRE, R. J., EVANS, A. C., MEYER, E. & GJEDDE, A. 1992. Lateralization of phonetic and pitch discrimination in speech processing. *Science*, 256, 846-9.

ę.

- ZATORRE, R. J., BELIN, P. & PENHUNE, V. B. 2002a. Structure and function of auditory cortex: music and speech. *Trends Cogn Sci*, 6, 37-46.
- ZATORRE, R. J., BOUFFARD, M., AHAD, P. & BELIN, P. 2002b. Where is 'where' in the human auditory cortex? *Nat Neurosci*, 5, 905-9.
- ZENDEL, B. R. & ALAIN, C. 2009. Concurrent sound segregation is enhanced in musicians. *J Cogn Neurosci*, 21, 1488-1498.

¢